SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: 2022-11-17 | Period of Report: 2022-11-15 SEC Accession No. 0001213900-22-073255

(HTML Version on secdatabase.com)

FILER

Quantum FinTech Acquisition Corp

CIK:1830795| IRS No.: 853286402 | State of Incorp.:DE | Fiscal Year End: 1231 Type: 8-K | Act: 34 | File No.: 001-40009 | Film No.: 221396242 SIC: 6770 Blank checks

Mailing Address 4221 W. BOY SCOUT BLVD. 4221 W. BOY SCOUT BLVD. SUITE 300 TAMPA FL 33607

Business Address SUITE 300 TAMPA FL 33607 720-656-2176

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): November 15, 2022

QUANTUM FINTECH ACQUISITION CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

001-40009

(Commission File Number) 85-3286402

(IRS Employer Identification No.)

4221 W. Boy Scout Blvd., Suite 300

Tampa, FL 33607

(Address of principal executive offices, including zip code)

Registrant's telephone number, including area code: (813) 257-9366

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Dere-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Units, each consisting of one share of Common Stock and one redeemable Warrant	QFTA.U	The New York Stock Exchange
Common Stock, par value \$0.0001 per share	QFTA	The New York Stock Exchange

QFTAW

Warrants, each warrant exercisable for one-half of one share of Common Stock at an exercise price of \$11.50

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company $\mathbf{\overline{X}}$

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01 Entry Into a Material Definitive Agreement.

Business Combination Agreement

On November 16, 2022, Quantum FinTech Acquisition Corporation (the "Company"), a Delaware corporation, entered into a Business Combination Agreement (the "Business Combination Agreement") by and among the Company, Calculator New Pubco, Inc., a Delaware corporation and a wholly-owned subsidiary of the Company ("New Pubco"), Calculator Merger Sub 1, Inc., a Delaware corporation and a wholly-owned subsidiary of New Pubco ("Merger Sub 1"), Calculator Merger Sub 2, Inc., a Delaware corporation and a wholly-owned subsidiary of New Pubco ("Merger Sub 1"), Calculator Merger Sub 2, Inc., a Delaware corporation and a wholly-owned subsidiary of New Pubco ("Merger Sub 2"), AtlasClear, Inc., a Wyoming corporation ("AtlasClear"), Atlas FinTech Holdings Corp., a Delaware corporation ("Atlas FinTech") and Robert McBey.

The Business Combination Agreement was unanimously approved by the Company's board of directors based upon the unanimous recommendation of a special committee of independent directors. If the Business Combination Agreement is approved by the Company's stockholders, and the transactions contemplated by the Business Combination Agreement are consummated, (i) Merger Sub 1 will merge with and into the Company, with the Company continuing as the surviving corporation and a wholly-owned subsidiary of New Pubco and (ii) Merger Sub 2 will merge with and into AtlasClear, with AtlasClear continuing as the surviving corporation and a wholly-owned subsidiary of New Pubco (collectively, the "Business Combination").

Prior to the closing of the Business Combination (the "Closing"), AtlasClear will receive certain assets from Atlas FinTech and Atlas Financial Technologies Corp., will complete the acquisition of broker-dealer Wilson-Davis & Co., Inc. ("WDCO") and will consummate the transaction with Pacsquare Technologies, LLC ("Pacsquare"). In addition, at Closing, the definitive agreement pursuant to which AtlasClear has agreed to acquire Commercial Bancorp, a Wyoming corporation ("CB") shall continue to be in full force and effect (the "CB Merger Agreement"). The Company expects the Closing to occur before the closing of the transactions contemplated by the CB Merger Agreement (the "CB Closing")

At the Closing, AtlasClear stockholders will receive merger consideration in shares of New Pubco common stock equal to the quotient of (i) \$75.4 million, less the purchase prices for WDCO and CB, divided by (ii) \$10.

In addition, the AtlasClear stockholders will receive up to 5,944,444 shares of New Pubco common stock (the "Earn Out Shares"). The Earn Out Shares will be issued to AtlasClear stockholders upon certain milestones (based on the achievement of certain price targets of New Pubco common stock following the Closing). In the event such milestones are not met within the first 18 months following the Closing, the Earn Out Shares will be cancelled. Atlas FinTech will also receive up to \$20 million of New Pubco common stock ("Software Products Earn Out Shares"), which will be issued to Atlas FinTech upon certain milestones based on the achievement of certain revenue targets of software products contributed to AtlasClear by Atlas FinTech and Atlas Financial Technologies Corp. following the Closing. The revenue targets will be measured yearly for the five years following Closing, with no catch-up between the years.

In connection with the Closing, each share of the Company's common stock ("Company Common Stock") (other than shares held by Atlas FinTech) that is outstanding and has not been redeemed will be converted into one share of New Pubco common stock.

Each outstanding warrant to purchase Company Common Stock ("Company Warrant") (other than Private Warrants, described below) will become a warrant to purchase one-half of a share of New Pubco common stock. Each outstanding warrant to purchase Company Common Stock initially issued in a private placement in connection with the Company's initial public offering ("Private Warrant") will become a warrant to purchase one share of New Pubco common stock.

Atlas FinTech, which directly or indirectly holds shares of Company Common Stock and Private Warrants, has agreed to transfer, or cause to transfer, up to 1,279,426.82 of Company Common Stock and up to 1,657,578.65 of the Private Warrants held indirectly by it to potential sources of debt, equity or financing if the Company pursues financing between signing and the Closing. Any of such Company Common Stock or Private Warrants remaining following any transfers for potential financing will be forfeited for no consideration.

Conditions to Closing

The Closing is subject to certain customary conditions, including, among other things, (i) approval by the Company's stockholders of the Business Combination Agreement, (ii) the effectiveness of a registration statement on Form S-4 by New Pubco relating to the Business Combination and containing a proxy statement of the Company, (iii) the approval for listing on The New York Stock Exchange of the shares of New Pubco common stock to be issued in the Business Combination, (iv) the completion of the contribution of certain specified assets from Atlas FinTech and Atlas Financial Technologies Corp. to AtlasClear, (v) the completion by AtlasClear of the acquisition of WDCO, (vi) the CB Merger Agreement remaining in effect and (vii) available cash at closing of at least \$40 million. In addition, the Closing is subject to the condition that the Company has at least \$5,000,001 of net tangible assets (as determined in accordance with Rule 3a51-1(g)(1) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")).

1

Representations, Warranties and Covenants

The parties to the Business Combination Agreement have made representations, warranties and covenants that are customary for transactions of this nature.

Termination

The Business Combination Agreement may be terminated by either AtlasClear or the Company under certain circumstances, including, among others, (i) by mutual written consent of both AtlasClear and the Company, (ii) by either AtlasClear or the Company if the Closing has not occurred on or before February 9, 2023 (provided, that if the Company seeks and obtains an extension to the deadline by which it must complete its business combination, the Company shall have the right by providing notice to AtlasClear to extend such date for up to the shorter of (a) up to two additional periods equal to three additional months each (for a total of not more than six months in the aggregate), (b) the period ending on the last date for the Company to consummate its business combination pursuant to such extension and (c) such period as determined by the Company), (iii) by either AtlasClear or the Company if a government authority has issued an order or taken any other action permanently restraining, enjoining or otherwise prohibiting the Business Combination, (iv) by either AtlasClear or the Company if the other party has materially breached their respective representations or covenants under the Business Combination Agreement and has not timely cured such breach, (v) by the Company if there has been a material adverse effect on the Company that is not timely cured, (vi) by the Company if the agreements for the contribution of Atlas FinTech assets or the acquisitions of WDCO, CB or Pacsquare assets are terminated or if it is reasonably likely that the conditions in the acquisition agreements for WDCO or CB cannot be satisfied, (vii) by either AtlasClear or the Company if the Company has held a stockholder meeting to approve the Business Combination and approval of the Business Combination has not been obtained by the requisite number of stockholders of the Company and (viii) by the Company if all required financial statements for the Form S-4 are not delivered within 30 days from the date of the Business Combination Agreement.

The foregoing description of the Business Combination Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Business Combination Agreement, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference. The Business Combination Agreement contains representations, warranties and covenants that the parties to the Business Combination Agreement made to each other as of the date of the Business Combination Agreement or other specific dates. The assertions embodied in those representations, warranties and covenants were made for purposes of the contract among the parties and are subject to important qualifications and limitations agreed to by the parties in connection with negotiating the Business Combination Agreement. The Business Combination Agreement has been attached to provide investors with information regarding its terms and is not intended to provide any other factual information about the Company, AtlasClear or any other party to the Business Combination

Agreement. In particular, the representations, warranties, covenants and agreements contained in the Business Combination Agreement, which were made only for purposes of the Business Combination Agreement and as of specific dates, were solely for the benefit of the parties to the Business Combination Agreement (other than as expressly provided for in the Business Combination Agreement), may be subject to limitations agreed upon by the contracting parties (including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the Business Combination Agreement instead of establishing these matters as facts) and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors and reports and documents filed with the U.S. Securities and Exchange Commission (the "SEC"). Investors should not rely on the representations, warranties, covenants or agreements, or any descriptions thereof, as characterizations of the actual state of facts or condition of any party to the Business Combination Agreement may be subject to subsequent waiver or modification. Moreover, information concerning the subject matter of the representations and warranties and other terms may change after the date of the Business Combination Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Purchaser Support Agreement

In connection with the Company's entrance into the Business Combination Agreement, it also entered into a Purchaser Support Agreement (the "Purchaser Support Agreement") with AtlasClear, Quantum Ventures LLC ("Quantum Ventures") and the Company's directors and officers (collectively with Quantum Ventures, the "Sponsors"), pursuant to which, among other things, the Sponsors will agree to vote (or cause to vote) any of the shares of Company Common Stock held by them in favor of the Business Combination at any meeting of the Company stockholders and to not redeem any such shares in connection with the transactions contemplated by the Business Combination Agreement. Quantum Ventures also agreed to transfer up to 1,279,426.82 of the Company Common Stock and up to 1,657,578.65 of the Private Warrants held directly or indirectly by Atlas FinTech to potential sources of debt, equity or financing if the Company pursues financing between signing and the Closing and that any of such Company Common Stock or Private Warrants remaining following any transfers for potential financing will be forfeited for no consideration at the Closing.

2

The foregoing description of the Purchaser Support Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Purchaser Support Agreement, the form of which is filed as Exhibit 10.1 hereto and is incorporated by reference herein.

Registration Rights Agreement and Lock-Up Agreement

In connection with the Company's entry into the Business Combination Agreement, the Company, New Pubco, AtlasClear's stockholders and the Sponsors entered into a Registration Rights and Lock-Up Agreement (the "Registration Rights Agreement"), which will be effective at Closing and pursuant to which New Pubco agreed, among other things, to provide the parties thereto customary demand, shelf and piggy-back rights on secondary offerings, subject to customary cut-back provisions and coordinated offerings. New Pubco has agreed to file a resale shelf registration statement within 45 days of Closing and to effect up to five underwritten offerings each expected to yield gross proceeds of more than \$10 million.

The Sponsors will agree to transfer shares of New Pubco common stock received at Closing into an escrow account, which will be released (i) 50% on the earlier of six months from Closing and the date on which the New Pubco common stock equals or exceeds \$12.50 for 20 days in any 30-day trading window; and (ii) 50% on the six-month anniversary of Closing. In addition, the Sponsors will agree to a lockup of Private Warrants for 30 days after Closing.

AtlasClear's stockholders, including Atlas FinTech, will agree to a lockup of shares of New Pubco common stock received at Closing for one year following Closing, provided that such lockup shall be released early if the New Pubco common stock equals or exceeds \$12.50 for 20 days in any 30-day trading window.

The foregoing description of the Registration Rights Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Registration Rights Agreement, the form of which is filed as Exhibit 10.2 hereto and is incorporated by reference herein.

Commercial Bancorp Merger Agreement

In connection with the Company's entry into the Business Combination Agreement, AtlasClear, CB and New Pubco, with respect to Section 6.16 only, entered into the CB Merger Agreement.

If the CB Merger Agreement is approved by CB's stockholders, and the transactions contemplated by the CB Merger Agreement are consummated, a wholly-owned subsidiary of AtlasClear shall merge with and into CB, with CB continuing as the surviving corporation and a wholly-owned subsidiary of AtlasClear, and immediately thereafter, CB will merge with and into AtlasClear, with AtlasClear continuing as the surviving corporation and a wholly-owned subsidiary of New Pubco (the "CB Acquisition").

The CB Closing is subject to certain customary conditions, including, among other things, (i) approval by the stockholders of CB, CB's wholly-owned subsidiary, Farmers State Bank, AtlasClear and New Pubco, (ii) receipt of certain regulatory approvals and (iii) the prior completion of the Business Combination. The Company expects the Closing to occur before the CB Closing.

At the CB Closing, CB stockholders will be entitled to elect to receive merger consideration (i) in cash or (ii) in a mix of cash and shares of New Pubco common stock. CB stockholders electing to receive merger consideration in cash will receive two times the adjusted estimated book value of common stock of CB at the CB Closing, plus a pro rata share of certain valuation and tax benefits. CB stockholders electing to receive merger consideration in a mix of cash and shares of New Pubco common stock will receive three times the adjusted estimated book value of common stock of CB at the CB Closing, plus a pro rata share of certain valuation and tax benefits, receive three times the adjusted estimated book value of common stock of CB at the CB Closing, plus a pro rata share of certain valuation and tax benefits, paid one-third in cash and two-thirds in New Pubco common stock, valued at \$10 per share.

New Pubco is party to the CB Merger Agreement solely with respect to Section 6.16, pursuant to which New Pubco has agreed to use reasonable best efforts to include New Pubco common stock issuable in connection with the CB Acquisition on the Form S-4 to be filed in connection with the Business Combination. To the extent the SEC does not permit the New Pubco common stock issuable in connection with the CB Acquisition to be included on the Form S-4 to be filed in connection with the Business Combination, New Pubco has agreed to offer and issue the New Pubco common stock to CB stockholders in a transaction exempt from the registration requirements of the Securities Act of 1933, as amended (the "Securities Act"). If shares of New Pubco common stock are issued to CB stockholders pursuant to a transaction exempt from the registration requirements of the Securities Act, New Pubco has agreed to a resale shelf registration statement within 30 days of the CB Closing.

3

New Pubco's obligations under the CB Merger Agreement automatically terminate if the Business Combination Agreement is terminated.

The foregoing description of the CB Merger Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Merger Agreement, which is filed as Exhibit 10.3 hereto and is incorporated by reference herein.

Item 1.02. Termination of a Material Definitive Agreement

As previously disclosed, on November 4, 2021, the Company entered into an Agreement and Plan of Merger (the "TS Merger Agreement") by and among the Company, TradeStation Group, Inc., a Florida corporation ("TradeStation"), and TSG Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of TradeStation ("TS Merger Sub") pursuant to which, among other things, TS Merger Sub will merge with and into the Company, with the Company surviving as the surviving corporation and a wholly-owned subsidiary of TradeStation.

On November 15, 2022, the Company sent a notice to TradeStation terminating the TS Merger Agreement pursuant to Section 12.01(b) thereof.

The foregoing description of the TS Merger Agreement is subject to and qualified in its entirety by reference to the full text of the TS Merger Agreement, as amended, which is included as Exhibit 2.1 to the Company's current report on Form 8-K filed with the SEC on November 10, 2021, Exhibit 2.1 to the Company's current report on Form 8-K filed with the SEC on December 20, 2021 and Exhibit 2.1 to the Company's current report on Form 8-K filed with the SEC on April 28, 2022.

Item 7.01. Regulation FD Disclosure.

On November 16, 2022, the Company and AtlasClear issued a joint press release announcing the execution of the Business Combination Agreement. A copy of the press release is attached hereto as Exhibit 99.1 and is incorporated herein by reference.

The information in this Item 7.01, including Exhibit 99.1, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Exchange Act, or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of the Company under the Securities Act or the Exchange Act, regardless of any general incorporation language in such filings. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information of the information in this Item 7.01, including Exhibit 99.1.

Additional Information and Where to Find It

In connection with the proposed Business Combination and related transactions contemplated in connection therewith (the "Proposed Transaction"), New Pubco intends to file with the SEC a registration statement on Form S-4 (the "Registration Statement") containing a preliminary proxy statement of the Company and prospectus of New Pubco. After the Registration Statement is declared effective, the Company will mail a definitive proxy statement/prospectus relating to the Proposed Transaction to its stockholders. Company stockholders and other interested persons are advised to read, when available, the preliminary proxy statement/prospectus included in the Registration Statement and the amendments thereto and the definitive proxy statement/ prospectus and other documents filed in connection with the Proposed Transaction, as these materials will contain important information about New Pubco, the Company, AtlasClear, WDCO, CB and the Proposed Transaction. When available, the definitive proxy statement/ prospectus will be mailed to Company stockholders as of a record date to be established for voting on the Proposed Transaction.

The Company's stockholders will also be able to obtain copies of the preliminary proxy statement/prospectus, definitive proxy statement/prospectus and other documents filed with the SEC, without charge, once available, at the SEC's website at www.sec.gov, or by directing a request to: Quantum FinTech Acquisition Corporation, 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607, Attention: Investor Relations or by email at IR@qftacorp.com.

Participants in Solicitation

The Company, AtlasClear, and their respective directors and executive officers may be deemed participants in the solicitation of proxies from the Company stockholders with respect to the Proposed Transaction. Company stockholders and other interested persons may obtain, without charge, more detailed information regarding the directors and officers of the Company in the Company's Annual Report on Form 10-K, filed with the SEC on March 10, 2022, as amended, which is available free of charge at the SEC's website at www.sec.gov. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to Company stockholders in connection with the Proposed Transaction and other matters to be voted upon at its special meeting of stockholders will be set forth in the proxy statement/prospectus for the Proposed Transaction when available. Additional information regarding the interests of participants in the solicitation of proxies from the Company's stockholders with respect to the Proposed Transaction when available.

4

Cautionary Statement Regarding Forward-Looking Statements

This Current Report on Form 8-K contains, and certain oral statements made by representatives of the Company and AtlasClear and their respective affiliates from time to time may contain, forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, as amended, that reflect the Company's and AtlasClear's current views with respect to, among other things, the future operations and financial performance of the Company, AtlasClear and the combined company. Forward-looking statements in this Current Report on Form 8-K or such oral statements, may be identified by the use of words such as "anticipate," "assume," "believe," "continue," "could," "estimate," "expect," "foreseeable," "future," "intend," "may," "outlook," "plan," "potential," "proposed" "predict," "project," "seek," "should," "target," "trends," "will," "would" and similar terms and phrases . These forwardlooking statements include, but are not limited to, statements as to (i) expectations regarding the Proposed Transaction, including timing for its consummation, (ii) anticipated use of proceeds from the transaction, (iii) the Company's and AtlasClear's expectations as to various operational results and market conditions, (iv) AtlasClear's anticipated growth strategy, including the proposed acquisitions, (v) anticipated benefits of the Proposed Transaction and proposed acquisitions, (vi) the financial technology of the combined entity and (vii) expected listing of the combined company. The forward-looking statements contained in this Current Report on Form 8-K are based on the current expectations of the Company, AtlasClear and their respective management and are subject to risks and uncertainties. No assurance can be given that future developments affecting the Company, AtlasClear or the combined company will be those that are anticipated. Actual results may differ materially from current expectations due to changes in global, regional or local economic, business, competitive, market, regulatory and other factors, many of which are beyond the control of the Company and AtlasClear. Should one or more of these risks or uncertainties materialize, or should any of the assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. Factors that could cause actual results to differ may emerge from time to time, and it is not possible to predict all of them.

Such factors include, but are not limited to: the risk that the transaction may not be completed in a timely manner or at all; the risk that the transaction closes but AtlasClear's acquisition of CB does not close as a result of the failure to satisfy the conditions to closing such acquisition (including, without limitation, the receipt of approval of CB's stockholders and receipt of required regulatory approvals), the failure to obtain requisite approval for the transaction or meet other closing conditions; the occurrence of any event, change or other circumstances that could give rise to the termination of the definitive agreement in respect of the transaction; failure to achieve sufficient cash available (taking into account all available financing sources) following any redemptions of the Company's public stockholders; failure to obtain the requisite approval of the Company's stockholders; failure to meet relevant listing standards in connection with the consummation of the transaction; failure to recognize the anticipated benefits of the transaction, which may be affected by, among other things, competition, the ability of the combined entity to maintain relationships with customers and suppliers and strategic alliance third parties, and to retain its management and key employees; potential litigation relating to the proposed transaction; changes to the proposed structure of the transaction that may be required or appropriate as a result of the announcement and execution of the transaction; unexpected costs and expenses related to the transaction; estimates of AtlasClear and the combined company's financial performance being materially incorrect predictions; AtlasClear's failure to complete the proposed acquisitions on favorable terms to AtlasClear or at all; AtlasClear's inability to integrate, and to realize the benefits of, the proposed acquisitions; changes in general economic or political conditions; changes in the markets that AtlasClear targets or the combined company will target; slowdowns in securities or cryptocurrency trading or shifting demand for trading, clearing and settling financial products; the impact of the ongoing COVID-19 pandemic; any change in laws applicable to the Company or AtlasClear or any regulatory or judicial interpretation thereof; and other factors, risks and uncertainties, including those to be included under the heading "Risk Factors" in the proxy statement/ prospectus to be later filed with the SEC, and those included under the heading "Risk Factors" in the Company's 2021 Form 10-K and its subsequent filings with the SEC. The Company and AtlasClear caution that the foregoing list of factors is not exhaustive. Any forwardlooking statement made in this Current Report on Form 8-K speaks only as of the date hereof. Plans, intentions or expectations disclosed in forward-looking statements may not be achieved and no one should place undue reliance on such forward-looking statements. Neither the Company nor AtlasClear undertake any obligation to update, revise or review any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by any applicable securities laws.

No Offer or Solicitation

This Current Report on Form 8-K shall not constitute a "solicitation" as defined in Section 14 of the Securities Exchange Act of 1934, as amended. This Current Report on Form 8-K does not constitute an offer, or a solicitation of an offer, to buy or sell any securities, investment or other specific product, or a solicitation of any vote or approval, nor shall there be any offer, solicitation or sale of securities, investment or other specific product in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to the registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act, or an exemption therefrom.

5

No Assurances

There can be no assurance that the proposed Business Combination will be completed, nor can there be any assurance, if the proposed Business Combination is completed, that the potential benefits of combining the companies will be realized.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
2.1	Business Combination Agreement, dated November 16, 2022, by and among Quantum FinTech Acquisition Corporation, Calculator New Pubco, Inc., Calculator Merger Sub 1, Inc., Calculator Merger Sub 2, Inc., AtlasClear, Inc., Atlas FinTech Holdings Corp., and Robert McBey.
10.1	Purchaser Support Agreement, dated November 16, 2022, by and among Quantum Ventures LLC, Quantum FinTech Acquisition Corporation, AtlasClear, Inc. and certain directors and officers of Quantum FinTech Acquisition Corporation party thereto.
10.2	<u>Registration Rights and Lock-Up Agreement, dated November 16, 2022, by and among Calculator New Pubco, Inc. and the</u> stockholders of AtlasClear and Quantum FinTech Acquisition Corporation party thereto.
10.3	Agreement and Plan of Merger, dated November 16, 2022, by and among AtlasClear, Inc. and Commercial Bancorp and, with respect to Section 6.16 only, Calculator New Pubco, Inc.
99.1	Joint Press Release dated November 16, 2022.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

6

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

QUANTUM FINTECH ACQUISITION CORPORATION

By: /s/ John Schaible

Name: John Schaible Title: Chief Executive Officer

Dated: November 16, 2022

7

Exhibit 2.1

Page

EXECUTION VERSION

BUSINESS COMBINATION AGREEMENT

by and among

QUANTUM FINTECH ACQUISITION CORPORATION, as Purchaser,

CALCULATOR NEW PUBCO, INC., as New Pubco,

CALCULATOR MERGER SUB 1, INC., as Merger Sub 1,

CALCULATOR MERGER SUB 2, INC., as Merger Sub 2,

ATLASCLEAR, INC., as the Company,

ATLAS FINTECH HOLDINGS CORP., as a Company Stockholder,

and

ROBERT MCBEY, as a Company Stockholder

Dated as of November 16, 2022

TABLE OF CONTENTS

	ARTICLE I.	
	MERGER	3
1.1	Mergers	3
1.2	Effective Times	3
1.3	Effect of the Mergers	4
1.4	Tax Treatment	5
1.5	Governing Documents	6
1.6	Directors and Officers of the Surviving Corporations	6
1.7	No Exercise of Appraisal Rights; Waivers	6
1.8	Exchange Procedures	7
1.9	New Pubco Charter and Bylaws	7
1.10	Taking of Necessary Action; Further Action	7

Copyright © 2022 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

1.11	Earn-Outs	8
1.12	Withholding Rights	10
	ARTICLE II.	
	CLOSING	11

2.1	Closing	11
2.2	Transaction Expenses	11
2.3	Closing Transactions	11
2.4	Allocation Schedule	12

ARTICLE III.

	REPRESENTATIONS AND WARRANTIES OF THE PURCHASER PARTIES	12
3.1	Organization and Standing	12
3.2	Authorization; Binding Agreement	13
3.3	Governmental Approvals	13
3.4	Non-Contravention	13
3.5	Capitalization.	14
3.6	SEC Filings and Purchaser Financials	15
3.7	Absence of Certain Changes	17
3.8	Compliance with Laws	17
3.9	Actions; Orders; Permits	17
3.10	Taxes and Returns'	17
3.11	Employees and Employee Benefit Plans	18
3.12	Properties	19
3.13	Material Contracts.	19
3.14	Litigation	19
3.15	Transactions with Affiliates	19
3.16	Business Activities	19
3.17	Finders and Brokers	20
3.18	Ownership of Merger Consideration	20
3.19	Certain Business Practices.	20
3.20	Lock-Up Agreements	20
3.21	Purchaser Trust Account	20
3.22	Independent Investigation	21

•
1
1

	ARTICLE IV.	
	REPRESENTATIONS AND WARRANTIES OF THE COMPANY AND THE COMPANY STOCKHOLDERS	21
4.1	Organization and Standing	21
4.2	Authorization; Binding Agreement	22
4.3	Capitalization.	22
4.4	Subsidiaries.	23
4.5	Governmental Approvals	23
4.6	Non-Contravention	23
4.7	Financial Statements.	24
4.8	Absence of Certain Changes	25
4.9	Compliance with Laws	25
4.10	Company Permits	25
4.11	Litigation	25
4.12	Material Contracts.	26
4.13	Intellectual Property.	27

4.14	Taxes and Returns.	31
4.15	Real Property	32
4.16	Personal Property	33
4.17	Title to and Sufficiency of Assets	33
4.18	Employee Matters.	34
4.19	Benefit Plans.	35
4.20	Environmental Matters	37
4.21	Transactions with Related Persons	38
4.22	Insurance	38
4.23	Books and Records	38
4.24	Certain Business Practices.	38
4.25	Compliance with Privacy Laws, Privacy Policies and Certain Contracts	39
4.26	Compliance with Broker Dealer Laws and Regulations	40
4.27	Finders and Brokers	41
4.28	Independent Investigation	41
4.29	Information Supplied	41

ARTICLE V. REPRESENTATIONS AND WARRANTIES OF THE STOCKHOLDERS 42 Organization and Standing 42 Authorization; Binding Agreement 42 Title to Equity Interests 42 Governmental Approvals 42 Non-Contravention 43 Finders and Brokers 43

5.1

5.2

5.3

5.4

5.5

5.6

ii

	ARTICLE VI.	
	COVENANTS	43
6.1	Access and Information.	43
6.2	Conduct of Business of the Company.	44
6.3	Conduct of Business of Purchaser.	47
6.4	Annual and Interim Financial Statements	50
6.5	Purchaser Public Filings	50
6.6	No Solicitation.	51
6.7	No Trading	51
6.8	Notification of Certain Matters	52
6.9	Efforts.	52
6.10	Tax Matters	54
6.11	Further Assurances	55
6.12	The Registration Statement/ Proxy Statement.	55
6.13	Company Stockholders Written Consent	57
6.14	Public Announcements	57
6.15	Confidential Information	58
6.16	Post-Closing Board of Directors and Officers; Employment Agreements.	58
6.17	Indemnification of Directors and Officers; Tail Insurance.	58
6.18	Trust Account Proceeds	59
6.19	Financing	59
6.20	Pacsquare Acquisition Agreement	60

ARTICLE VII.

NO SURVIVAL

7.1	No Survival	60
	ARTICLE VIII.	
	CLOSING CONDITIONS	60
8.1	Conditions to Each Party's Obligations	60
8.2	Conditions to Obligations of the Company	61
8.3	Conditions to Obligations of Purchaser Parties	63
8.4	Frustration of Conditions	64
	ARTICLE IX.	
	TERMINATION AND EXPENSES	65
9.1	Termination	65
9.2	Effect of Termination	66
9.3	Fees and Expenses	66
	ARTICLE X.	
	WAIVERS AND RELEASES	66
10.1	Waiver of Claims Against Trust	66

iii

		ARTICLE XI.	
		MISCELLANEOUS	67
11.1	Notices		67
11.2	Binding Effect; Assignment		67
11.3	Third Parties		67
11.4	Governing Law; Jurisdiction		67
11.5	WAIVER OF JURY TRIAL		68
11.6	Specific Performance		68
11.7	Severability		68
11.8	Amendment		68
11.9	Waiver		69
11.10	Entire Agreement		69
11.11	Interpretation		69
11.12	Counterparts		70
11.13	Legal Representation		70
		ARTICLE XII.	
		DEFINITIONS	71
12.1	Certain Definitions		71
12.2	Section References		84

INDEX OF EXHIBITS

Exhibit	Description
Exhibit A	Form of Purchaser Support Agreement
Exhibit B	Form of Registration Rights and Lock-Up Agreement
Exhibit C	Broker-Dealer Acquisition Agreement
Exhibit D	Bank Acquisition Agreement
Exhibit E	Contribution Agreement

v

BUSINESS COMBINATION AGREEMENT

This Business Combination Agreement (this "Agreement") is made and entered into as of November 16, 2022 by and among (i) Quantum FinTech Acquisition Corporation, a company incorporated in Delaware (together with its successors, the "Purchaser"), (ii) Calculator New Pubco, Inc., a Delaware corporation and a wholly-owned Subsidiary of Purchaser ("New Pubco"), (iii) Calculator Merger Sub 1, Inc., a Delaware corporation and a wholly-owned Subsidiary of New Pubco ("Merger Sub 1"), (iv) Calculator Merger Sub 2, Inc., a Delaware corporation and a wholly-owned Subsidiary of New Pubco ("Merger Sub 1"), (iv) Calculator Merger Sub 2, Inc., a Delaware corporation and a wholly-owned Subsidiary of New Pubco ("Merger Sub 2") (v) AtlasClear, Inc., a Wyoming corporation (the "Company"), (vi) Atlas FinTech Holdings Corp, a Delaware corporation ("Atlas FinTech") and (vii) Robert McBey ("McBey", and together with Atlas FinTech, each a "Company Stockholders"). Purchaser, New Pubco, Merger Sub 1, Merger Sub 2, the Company and the Company Stockholders are sometimes referred to herein individually as a "Party" and, collectively, as the "Parties."

RECITALS:

A. The Company, indirectly through its Subsidiaries, at and following the Second Effective Time will be engaged in the business of providing specialized banking and clearing services to other financial services firms;

B. New Pubco is a newly incorporated Delaware corporation, wholly owned by Purchaser, and was formed for the purpose of acquiring the Company;

C. Each of Merger Sub 1 and Merger Sub 2 is a newly incorporated corporation, wholly owned by New Pubco, Merger Sub 1 was formed for the sole purpose of effectuating the First Merger and Merger Sub 2 was formed for the sole purposes of effectuating the Second Merger;

D. The Parties intend to effect a business combination transaction whereby (a) Purchaser will merge with and into Merger Sub 1 (the "*First Merger*"), with Purchaser being the surviving entity and becoming a wholly owned subsidiary of New Pubco, and (b) following the First Merger, Merger Sub 2 will merge with and into the Company (the "*Second Merger*" and together with the First Merger, the "*Mergers*"), with the Company being the surviving entity and becoming a wholly owned subsidiary of New Pubco, each Merger to occur upon the terms and subject to the conditions set forth in this Agreement and in accordance with the applicable provisions of the DGCL (with respect to the First Merger) and the WBCA (with respect to the Second Merger);

E. The Special Committee of the board of directors of Purchaser (the "Special Committee") and the board of directors of Purchaser have each (i) determined that the Mergers are fair, advisable and in the best interests of Purchaser and its Public Stockholders, (ii) approved this Agreement and the Contemplated Transactions, including the Mergers, upon the terms and subject to the conditions set forth herein, and (iii) determined to recommend to Purchaser's stockholders the approval and adoption of this Agreement and the Contemplated Transactions hereby, including the Mergers;

F. The boards of directors of New Pubco and Merger Sub 1 have each (i) determined that the First Merger is fair, advisable and in the best interests of their respective companies and stockholders, (ii) approved this Agreement and the Contemplated Transactions, including the First Merger, upon the terms and subject to the conditions set forth herein, and (iii) determined to recommend to their respective stockholders the approval and adoption of this Agreement and the Contemplated Transactions, including the First Merger;

G. The boards of directors of the Company, New Pubco and Merger Sub 2 have each (i) determined that the Second Merger is fair, advisable and in the best interests of their respective companies and stockholders, (ii) approved this Agreement and the Contemplated Transactions, including the Second Merger, upon the terms and subject to the conditions set forth herein, and (iii) determined to recommend to their respective stockholders the approval and adoption of this Agreement and the Contemplated Transactions, including the Second Merger;

H. Contemporaneously with the execution of this Agreement, the Company Stockholders have executed, and the Company has delivered to Purchaser, a unanimous written consent of the Company Stockholders approving and adopting this Agreement, the Second Merger and the Contemplated Transactions (the "*Company Stockholders Written Consent*");

I. Contemporaneously with the execution of, and as a condition and an inducement to Purchaser and the Company entering into this Agreement, specified stockholders of Purchaser, including the Sponsors, are entering into and delivering Support Agreements, substantially in the form attached hereto as <u>Exhibit A</u> (each, a "*Purchaser Support Agreement*"), pursuant to which each such Purchaser stockholder has agreed (x) not to transfer or redeem any shares of Purchaser Common Stock held by such Purchaser stockholder, (y) to vote in favor of this Agreement and the Mergers at Purchaser Special Meeting and (z) waive any anti-dilution or similar protection, if any, with respect to Purchaser Common Stock (resulting from the Contemplated Transactions);

J. Simultaneously with the execution and delivery of this Agreement, New Pubco and all of the Company Stockholders as of the date hereof have each entered into a Registration Rights and Lock-Up Agreement with the Company and Purchaser, the form of which is attached as <u>Exhibit B</u> hereto (each, a "*Registration Rights and Lock-Up Agreement*"), which will become effective as of the Closing;

K. The Parties intend that, for U.S. federal income tax purposes, that, taken together, the Mergers will qualify as an exchange governed by the provisions of Section 351 of the Code, and this Agreement constitute and hereby is adopted as a "plan of reorganization" within the meaning of Treasury Regulations Sections 1.368-2(g) and 1.368-3(a) (collectively, the "*Intended Tax Treatment*")

L. On April 15, 2022, the Company entered into that certain Stock Purchase Agreement, as amended by that certain Amendment to Stock Purchase Agreement dated June 15, 2022, and by that certain Amendment No. 2 to the Stock Purchase Agreement dated November 16, 2022, copies of which are attached hereto as <u>Exhibit C</u> (as so amended, the "*Broker-Dealer Acquisition Agreement*"), with Wilson-Davis & Co., Inc., a Utah corporation, and its selling shareholders (the "*Target Broker-Dealer*") pursuant to which, among other things, the Company has agreed to acquire all of the issued and outstanding shares of the Target Broker-Dealer in a transaction that shall be consummated immediately prior to the Second Effective Time and in consideration therefore, the Company will pay the shareholders of the Target Broker Dealer the Purchase Price (as defined under the Broker-Dealer Acquisition Agreement) (the "*Broker Dealer Acquisition Purchase Price*");

M. On November 16, 2022, the Company entered into that certain Agreement and Plan of Merger, copy of which is attached hereto as <u>Exhibit D</u> (the "*Bank Acquisition Agreement*"), with Commercial Bancorp, a Wyoming corporation (the "*Target Bank*") pursuant to which, among other things, the Target Bank will be merged with and into a Subsidiary of the Company whereby the separate corporate existence of such Subsidiary will cease, and the Target Bank will continue as the surviving entity of such merger in a transaction that shall be consummated after the Second Effective Time and in consideration therefore, the Company will pay the shareholders of the Target Bank the Merger Consideration (as defined under the Bank Acquisition Agreement), including, for the avoidance of doubt, any gross-up for Taxes paid to the stockholders of the Target Bank pursuant to the Bank Acquisition Agreement (the "*Bank Acquisition Purchase Price*");

2

N. Prior to the Closing, pursuant to that certain Assignment and Assumption Agreement and Bill of Sale (the "*Contribution Agreement*"), dated as of the date hereof, by and among the Company, Atlas FinTech and Atlas Financial Technologies Corp., a copy of which is attached hereto as <u>Exhibit E</u>, Atlas FinTech will contribute to the Company all rights, title and interest in the Transferred Intellectual Property (as defined in the Contribution Agreement) (the "*Contributed Assets*"); and

O. Certain capitalized terms used herein are defined in Article XII hereof.

NOW, THEREFORE, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and the representations, warranties, covenants and agreements contained in this Agreement, and intending to be legally bound hereby, the Parties hereto agree as follows:

ARTICLE I. MERGER

1.1 Mergers.

(a) At the First Effective Time (as defined below), on the terms and subject to the conditions set forth herein and in accordance with the applicable provisions of the DGCL, Purchaser and Merger Sub 1 shall consummate the First Merger, pursuant to which Merger Sub 1 shall be merged with and into Purchaser, Purchaser's stockholders shall receive shares of New Pubco Common Stock in exchange for their shares of Purchaser Common Stock and New Pubco Warrants in exchange for their Purchaser Warrants, and the separate existence of Merger Sub 1 shall cease and Purchaser shall continue as the surviving corporation after the First Merger as a direct, wholly-owned Subsidiary of New Pubco. Purchaser, as the surviving corporation after the First Merger, is hereinafter sometimes referred to as the "*First Surviving Corporation*" (provided that references to Purchaser for periods after the First Effective Time shall include the Surviving Corporation).

(b) At the Second Effective Time (as defined below), and subject to and upon the terms and conditions of this Agreement, and in accordance with the applicable provisions of the Wyoming Business Corporations Act (the "*WBCA*"), Merger Sub 2 and the Company shall consummate the Second Merger, pursuant to which Merger Sub 2 shall be merged with and into the Company, following which the separate corporate existence of Merger Sub 2 shall cease and the Company shall continue as the surviving corporation. The Company, as the surviving corporation after the Second Merger, is hereinafter sometimes referred to as the "*Second Surviving Corporation*" (provided, that references to the Company for periods after the Second Effective Time shall include the Second Surviving Corporation).

1.2 Effective Times.

(a) On the terms and subject to the conditions set forth herein, on the Closing Date, Purchaser and Merger Sub 1 shall cause the First Merger to be consummated by filing a certificate of merger in a form and substance to be agreed upon by the Parties (the "*First Certificate of Merger*") with the Secretary of State of the State of Delaware in accordance with the applicable provisions of the DGCL, and the time of such filing, or such later time as may be agreed in writing by the Company and Purchaser and specified in the First Certificate of Merger, will be the effective time of and constitute the consummation of the First Merger (the "*First Effective Time*").

3

(b) On the terms and subject to the conditions set forth herein, on the Closing Date, but after the First Effective Time, the Company and Merger Sub 2 shall cause the Second Merger to be consummated by filing a certificate of merger in a form and substance to be agreed upon by the Parties (the "*Second Certificate of Merger*") with the Wyoming Secretary of State in accordance with the applicable provisions of the WBCA, and the time of such filing, or such later time as may be agreed in writing by the Company and Purchaser and specified in the Second Certificate of Merger, will be the effective time of and constitute the consummation of the Second Merger (the "*Second Effective Time*").

1.3 Effect of the Mergers.

(a) <u>The First Merger</u>. On the terms and subject to the conditions set forth herein, at the First Effective Time, by virtue of the First Merger and without any further action on the part of any Party or the holders of any securities of Purchaser, the following shall occur:

(i) Immediately prior to the First Effective Time, every issued and outstanding Purchaser Unit shall be automatically separated, to the extent Purchaser Units have not already separated, and the holder thereof shall be deemed to hold one share of Purchaser Common Stock and one Purchaser Warrant to purchase one-half of a share of Purchase Common Stock in accordance with the terms of the applicable Purchaser Unit, which underlying securities shall be converted in accordance with this <u>Section 1.3</u>.

(ii) Each share of Purchaser Common Stock issued and outstanding immediately prior to the First Effective Time shall be automatically cancelled and extinguished and converted into one share of New Pubco Common Stock, following which, all shares of Purchaser Common Stock shall cease to be outstanding and shall automatically be canceled and extinguished and shall cease to exist by virtue of the First Merger. The holders of Purchaser Common Stock outstanding immediately prior to the First Effective Time shall cease to have any rights with respect to such shares, except as expressly provided herein or by Law. Each certificate previously evidencing shares of Purchaser Common Stock, if any, shall thereafter represent only the right to receive the same number of shares of New Pubco Common Stock and shall be exchanged for a certificate (if requested) representing the same number of shares of New Pubco Common Stock upon the surrender of such certificate in accordance with Section 1.8.

(iii) Each Purchaser Private Warrant issued and outstanding immediately prior to the First Effective Time will automatically cease to be a warrant with respect to Purchaser Common Stock and will be assumed by New Pubco and converted into a New Pubco Private Warrant exercisable for one share of New Pubco Common Stock, and each Purchaser Public Warrant issued and outstanding immediately prior to the First Effective Time will automatically cease to be a warrant with respect to Purchaser Common Stock and will be assumed by New Pubco and converted into a New Pubco Public Warrant exercisable for one-half of a share of New Pubco Common Stock, in each case at the same exercise price per share and on the same terms in effect immediately prior to the First Effective Time, and the rights and obligations of Purchaser under the Warrant Agreement will be irrevocably assigned to, and assumed by, New Pubco.

(iv) If there are any shares of Purchaser that are owned by Purchaser or any Subsidiary of Purchaser, or that are held as treasury shares, in each case as of the First Effective Time, such shares shall automatically be canceled, retired and extinguished and shall cease to exist, without any conversion thereof or payment therefor.

4

(v) If any certificate for securities of Purchaser is to be issued in a name other than that in which the certificate surrendered in exchange therefor is registered, it shall be a condition of the issuance thereof that the certificate so surrendered shall be properly endorsed (or accompanied by an appropriate instrument of transfer) and otherwise in proper form for transfer and that the Person requesting such exchange shall have paid to New Pubco or any agent designated by it any transfer or other Taxes required by reason of the issuance of a certificate or securities of New Pubco in any name other than that of the registered holder of the certificate surrendered, or established to the satisfaction of New Pubco or any agent designated by it that such Tax has been paid or is not payable.

(vi) Without limiting the generality of the foregoing, and subject thereto, at the First Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of Purchaser and Merger Sub 1 shall become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the First Surviving Corporation.

(b) <u>The Second Merger</u>. On the terms and subject to the conditions set forth herein, at the Second Effective Time (except as otherwise set forth below), by virtue of the Second Merger and without any further action on the part of any Party or any other Person, the following shall occur:

(i) Each share of Company Stock issued and outstanding immediately prior to the Second Effective Time will be automatically cancelled and extinguished and converted (the "*Conversion*") into the right to receive the Per Share Merger Consideration. At the Second Effective Time, if there is any Company Stock that are owned by the Company as treasury shares or any Company Stock owned by any direct or indirect Subsidiary of the Company immediately prior to the Effective Time, such Company Stock shall be canceled and shall cease to exist without any conversion thereof or payment therefor.

(ii) Without limiting the generality of the foregoing, and subject thereto, at the Second Effective Time, all the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the Company and Merger Sub 2 shall become the property, rights, privileges, agreements, powers and franchises, debts, liabilities, duties and obligations of the Second Surviving Corporation.

(c) On the terms and subject to the conditions set forth herein, (i) at the First Effective Time, by virtue of the First Merger and without any action on the part of any holder thereof, all of the shares of New Pubco issued and outstanding immediately prior to the First Effective Time shall be canceled and extinguished without any conversion thereof or payment therefor; and (ii) at the Second Effective Time, by virtue of the Second Merger and without any action on the part of any holder thereof, each share of New Pubco Common Stock issued and outstanding immediately prior to the Second Effective Time shall remain outstanding.

1.4 Tax Treatment. For U.S. federal income tax purposes, taken together, the Mergers will qualify as an exchange governed by the provisions of Section 351 of the Code. The Parties adopt this Agreement as a "plan of reorganization" within the meaning of Sections 1.368-2(g) and 1.368-3(a) of the U.S. Treasury Regulations.

5

1.5 Governing Documents.

(a) At the First Effective Time, the certificate of incorporation and bylaws of Purchaser shall be automatically be amended and restated in their entirety in a form and substance to be agreed upon by the Parties, and such amended and restated certificate of incorporation and bylaws shall become the respective certificate of incorporation and bylaws of the First Surviving Corporation from and after the First Effective Time until thereafter amended in accordance with their respective terms and as provided by applicable Law.

(b) At the Second Effective Time, the certificate of incorporation and bylaws of the Company shall be automatically be amended and restated in their entirety in a form and substance to be agreed upon by the Parties, and such amended and restated certificate of incorporation and bylaws shall become the respective certificate of incorporation and bylaws of the Second Surviving Corporation from and after the Second Effective Time until thereafter amended in accordance with their respective terms and as provided by applicable Law.

1.6 Directors and Officers of the Surviving Corporations.

(a) The directors and officers of Purchaser as of immediately prior to the First Effective Time shall be the initial directors and officers of the First Surviving Corporation and shall hold office in accordance with the certificate of incorporation and bylaws of the First Surviving Corporation until their successors are duly elected or appointed and qualified or until their earlier death, resignation or removal.

(b) New Pubco shall take all actions necessary or appropriate (including securing resignations or removals and making such appointments as are necessary) to cause, effective as of the Second Effective Time, the board of directors and officers of the Second Surviving Corporation to consist of the persons contemplated by <u>Section 6.16</u> until their successors are duly elected or appointed and qualified or until their earlier death, resignation or removal. On the Closing Date, the Second Surviving Corporation shall enter into customary indemnification agreements in the form to be mutually agreed by New Pubco and the Company with such individuals elected as members of the board of directors, or appointed as officers, of the Second Surviving Corporation as of the Closing, which indemnification agreements shall continue to be effective immediately following the Closing.

1.7 No Exercise of Appraisal Rights; Waivers(a). In connection with the Contemplated Transactions, each Company Stockholder hereby expressly (a) waives, to the extent permitted under applicable Law, the applicability of the provisions for dissenters' or appraisal rights set forth in Section 17-18-308 of the WBCA (or any other similar applicable state Law), with respect to any Company Stock, (b) agrees that such Company Stockholder will not, under any circumstances in connection with the Contemplated Transactions, exercise any dissenters' or appraisal rights in respect of any Company Stock, and (c) agrees that such Company Stockholder will not bring, commence, institute, maintain, prosecute, participate in or voluntarily aid any action, claim, suit or cause of action, in law or in equity, in any court or before any Governmental Authority, which (i) challenges the validity of or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this Agreement by such Company Stockholder, or the approval of the Agreement by the Company Stockholder may defend against, contest or settle any such action, claim, suit or cause of action brought against such Company Stockholder that relates solely to such Company Stockholder's capacity as a director, officer or securityholder of the Company.

1.8 Exchange Procedures.

(a) Prior to the First Effective Time, Purchaser and New Pubco shall appoint as New Pubco's transfer agent, Continental Stock Transfer & Trust Company, or another agent reasonably acceptable to the Company (the "*Exchange Agent*"), for the purpose of disbursing the Merger Consideration. At the Second Effective Time, New Pubco will promptly issue and allot, credited as fully paid, or cause to be issued and allotted, credited as fully paid, to the Company Stockholders (and New Pubco will direct the Exchange Agent to take all necessary action to record and effect the same) the number of shares of New Pubco Common Stock equal to Merger Consideration in accordance with the Allocation Schedule, in each case in book-entry form (which shall have a customary Securities Act restrictive legend), unless otherwise reasonably requested by New Pubco.

(b) If any portion of the Merger Consideration is to be delivered or issued to a Person other than the Person in whose name the relevant Company Stock was registered on the books and records of the Company immediately prior to the Second Effective Time, it shall be a condition to such delivery that (i) the transfer of such Company Stock shall have been permitted in accordance with the terms of the Company's Organizational Documents and any stockholders agreement with respect to the Company, each as in effect immediately prior to the Second Effective Time, (ii) the certificate(s) representing such Company Stock ("*Company Certificate(s)*") shall be properly endorsed or shall otherwise be in proper form for transfer (or, in the event that such certificate(s) shall have been lost, stolen or destroyed, the Company Stockholder may instead deliver an affidavit of lost certificate and indemnity of loss in form and substance reasonably acceptable to New Pubco) and, (iii) the recipient of such portion of the Merger Consideration, or the Person in whose name such portion of the Merger Consideration is delivered counterparts to a Registration Rights and Lock-Up Agreement, and such other documents as are reasonably deemed necessary by the Exchange Agent or New Pubco and (iv) the Person requesting such delivery shall pay to the Exchange Agent any transfer or other Taxes required as a result of such delivery to a Person other than the registered holder of such Company Certificate or establish to the satisfaction of the Exchange Agent that such Tax has been paid or is not payable.

(c) All securities issued upon the surrender of Company Stock in accordance with the terms hereof shall be deemed to have been issued in full satisfaction of all rights pertaining to such Company Stock.

(d) Notwithstanding anything to the contrary contained herein, no fraction of a share of New Pubco Common Stock will be issued by virtue of the Mergers or the Contemplated Transactions, and each Person who would otherwise be entitled to a fraction of a share of New Pubco Common Stock (after aggregating all fractional shares of New Pubco Common Stock that otherwise would be received by such holder) shall instead have the number of shares of New Pubco Common Stock issued to such Person rounded down in the aggregate to the nearest whole share of New Pubco Common Stock.

1.9 New Pubco Charter and Bylaws. Upon the Second Effective Time, New Pubco shall adopt the New Pubco Charter and the New Pubco Bylaws, which shall, among other matters, (a) provide that the name of New Pubco shall be changed to "AtlasClear Holdings, Inc." or such other name as mutually agreed to by the Parties, and (b) provide for size and structure of the Newco Pubco Board in accordance with <u>Section 6.16</u>.

1.10 Taking of Necessary Action; Further Action. If, at any time after the First Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the First Surviving Corporation with full right, title and possession to all assets, property, rights, privileges, powers and franchises of Purchaser and Merger Sub 1, the officers and directors of Purchaser and Merger Sub 1 are fully authorized in the name of their respective corporations or otherwise to take, and will take, all such lawful and necessary action, so long as such action is not inconsistent with this Agreement. If, at any time after the Second Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Second Surviving Corporation with full right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company and Merger Sub 2, the officers and directors of the Company and Merger Sub 2 are fully authorized in the name of their respective corporations or otherwise to take, and will take, all such lawful and necessary action, so long as such action is not inconsistent with this Agreement.

1.11 Earn-Outs.

(a) Subject to the terms and conditions of this <u>Section 1.11</u>, following the Closing, and as additional consideration for the Second Merger, within five (5) Business Days after the occurrence of a Triggering Event, New Pubco shall issue to each Company Stockholder identified on the Allocation Schedule (the "*Share Price Earn Out Recipient*"), the following amount of Share Price Earn Out Shares (which shall have a customary Securities Act restrictive legend) in accordance with such Company Stockholder's respective Pro Rata Share:

(i) 2,333,333 of the Share Price Earn Out Shares upon the occurrence of Triggering Event I (the "*\$12.50 Earn Out Shares*");

(ii) 1,944,444 of the Share Price Earn Out Shares upon the occurrence of Triggering Event II (the "\$15.00 Earn Out Shares"); and

(iii) 1,666,667 of the Share Price Earn Out Shares upon the occurrence of Triggering Event III (the "\$17.50 Earn Out Shares").

For illustrative purposes, if, prior to the expiration of the Earn Out Period:

(i) the Share Price of New Pubco Common Stock is greater than or equal to \$12.50, all of the \$12.50 Earn Out Shares shall be issued in accordance with this <u>Section 1.11</u>;

(ii) the Share Price of New Pubco Common Stock is greater than or equal to \$15.00, all of the \$15.00 Earn Out Shares shall be issued in accordance with this <u>Section 1.11</u>; and

(iii) the Share Price of New Pubco Common Stock is greater than or equal to \$17.50, all of the \$17.50 Earn Out Shares shall be issued in accordance with this <u>Section 1.11</u>.

(b) Subject to the terms and conditions of this Section 1.11, following the Closing, and as additional consideration for the Second Merger, Atlas FinTech shall be entitled to receive (subject to the terms and conditions set forth herein) a number of Software Products Earn Out Shares based on the Software Products Gross Revenue in (i) the fiscal year ending December 31, 2023 (the "*First Software Product Earn Out Period*"), (ii) the fiscal year ending December 31, 2024 (the "*Second Software Product Earn Out Period*"), (iii) the fiscal year ending December 31, 2025 (the "*Third Software Product Earn Out Period*"), (iv) the fiscal year ending December 31, 2026 (the "*Fourth Software Product Earn Out Period*") and (v) the fiscal year ending December 31, 2027 (the "*Final Software Product Earn Out Period*", and, together with the First Software Product Earn Out Periods"). For the avoidance of doubt, if New Pubco fails to achieve the maximum potential Software Product Earn Out Periods.

(i) Within thirty (30) days following New Pubco's receipt of the consolidated audited financial statements of New Pubco and its Subsidiaries for each Software Product Earn Out Period, New PubCo will prepare and deliver, or cause to be prepared and delivered, to Atlas FinTech, at New PubCo's expense, a Software Products Earn Out Statement. Atlas FinTech will have the right to review, at New PubCo's expense, any of the applicable books, records and work papers relating to the preparation of the Software Products Earn Out Statement.

8

(ii) If Atlas FinTech disagrees with New PubCo's evaluation of the calculations as set forth in the Software Products Earn Out Statement, then Atlas FinTech shall, within thirty (30) days after receipt of the Software Products Earn Out Statement, deliver a written notice to New PubCo disagreeing with such evaluation and setting forth with reasonable specificity Atlas FinTech's evaluation of the Software Products Earn Out Statement and objections thereto (the "Software Products Earn Out Statement Objection Notice"). If Atlas Fintech does not provide a Software

Products Earn Out Statement Objection Notice within such thirty (30) day period, then the Parties shall be deemed to have accepted the Software Products Earn Out Statement and all calculations set forth therein, which shall be final, binding and conclusive upon the Parties. Upon receipt of a Software Products Earn Out Statement Objection Notice, New PubCo and Atlas FinTech shall, during the following thirty (30) days, use commercially reasonable efforts to reach an agreement on the disputed calculations.

(iii) If New PubCo and Atlas FinTech are unable to reach agreement, then they shall thereafter cause an independent, nationally recognized accountant (the "*Independent Accountant*") to promptly review the Software Products Earn Out Statement Objection Notice and resolve the disputed calculations therein. The Independent Accountant shall deliver to New PubCo and Atlas FinTech, as promptly as practicable, a certified written report setting forth the Independent Accountants' analysis and conclusions in accordance with the terms and conditions contained in this Section 1.11(b), which report shall be final and binding upon the parties (absent fraud or manifest error). New PubCo and Atlas FinTech shall endeavor in good faith to cause the Independent Accountant to deliver its certified written report within thirty (30) days after the date of submission of the dispute to the Independent Accountant. This review by the Independent Accountant shall be limited solely to the disputed calculations contained in the Software Products Earn Out Statement and shall not re-open any other aspect or undisputed figure of the calculations contained therein. The Independent Accountant shall not be permitted to attribute a value to any disputed amount greater than the greatest amount or less than the least amount proposed by either party in the case of a value that has the effect of increasing or decreasing the applicable issuance of the Software Products Earn Out Shares.

(iv) Responsibility for the fees, costs and expenses of the Independent Accountant shall be based on the inverse of the percentage that the Independent Accountant's determination (before such allocation) bears to the total amount of the total items in dispute as originally submitted to the Independent Accountant. The Independent Accountant's determination will be based solely on written presentations by New PubCo and Atlas FinTech that are in accordance with the guidelines and procedures set forth in this Section 1.11(b). (i.e., not on the basis of an independent review).

9

(c) If during a Share Price Earn Out Period and/or a Software Products Earn Out Period, a Company Sale occurs, then immediately prior to the consummation of such Company Sale, (i) any Share Price Earn Out Shares that have not previously been distributed to the Share Price Earn Out Recipients and/or Software Products Earn Out Shares that have not previously been distributed to Atlas FinTech (whether or not previously earned) shall be deemed earned (and the applicable Triggering Event achieved with respect to the Share Price Earn Out Shares), (ii) New Pubco shall issue the applicable Share Price Earn Out Shares to the Share Price Earn Out Recipients and the Software Products Earn Out Shares to Atlas FinTech and (iii) the holders of such Share Price Earn Out Shares and Software Products Earn Out Shares shall be eligible to participate in such Company Sale; *provided, however, that* with respect to the Share Price Earn Out Shares only, such shares shall be deemed earned (and the applicable Triggering Event achieved) only if implied the value per share of New Pubco Common Stock in the Company Sale at the time of first announcement of the Company Sale (as determined by New Pubco Board after including the Earn Out Shares in such determination) equals or exceeds \$12.50, \$15.00 or \$17.50, as applicable, as adjusted pursuant to Section 1.11(d) below.

(d) The price targets included in the definitions of Triggering Event I, Triggering Event II and Triggering Event II shall be equitably adjusted as appropriate to reflect any stock splits, reverse stock splits, stock dividends (including any dividend or distribution of securities convertible into New Pubco Common Stock), extraordinary cash dividend, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change or transaction with respect to New Pubco Common Stock.

(e) At all times during the Share Price Earn Out Period and a Software Products Earn Out Period, New Pubco shall keep available for issuance a sufficient number of shares of unissued New Pubco Common Stock to permit New Pubco to satisfy in full its issuance obligations set forth in this <u>Section 1.11</u> and shall take all actions reasonably required (including by convening any stockholder meeting) to increase the authorized number of New Pubco Common Stock if at any time there shall be insufficient unissued New Pubco Common Stock to permit such reservation.

(f) During the Share Price Earn Out Period and a Software Products Earn Out Periods, New Pubco shall use commercially reasonable efforts for New Pubco to remain listed as a public company on, and for New Pubco Common Stock

(including, when issued, the Earn Out Shares) to be tradable over the national securities exchange (as defined under Section 6 of the Exchange Act) on which the shares of New Pubco Common Stock are then listed.

(g) Any issuance of the Share Price Earn Out Shares and the Software Earn Out Shares, including any issuance of the Share Price Earn Out Shares and the Software Earn Out Shares made in connection with a Company Sale pursuant to <u>Section 1.11(c)</u>, shall be treated as an adjustment to the Merger Consideration by the Parties hereto for Tax purposes and not treated as "other property" within the meaning of Section 351(b) or Section 356 of the Code, unless otherwise required by applicable law as a result of a "determination" (within the meaning of Section 1313(a) of the Code).

1.12 Withholding Rights. Notwithstanding anything in this Agreement to the contrary, Purchaser, the Company, New Pubco, Merger Subs and the First Surviving Corporation and the Second Surviving Corporation and their respective Affiliates and agents shall be entitled to deduct and withhold from any amounts otherwise payable pursuant to this Agreement, any amounts that are required to be deducted or withheld with respect to the making of such payments under applicable Law. To the extent that any deduction or withholding is to be made from the amounts otherwise payable hereunder (other than any such value that is subject to withholding because it is properly treated as compensation for applicable Tax purposes), Purchaser shall use commercially reasonable efforts to provide notice to the applicable Person at least ten (10) days prior to the day the relevant withholding). To the extent that any amounts are deducted or withheld consistent with the terms of this <u>Section 1.12</u> and timely paid over to the appropriate Governmental Authority, such deducted or withholding was made. In the case of any such payment payable to employees of the Company or any of its Subsidiaries in connection with the Second Merger treated as compensation for applicable Tax purposes, the Parties shall cooperate to pay such amounts through Purchaser's, New Pubco's, or its Affiliate's payroll to facilitate applicable to withholding.

10

ARTICLE II. CLOSING

2.1 Closing. Subject to the satisfaction or waiver of the conditions set forth in <u>Article VIII</u>, the consummation of the transactions contemplated by this Agreement (the "*Closing*") shall take place electronically by the mutual exchange of electronic signatures (including portable document format (.PDF)), on a date and at a time to be agreed upon by Purchaser and the Company, which date shall be no later than the second (2nd) Business Day after all the Closing conditions to this Agreement have been satisfied or waived, or at such other date, or time or both as Purchaser and the Company may agree (the date and time at which the Closing is actually held being the "*Closing Date*").

2.2 Transaction Expenses. At least two (2) Business Days prior to the Closing Date, the Company shall deliver to Purchaser the Allocation Schedule, and at least five (5) Business Days prior to the Closing Date, the Company shall deliver to Purchaser a written statement (the "Company Closing Statement") setting forth its good faith estimate and calculation of the Company Transaction Expenses, including a detailed itemization of the components thereof and reasonable supporting documentation and detail therefor (including invoices), and determined pursuant to the definitions contained in this Agreement. From and after delivery of the Company Closing Statement and through the Closing Date, Purchaser shall have the right to review and comment on such calculations and estimates, the Company shall consider in good faith any such comments made by Purchaser, and the Company and Purchaser shall cooperate with each other through the Closing Date and use good faith efforts to resolve any differences regarding the calculations and estimates contained in the Company Closing Statement (and any updates or revisions as may be agreed by the Company and Purchaser shall be included in the Company Closing Statement). The Company shall, and shall cause its Representatives to, (x) reasonably cooperate with Purchaser and its Representatives to the extent related to Purchaser's review of the Company Closing Statement and the calculations and estimates contained therein (including engaging in good faith discussion related thereto) and (y) provide reasonable access to personnel, books, records and other information during normal business hours to the extent related to the preparation of the Company Closing Statement and reasonably requested by Purchaser or its Representatives in connection with such review; provided, that Purchaser shall not, and shall cause its Representatives to not, unreasonably interfere with the business of the Company and its Subsidiaries in connection with any such access.

2.3 Closing Transactions. Upon the terms and subject to the satisfaction or written waiver of the conditions contained in this Agreement, at the Closing, (a) Purchaser shall deliver, or caused to be delivered, to Seller all of the agreements, instruments, and

documents set forth in Section 8.2(d) in accordance with the terms thereof; and (b) the Company shall deliver, or caused to be delivered, to Seller all of the agreements, instruments, and documents set forth in Section 8.3(d) in accordance with the terms thereof.

2.4 Allocation Schedule. The Company and each Company Stockholder acknowledges and agrees that (a) the Merger Consideration, the Share Price Earn Out Shares are being allocated among Company Stockholders pursuant to the schedule in the form set forth on Schedule 2.4 and delivered by the Company to Purchaser at least two (2) Business Days prior to the anticipated Closing Date (the "Allocation Schedule") and such allocation (i) is and will be in accordance with the Organizational Documents of the Company and applicable Law, (b) does and will set forth (A) the mailing addresses and email addresses, for each Company Stockholder, (B) the number and class of equity securities owned by each Company Stockholders, (C) the number of shares of Purchaser Common Stock allocated to each Company Stockholder, and (D) the portion of the Earn Out Shares allocated to each Company Stockholder and (c) is and will otherwise be accurate in all respects (except for *de minimis* inaccuracies that are not material). For illustrative purposes only, set forth on Schedule 2.4 is the Allocation Schedule as it would have been prepared if the Closing Date were the date hereof (it being understood that such illustrative Allocation Schedule set forth on Schedule 2.4 is illustrative only and not binding in any manner on the Parties); provided that, the Parties agree that such illustrative Allocation Schedule shall not be required to set forth the mailing addresses and email addresses for the Company Stockholders. Notwithstanding anything in this Agreement to the contrary, upon delivery, payment, issuance, reserve for issuance or any other treatment of the Merger Consideration on the Closing Date in accordance with the Allocation Schedule, subject to Section 1.7 and Section 1.11, Purchaser and its respective Affiliates shall be deemed to have satisfied all obligations with respect to the payment of consideration under this Agreement, and none of them shall have (I) any further obligations to the Company, any Company Stockholder or any other Person with respect to the payment of any consideration under this Agreement (including with respect to the Merger Consideration) (other than the Earn Out Shares), or (II) any liability with respect to the allocation of the consideration under this Agreement, and the Company hereby irrevocably waives and releases Purchaser and its Affiliates (but excluding, on and after the Closing, the Company and its Affiliates) from all claims arising from or related to such Allocation Schedule and the allocation of the Merger Consideration, as the case may be, among each Company Stockholder as set forth in such Allocation Schedule.

ARTICLE III. REPRESENTATIONS AND WARRANTIES OF THE PURCHASER PARTIES

Except as set forth in (i) the disclosure schedules delivered by Purchaser Parties to the Company on the date hereof (the "*Purchaser Disclosure Schedules*"), the Section numbers of which are numbered to correspond to the Section numbers of this Agreement to which they refer, or (ii) the SEC Reports that are available on the SEC's website through EDGAR (to the extent the qualifying nature of such disclosure is readily apparent from the content of such SEC Reports, but excluding any disclosures solely contained or referenced therein under "Risk Factors," "Forward-Looking Statements," "Quantitative and Qualitative Disclosures About Market Risk," and any other disclosures solely contained or referenced therein of information, factors, or risks that are predictive, cautionary, or forward-looking in nature) (it being acknowledged that nothing disclosed in such an SEC Report will be deemed to modify or qualify the representations and warranties set forth in <u>Section 3.1</u> (Organization and Standing), <u>Section 3.2</u> (Authorization; Binding Agreement) and <u>Section 3.5</u> (Capitalization)), each Purchaser Party, jointly and severally, hereby represents and warrants to the Company as of the date of this Agreement as follows:

3.1 Organization and Standing. Each Purchaser Party is a corporation duly incorporated, validly existing and in good standing under the laws of its jurisdiction of incorporation. Each Purchaser Party has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted and as proposed to be conducted. Each Purchaser Party is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except where the failure to be so qualified or licensed or in good standing can be cured without material cost or expense. Each Purchaser Party has heretofore made available to the Company accurate and complete copies of its Organizational Documents, as currently in effect. No Purchaser Party is in violation of any provision of their respective Organizational Documents in any material respect.

3.2 Authorization; Binding Agreement. Each Purchaser Party has all requisite corporate power and authority to execute and deliver this Agreement and each Ancillary Document to which it is a party, to perform each such Purchaser Party's respective obligations hereunder and thereunder and to consummate Contemplated Transactions, subject to obtaining the Required Purchaser Stockholder Approval. The execution and delivery of this Agreement and each Ancillary Document to which it is a party and the consummation of the Contemplated Transactions (a) have been duly and validly authorized by the board of directors of Purchaser and the board of directors of the Merger Subs and New Pubco, and (b) other than the Required Purchaser Stockholder Approval, no other corporate proceedings, other than as set forth elsewhere in the Agreement, on the part of Purchaser Parties are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party or to consummate the Contemplated Transactions. This Agreement has been, and each Ancillary Document to which Purchaser Parties is a party shall be when delivered, duly and validly executed and delivered by Purchaser Parties and, assuming the due authorization, execution and delivery of this Agreement and such Ancillary Documents by the other parties hereto and thereto, constitutes, or when delivered shall constitute, the valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with its terms, except to the extent that enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization and moratorium laws and other laws of general application affecting the enforcement of creditors' rights generally or by any applicable statute of limitation or by any valid defense of set-off or counterclaim, and the fact that equitable remedies or relief (including the remedy of specific performance) are subject to the discretion of the court from which such relief may be sought (collectively, the "Enforceability Exceptions"). The Purchaser's board of directors, by resolutions duly adopted at a meeting duly called and held (i) determined that this Agreement and the Mergers and the Contemplated Transactions are advisable, fair to, and in the best interests of, Purchaser and its stockholders, (ii) approved this Agreement and the Mergers and the Contemplated Transactions in accordance with the DGCL, (iii) directed that this Agreement be submitted to Purchaser's stockholders for adoption and (iv) resolved to recommend that Purchaser's stockholders adopt this Agreement.

3.3 Governmental Approvals. Except as otherwise described in <u>Schedule 3.3</u>, no Consent of or with any Governmental Authority, on the part of Purchaser Parties is required to be obtained or made in connection with the execution, delivery or performance by Purchaser Parties of this Agreement and each Ancillary Document to which Purchaser Parties is a party or the consummation by Purchaser Parties of the Contemplated Transactions, other than (a) pursuant to Antitrust Laws, (b) such filings as contemplated by this Agreement, (c) any filings required with NYSE or the SEC with respect to the Contemplated Transactions, (d) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder, (e) required filings with, and approvals of, the Board of Governors of the Federal Reserve System (the "*Federal Reserve Board*") and the FDIC and (f) where the failure to obtain or make such Consents or to make such filings or notifications, would not reasonably be expected to prevent or materially delay consummation of Contemplated Transactions or otherwise prevent a Purchaser Party from performing its material obligations under this Agreement.

3.4 Non-Contravention. Except as otherwise described in Schedule 3.4, the execution and delivery by each Purchaser Party of this Agreement and each Ancillary Document to which it is a party, the consummation by each Purchaser Party of the Contemplated Transactions, and compliance by each Purchaser Party with any of the provisions hereof and thereof, will not (a) contravene or conflict with or violate any provision of a Purchaser Party's Organizational Documents, (b) contravene or conflict with or constitute a violation of any provisions of Law or Order binding upon or applicable to either Purchaser Party, (c) subject to obtaining the Consents from Governmental Authorities referred to in Section 3.3 hereof, and the waiting periods referred to therein having expired, and any condition precedent to such Consent or waiver having been satisfied, conflict with or violate any Law, Order or Consent applicable to a Purchaser Party, or any of its properties or assets, or (d) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by Purchaser under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien upon any of the properties or assets of Purchaser under, (viii) give rise to any obligation to obtain any third party Consent or provide any notice to any Person or (ix) give any Person the right to declare a default, exercise any remedy, claim a rebate, chargeback, penalty or change in delivery schedule, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any Purchaser Material Contract, except for any deviations from any of the foregoing clauses (b) or (c) that would not reasonably be expected to have a Material Adverse Effect on a Purchaser Party.

3.5 Capitalization.

(a) Purchaser is authorized to issue 101,000,000 shares of common stock, par value \$0.0001 per share; and is authorized to issue 1,000,000 shares of Purchaser Preferred Stock. The issued and outstanding Purchaser Securities as of the date of

this Agreement are set forth on <u>Schedule 3.5(a)</u>. There are no issued or outstanding shares of Purchaser Preferred Stock. All outstanding shares of Purchaser Common Stock are duly authorized, validly issued, fully paid and non-assessable and are not subject to or issued in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the Delaware General Corporation Law (as amended, the "*DGCL*"), Purchaser's Organizational Documents or any Contract to which Purchaser is a party. None of the outstanding Purchaser Securities has been issued in violation of any applicable securities Laws.

(b) Prior to giving effect to the Mergers, (i) Merger Sub 1 is authorized to issue 1,000 shares, par value \$0.001, of Merger Sub 1 Common Stock, of which 1,000 shares are issued and outstanding, and all of which are owned by Purchaser, and (ii) Merger Sub 2 is authorized to issue 1,000 shares, par value \$0.001, of Merger Sub 2 Common Stock, of which 1,000 shares are issued and outstanding, and all of which are owned by Purchaser. Prior to giving effect to the Contemplated Transactions, other than New Pubco and the Merger Subs, Purchaser does not have any Subsidiaries or own any equity interests in any other Person.

(c) Except as set forth in Schedule 3.5(a) or Schedule 3.5(c), there are no (i) outstanding options, warrants, puts, calls, convertible securities, preemptive or similar rights, (ii) bonds, debentures, notes or other Indebtedness having general voting rights or that are convertible or exchangeable into securities having such rights or (iii) subscriptions or other rights, agreements, arrangements, Contracts or commitments of any character (other than this Agreement and the Ancillary Documents), (A) relating to the issued or unissued shares of Purchaser or (B) obligating Purchaser to issue, transfer, deliver or sell or cause to be issued, transferred, delivered, sold or repurchased any options or shares or securities convertible into or exchangeable for such shares, or (C) obligating Purchaser to grant, extend or enter into any such option, warrant, call, subscription or other right, agreement, arrangement or commitment for such capital shares. Other than the Redemption or as expressly set forth in this Agreement, there are no outstanding obligations of Purchaser to repurchase, redeem or otherwise acquire any shares of Purchaser or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any Person. Except as set forth in Schedule 3.5(c), there are no stockholders agreements, voting trusts or other agreements or understandings to which Purchaser is a party with respect to the voting of any shares of Purchaser.

(d) All Indebtedness of Purchaser as of the date of this Agreement is disclosed on <u>Schedule 3.5(d)</u>. No Indebtedness of Purchaser contains any restriction upon (i) the prepayment of any of such Indebtedness, (ii) the incurrence of Indebtedness by Purchaser or (iii) the ability of Purchaser to grant any Lien on its properties or assets.

(e) Since the date of formation of Purchaser, and except as contemplated by this Agreement, Purchaser has not declared or paid any distribution or dividend in respect of its shares and has not repurchased, redeemed or otherwise acquired any of its shares, and Purchaser's board of directors has not authorized any of the foregoing.

3.6 SEC Filings and Purchaser Financials.

(a) The Purchaser, since the IPO and prior to the date of this Agreement, has timely filed all forms, reports, schedules, statements, registration statements, prospectuses and other documents required to be filed or furnished by Purchaser with the SEC under the Securities Act and/or the Exchange Act, together with any amendments, restatements or supplements thereto, and will file all such forms, reports, schedules, statements and other documents required to be filed subsequent to the date of this Agreement. Except to the extent available on the SEC's web site through EDGAR, Purchaser has delivered to the Company copies in the form filed with the SEC of all of the following: (i) Purchaser's annual reports on Form 10-K for each fiscal year of Purchaser beginning with the first year Purchaser was required to file such a form, (ii) Purchaser's quarterly reports on Form 10-Q for each fiscal quarter that Purchaser filed such reports to disclose its quarterly financial results in each of the fiscal years of Purchaser referred to in clause (i) above, (iii) all other forms, reports, registration statements, prospectuses and other documents (other than preliminary materials) filed by Purchaser with the SEC since the beginning of the first fiscal year referred to in clause (i) above (the forms, reports, registration statements, prospectuses and other documents referred to in clauses (i), (ii) and (iii) above, whether or not available through EDGAR, are, collectively, the "SEC Reports") and (iv) all certifications and statements required by (A) Rules 13a-14 or 15d-14 under the Exchange Act, and (B) 18 U.S.C. § 1350 (Section 906 of SOX) with respect to any report referred to in clause (i) above (collectively, the "Public Certifications"). The SEC Reports (x) were prepared in all material respects in accordance with the requirements of the Securities Act and the Exchange Act, as the case may be, and the rules and regulations thereunder and (y) did not, as of their respective effective dates (in the case of SEC Reports that are

registration statements filed pursuant to the requirements of the Securities Act) and at the time they were filed with the SEC (in the case of all other SEC Reports) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. As of the date of this Agreement, there are no outstanding or unresolved comments in comment letters received from the SEC with respect to any SEC Reports. None of the SEC Reports filed on or prior to the date of this Agreement is subject to ongoing SEC review or investigation as of the date of this Agreement. The Public Certifications are each true as of their respective dates of filing. As used in this <u>Section 3.6</u>, the term "file" shall be broadly construed to include any manner permitted by SEC rules and regulations in which a document or information is furnished, supplied or otherwise made available to the SEC. As of the date of this Agreement, (A) Purchaser Units, Purchaser Common Stock and Purchaser Public Warrants are listed on NYSE, (B) Purchaser has not received any written deficiency notice from NYSE relating to the continued listing requirements of such Purchaser Securities, (C) there are no Actions pending or, to the Knowledge of Purchaser, threatened against Purchaser by the Financial Industry Regulatory Authority with respect to any intention by such entity to suspend, prohibit or terminate the quoting of such Purchaser Securities on NYSE and (D) such Purchaser Securities are in compliance with all of the applicable corporate governance rules of NYSE.

15

(b) The Purchaser maintains disclosure controls and procedures required by Rule 13a-15 or Rule 15d-15 under the Exchange Act; such controls and procedures are reasonably designed to ensure that all material information concerning Purchaser and other material information required to be disclosed by Purchaser in the reports and other documents that it files or furnishes under the Exchange Act is made known on a timely basis to the individuals responsible for the preparation of Purchaser's SEC filings and other public disclosure documents. Such disclosure controls and procedures are effective in timely alerting Purchaser's principal executive officer and principal financial officer to material information required to be included in Purchaser's periodic reports required under the Exchange Act.

(c) The Purchaser maintains a standard system of accounting established and administered in accordance with GAAP. The Purchaser has designed and maintains a system of internal controls over financial reporting, as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act, sufficient to provide reasonable assurances regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. The Purchaser maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

(d) The financial statements and notes of Purchaser contained or incorporated by reference in the SEC Reports (the "*Purchaser Financials*"), fairly present in all material respects the financial position and the results of operations, changes in stockholders' equity, and cash flows of Purchaser at the respective dates of and for the periods referred to in such financial statements, all in accordance with (i) GAAP methodologies applied on a consistent basis throughout the periods involved and (ii) Regulation S-X or Regulation S-K, as applicable (except as may be indicated in the notes thereto and for the omission of notes and audit adjustments in the case of unaudited quarterly financial statements to the extent permitted by Regulation S-X or Regulation S-K, as applicable).

(e) Except as and to the extent reflected or reserved against in Purchaser Financials, Purchaser has not incurred any Liabilities or obligations of the type required to be reflected on a balance sheet in accordance with GAAP that are not adequately reflected or reserved on or provided for in Purchaser Financials, other than Liabilities of the type required to be reflected on a balance sheet in accordance with GAAP that have been incurred since Purchaser's formation in the ordinary course of business. All debts and Liabilities, fixed or contingent, which should be included under U.S. GAAP on a balance sheet are included in all material respects in Purchaser Financials as of the date of such Purchaser Financial. The Purchaser has no off-balance sheet arrangements.

3.7 Absence of Certain Changes. As of the date of this Agreement, except as set forth in <u>Schedule 3.7</u>, Purchaser has, (a) since its formation, conducted no business other than its formation, the public offering of its securities (and the related private offerings), public reporting and its search for an initial Business Combination as described in the IPO Prospectus (including the investigation of the Company and the negotiation and execution of this Agreement) and related activities and (b) since December 31, 2021, not been subject to a Material Adverse Effect on Purchaser.

3.8 Compliance with Laws. The Purchaser is, and has since its formation been, in compliance with all Laws applicable to it and the conduct of its business except for such noncompliance which would not reasonably be expected to prevent or materially delay consummation of the Contemplated Transactions or otherwise prevent a Purchaser Party from performing its material obligations under this Agreement, and Purchaser has not received written notice alleging any violation of applicable Law in any material respect by Purchaser. Purchaser is not under investigation with respect to any violation or alleged violation of, any law, or judgment, order or decree entered by any court, arbitrator or Governmental Authority, domestic or foreign, and Purchaser has not previously received any subpoenas from any Governmental Authority.

3.9 Actions; Orders; Permits. There is no pending or, to the Knowledge of Purchaser, threatened material Action to which Purchaser is subject which would not reasonably be expected to prevent or materially delay consummation of the Contemplated Transactions or otherwise prevent Purchaser Parties from performing its material obligations under this Agreement. There is no material Action that Purchaser has pending against any other Person. The Purchaser is not subject to any material Orders of any Governmental Authority, nor are any such Orders pending. The Purchaser holds all material Permits necessary to lawfully conduct its business as presently conducted, and to own, lease and operate its assets and properties, all of which are in full force and effect, except where the failure to hold such Consent or for such Consent to be in full force and effect would not reasonably be expected to prevent or materially delay consummation of the Contemplated Transactions or otherwise prevent a Purchaser Party from performing its material obligations under this Agreement. To the Knowledge of Purchaser Party, none of their respective directors or officers, nor, any of their respective employees, agents, or any other Persons acting for or on behalf of such Purchaser Party has (i) made, offered, promised, authorized, paid or received any unlawful bribes, kickbacks or other similar payments to or from any Person, (ii) made, offered, promised, authorized or paid any unlawful contributions to a domestic or foreign political party or candidate or (iii) otherwise made, offered, promised, authorized, paid or received any improper payment in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977 or any other local or foreign anti-corruption or bribery Law. To the Knowledge of Purchaser Party, no Purchaser Party, none of their respective directors or officers, nor any of their employees, agents, or any other Persons acting for or on behalf of such Purchaser Party is or has been a Person named on any economic sanctions administered, enacted or enforced by any Governmental Authority.

3.10 Taxes and Returns'

(a) The Purchaser Parties have or will have timely filed, or caused to be timely filed, all income and other material Tax Returns required to be filed by them (taking into account all available extensions), which Tax Returns are true, accurate, correct and complete in all material respects, and has paid, collected or withheld, or caused to be paid, collected or withheld, all material Taxes required to be paid, collected or withheld, other than such Taxes for which adequate reserves in Purchaser Financials have been established in accordance with GAAP.

(b) There is no Action currently pending or, to the Knowledge of Purchaser, threated against any Purchaser Party by a Governmental Authority in a jurisdiction where a Purchaser Party does not file Tax Returns that it is or may be subject to taxation in that jurisdiction.

17

(c) There are no claims, assessments, audits, examinations, investigations or other Actions pending against any Purchaser Party in respect of any Tax, and no Purchaser Party has been notified in writing of any proposed Tax claims or assessments against it (other than, in each case, claims or assessments for which adequate reserves in Purchaser Financials have been established in accordance with GAAP).

(d) There are no Liens with respect to any Taxes upon any assets of any of Purchaser Party, other than Permitted Liens.

(e) No Purchaser Party has any outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes. There are no outstanding requests by any Purchaser Party for any extension of time within which to file any Tax Return or within which to pay any Taxes shown to be due on any Tax Return.

(f) No Purchaser Party has made any change in accounting method (except as required by a change in Law) or received a ruling from, or signed an agreement with, any taxing authority that would reasonably be expected to have a material impact on its Taxes following the Closing.

(g) No Purchaser Party has engaged in any "listed transaction," as defined in U.S. Treasury Regulation section 1.6011-4(b)(2).

(h) No Purchaser Party has any Liability for the Taxes of another Person (i) under any applicable Tax Law, (ii) as a transferee or successor, or (iii) by contract or indemnity (excluding commercial agreements entered into in the ordinary course of business the primary purpose of which is not the sharing of Taxes). No Purchaser Party is a party to or bound by any Tax indemnity agreement, Tax sharing agreement or Tax allocation agreement or similar agreement, arrangement or practice (excluding commercial agreements, arrangements or practices entered into in the ordinary course of business the primary purpose of which is not the sharing of Taxes) with respect to Taxes (including an advance pricing agreement, closing agreement or other agreement relating to Taxes with any Governmental Authority) that will be binding on such Purchaser Party with respect to any period following the Closing Date

(i) No Purchaser Party has requested, nor is it the subject of or bound by, any private letter ruling, technical advice memorandum, closing agreement or similar ruling, memorandum or agreement with any Governmental Authority with respect to any Taxes, nor is any such request outstanding.

(j) No Purchaser Party: (i) has constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of securities qualifying for, or intended to qualify for, Tax-free treatment under Section 355 of the Code within the two-year period ending on the date hereof; or (ii) is or has ever been (A) a U.S. real property holding corporation within the meaning of Section 897(c)(2) of the Code, or (B) a member of any consolidated, combined, unitary or affiliated group of corporations for any Tax purposes other than a group of which a Purchaser Party is or was the common parent corporation.

(k) To the Knowledge of Purchaser, there are no facts or circumstances that would reasonably be expected to prevent the Contemplated Transactions from qualifying for the Intended Tax Treatment.

3.11 Employees and Employee Benefit Plans. The Purchaser does not (a) have any paid employees or (b) maintain, sponsor, contribute to or otherwise have any Liability under, any Benefit Plans.

18

3.12 Properties. The Purchaser does not own, license or otherwise have any right, title or interest in any material Intellectual Property Rights. The Purchaser does not own or lease any material real property or material Personal Property.

3.13 Material Contracts.

(a) Except as set forth on <u>Schedule 3.13(a)</u>, other than this Agreement and the Ancillary Documents, there are no Contracts to which Purchaser is a party or by which any of its properties or assets may be bound, subject or affected, which (i) creates or imposes a Liability greater than \$100,000, (ii) may not be cancelled by Purchaser on less than sixty (60) days' prior notice without payment of a material penalty or termination fee or (iii) prohibits, prevents, restricts or impairs in any material respect any business practice of Purchaser as its business is currently conducted, any acquisition of material property by Purchaser, or restricts in any material respect the ability of Purchaser to engage in business as currently conducted by it or compete with any other Person (each, a "*Purchaser Material Contract*"). All Purchaser Material Contracts have been made available to the Company other than those that are exhibits to the SEC Reports.

(b) With respect to each Purchaser Material Contract: (i) Purchaser Material Contract was entered into at arms' length and in the ordinary course of business; (ii) Purchaser Material Contract is legal, valid, binding and enforceable in all material

respects against Purchaser and, to the Knowledge of Purchaser, the other parties thereto, and is in full force and effect (except, in each case, as such enforcement may be limited by the Enforceability Exceptions); (iii) Purchaser is not in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default in any material respect by Purchaser, or permit termination or acceleration by the other party, under such Purchaser Material Contract; and (iv) to the Knowledge of Purchaser, no other party to any Purchaser Material Contract is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default by such other party, or permit termination or acceleration by Purchaser under any Purchaser Material Contract.

3.14 Litigation. There is no Proceeding pending, or to the Knowledge of Purchaser, threatened against Purchaser Parties of any of their respective properties or rights. There are no Proceedings (at Law or in equity) or investigations pending or, to the Knowledge of Purchaser, threatened, seeking to or that would reasonably be expected to prevent, hinder, modify, delay or challenge the Mergers or the Contemplated Transactions.

3.15 Transactions with Affiliates. Schedule 3.15 sets forth a true, correct and complete list of the Contracts and arrangements that are in existence as of the date of this Agreement under which there are any existing or future Liabilities or obligations between Purchaser and any (a) present or former director, officer or employee or Affiliate of Purchaser, or any immediate family member of any of the foregoing, or (b) record or beneficial owner of more than five percent (5%) of Purchaser's outstanding capital stock as of the date hereof.

3.16 Business Activities. Since its formation, Purchaser has not conducted any business activities other than activities directed toward completing a Business Combination. Since their formation, neither New Pubco nor the Merger Subs have engaged in any business activities other than as contemplated by this Agreement, own directly or indirectly any ownership, equity, profits or voting interest in any Person or have any assets or Liabilities except those incurred in connection with this Agreement and the Ancillary Documents to which New Pubco and such Merger Subs are a party and the Merger, and, other than this Agreement and the Ancillary Documents to which it is a party, neither New Pubco nor the Merger Subs are party to or bound by any Contract.

3.17 Finders and Brokers. Except as set forth on <u>Schedule 3.17</u>, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission from Purchaser, the Company or any of their respective Affiliates in connection with the Contemplated Transactions based upon arrangements made by or on behalf of Purchaser.

3.18 Ownership of Merger Consideration. All shares of New Pubco Common Stock to be issued and delivered to the Company Stockholders as Merger Consideration in accordance with <u>Article I</u> shall be, upon issuance and delivery of such New Pubco Common Stock, fully paid and non-assessable, free and clear of all Liens, other than restrictions arising from applicable securities Laws, any applicable Registration Rights and Lock-Up Agreement, and any Liens incurred by any Company Stockholder, and the issuance and sale of such New Pubco Common Stock pursuant hereto will not be subject to or give rise to any preemptive rights or rights of first refusal.

3.19 Certain Business Practices.

(a) The Purchaser has not, to its Knowledge, (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 or any other local or foreign anti-corruption or bribery Law, (iii) made any other unlawful payment or (iv) since the formation of Purchaser, directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder Purchaser or assist it in connection with any actual or proposed transaction.

(b) The Purchaser, the operations of Purchaser are and have been conducted at all times in material compliance with money laundering statutes in all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority, and no Action involving Purchaser with respect to any of the foregoing is pending or, to the Knowledge of Purchaser, threatened.

(c) None of Purchaser or any of its directors or officers, or, to the Knowledge of Purchaser, any other Representative acting on behalf of Purchaser is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("**OFAC**"), and Purchaser has not, in the last five (5) fiscal years, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any Subsidiary, joint venture partner or other Person, in connection with any sales or operations in Cuba, Iran, North Korea, Syria, the Donetsk People's Republic, Luhansk People's Republic, and Crimea regions of Ukraine, or any other country sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC.

3.20 Lock-Up Agreements. All existing lock-up agreements between Purchaser and any of its stockholders or holders of any Purchaser Securities entered into in connection with the initial public offering of Purchaser provide for a lock-up period that is in full force and effect.

3.21 Purchaser Trust Account. As of the date of this Agreement, the Trust Account has a balance of no less than \$201,000,000. Such monies are invested solely in United States "government securities" within the meaning of Section 2(a)(16) of the Investment Company Act or money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act, and held in trust by Continental Stock Transfer & Trust Company pursuant to the Trust Agreement. The Trust Agreement is valid and in full force and effect and enforceable in accordance with its terms (subject to the Enforceability Exceptions) and has not been amended or modified. The Purchaser has complied in all respects with the terms of the Trust Agreement and is not in breach thereof or default thereunder and there does not exist under the Trust Agreement any event which, with the giving of notice or the lapse of time, would constitute such a breach or default by Purchaser or, to the Knowledge of Purchaser, by the Trustee. There are no separate agreements, side letters or other agreements that would cause the description of the Trust Agreement in the SEC Reports to be inaccurate in any material respect and/or that would entitle any Person (other than the underwriters of the IPO, Public Stockholders who shall have elected to redeem their Purchaser Common Stock pursuant to the Purchaser Certificate of Incorporation (or in connection with an extension of Purchaser's deadline to consummate a Business Combination) or Governmental Authorities for Taxes) to any portion of the proceeds in the Trust Account. Prior to the Closing, none of the funds held in the Trust Account may be released except as described in the Trust Agreement.

3.22 Independent Investigation. The Purchaser has conducted its own independent investigation, review and analysis of the business, results of operations, prospects, condition (financial or otherwise) or assets of the Company (including the Contributed Assets), and acknowledges that it has been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of the Company and Atlas Fintech (solely with respect to the Contributed Assets) for such purpose. The Purchaser acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the Contemplated Transactions, it has relied solely upon its own investigation and the express representations and warranties of the Company set forth in <u>Article IV</u> of this Agreement (including the related portions of the Company Disclosure Schedules), the representations and warranties of the Target Bank set forth in the Bank Acquisition Agreement, and in any certificate delivered to Purchaser pursuant hereto, and the information provided by or on behalf of the Company for the Registration Statement/ Proxy Statement; and (b) none of the Company, Atlas Fintech (with respect to the Contributed Assets), nor its respective Representatives have made any representation or warranty as to the Company, Atlas Fintech, or this Agreement, except as expressly set forth in this Agreement (including the related portions of the Company for the Proxy Statement) or with respect to the information provided by or on behalf of the Proxy Statement.

ARTICLE IV. REPRESENTATIONS AND WARRANTIES OF THE COMPANY AND THE COMPANY STOCKHOLDERS

Except as set forth in the disclosure schedules delivered by the Company to Purchaser Parties on the date hereof (the "*Company Disclosure Schedules*"), the Section numbers of which are numbered to correspond to the Section numbers of this Agreement to which they refer, each of the Company and the Company Stockholders, jointly and severally, hereby represents and warrants to the Purchaser Parties, as of the date of this Agreement, as follows:

4.1 Organization and Standing. The Company is a corporation duly incorporated, validly existing and in good standing under the State of Wyoming and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its

business as now being conducted and as proposed to be conducted. Each Subsidiary of the Company is a corporation or other entity duly formed, validly existing and in good standing under the Laws of its jurisdiction of organization and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted and as proposed to be conducted.

4.2 Authorization; Binding Agreement. The Company has all requisite corporate power and authority to execute and deliver this Agreement and each Ancillary Document to which it is or is required to be a party, to perform the Company's obligations hereunder and thereunder and to consummate the Contemplated Transactions, subject to obtaining the Required Company Stockholder Approval. The execution and delivery of this Agreement and each Ancillary Document to which the Company is or is required to be a party and the consummation of the Contemplated Transactions, (a) have been duly and validly authorized by the Company's board of directors in accordance with the Company's Organizational Documents, the WBCA, any other applicable Law or any Contract to which the Company or any of its stockholders is a party or by which it or its securities are bound and (b) other than the Required Company Stockholder Approval, no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party or to consummate the Contemplated Transactions. This Agreement has been, and each Ancillary Document to which the Company is or is required to be a party shall be when delivered, duly and validly executed and delivered by the Company and assuming the due authorization, execution and delivery of this Agreement and any such Ancillary Document by the other parties hereto and thereto, constitutes, or when delivered shall constitute, the legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the Enforceability Exceptions. The Company's board of directors, by resolutions duly adopted at a meeting duly called and held or by action by unanimous written consent in accordance with the Company's Organizational Documents (i) determined that this Agreement and the Second Merger and the Contemplated Transactions are advisable, fair to, and in the best interests of, the Company and its stockholders, (ii) approved this Agreement and the Second Merger and the Contemplated Transactions in accordance with the WBCA, (iii) directed that this Agreement be submitted to the Company Stockholders for adoption and (iv) resolved to recommend that the Company Stockholders adopt this Agreement.

4.3 Capitalization.

(a) The Company is authorized to issue (i) 12,000,000 shares of Company Common Stock, of which 2,000,000 shares are issued and outstanding, and (ii) 10,000,000 shares of Company Preferred Stock, none of which are issued and outstanding. With respect to the Company Preferred Stock, the Company has not designated any series or other classification. Prior to giving effect to the Contemplated Transactions, all of the issued and outstanding Company Stock and other equity interests of the Company are set forth on <u>Schedule 4.3(a)</u>, along with the beneficial and record owners thereof, all of which shares and other equity interests are owned free and clear of any Liens other than those imposed under the Company Charter. All of the outstanding shares and other equity interests of the Company have been duly authorized, are fully paid and non-assessable and not in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the WBCA, any other applicable Law, the Company Charter or any Contract to which the Company is a party or by which it or its securities are bound. The Company holds no shares or other equity interests of the Company were issued in violation of any applicable securities Laws. The rights, privileges and preferences of the Company Preferred Stock are as stated in the Company Charter and as provided by the WBCA.

(b) The Company has not reserved any shares of Company Common Stock for issuance to officers, directors, employees and consultants of the Company pursuant to any Company equity plan, and no such plan exists or has ever been adopted by the Company's board of directors or approved by the Company Stockholders. There are no Company Convertible Securities, or preemptive rights or rights of first refusal or first offer, nor are there any Contracts, commitments, arrangements or restrictions to which the Company or, to the Knowledge of the Company, any of its stockholders is a party or bound relating to any equity securities of the Company, whether or not outstanding. There are no voting trusts, proxies, stockholder agreements or any other agreements or understandings with respect to the voting of the Company's equity interests. There are no outstanding contractual obligations of the Company to repurchase, redeem or otherwise acquire any equity interests or securities of the Company granted any registration rights to any Person with respect to the Company's securities. All shares of Company Stock have been granted, offered, sold and issued in compliance with all applicable securities Laws. As a result of the consummation of the Contemplated Transactions, no equity interests of the Company are issuable, and no rights

in connection with any interests, warrants, rights, options or other securities of the Company accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).

(c) The Company has never declared or paid any distribution or dividend in respect of its equity interests and has not repurchased, redeemed or otherwise acquired any equity interests of the Company, and the board of directors of the Company has not authorized any of the foregoing.

4.4 Subsidiaries. Prior to the consummation of the acquisition of the Broker-Dealer Target, the Company has no Subsidiaries. Except for the Broker-Dealer Acquisition Agreement and the Bank Acquisition Agreement, the Company does not own or have any rights to acquire, directly or indirectly, any equity interests of, or otherwise Control, any Person. None of the Company or its Subsidiaries is a participant in any joint venture, partnership or similar arrangement. There are no outstanding contractual obligations of the Company or its Subsidiaries to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any other Person.

4.5 Governmental Approvals(a). Except (a) filings with, and approvals by, the Federal Reserve Board under the Bank Holding Company Act of 1956, the FDIC, and with any state banking regulatory authorities listed on <u>Schedule 3.3</u> or <u>Schedule 4.5</u>, (b) the filing with, and approval by, FINRA of the Target Broker-Dealer's continuing membership application required by FINRA or (c) as otherwise described in <u>Schedule 4.5</u>, no Consent of or with any Governmental Authority on the part of any of the Company, any Subsidiary of the Company is required to be obtained or made in connection with the execution, delivery or performance by the Company of this Agreement or any Ancillary Documents or the consummation by the Company of the Contemplated Transactions, other than (i) such filings as are expressly contemplated by this Agreement or (ii) pursuant to Antitrust Laws.

4.6 Non-Contravention. Except as otherwise described in Schedule 4.6, the execution and delivery by the Company of this Agreement and each Ancillary Document to which the Company is or is required to be a party or otherwise bound, and the consummation by the Company of the Contemplated Transactions and compliance by the Company with any of the provisions hereof and thereof, will not (a) conflict with or violate any provision of the Company's Organizational Documents, (b) subject to obtaining the Consents from Governmental Authorities referred to in Section 4.6 hereof, the waiting periods referred to therein having expired, and any condition precedent to such Consent or waiver having been satisfied, conflict with or violate any Law, Order or Consent applicable to the Company, any of the Company's material properties or assets, or any Contributed Assets, (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by the Company under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien upon any of the properties or assets of the Company, or the Contributed Assets, under (other than Permitted Liens), (viii) give rise to any obligation to obtain any third party Consent or provide any notice to any Person or (ix) give any Person the right to declare a default, exercise any remedy, claim a rebate, chargeback, penalty or change in delivery schedule, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, in each case, any of the terms, conditions or provisions of any Company Material Contract, or any Contract materially affecting the Contributed Assets, except in the cases of clauses (b) or (c), as has not been a Material Adverse Effect on the Company.

4.7 Financial Statements.

(a) As used herein, the term "*Company Financials*" means the unaudited financial statements of the Company, consisting of the consolidated balance sheet of the Company as of June 30, 2022 (the "*Interim Balance Sheet Date*") and the related consolidated income statement, changes in stockholder equity and statement of cash flows for the period from March 17, 2022 (date of inception) to the Interim Balance Sheet Date. True and correct copies of the Company Financials have been provided to Purchaser. The Company Financials (i) accurately reflect the books and records of the Company as of the times and for the periods referred to therein, (ii) were prepared in accordance with GAAP, consistently applied throughout and among the periods involved (except the Company Financials exclude the footnote disclosures and other presentation items required for GAAP and exclude year-end adjustments which will not be material in amount), (iii) comply in all material respects with all

applicable accounting requirements under the Securities Act and the rules and regulations of the SEC thereunder, and (iv) fairly present in all material respects the consolidated financial position of the Company as of the respective dates thereof and the consolidated results of the operations and cash flows of the Company for the periods indicated. The Company has never been subject to the reporting requirements of Sections 13(a) and 15(d) of the Exchange Act.

(b) The Company maintains accurate books and records reflecting its assets and Liabilities and maintains proper and adequate internal accounting controls that provide reasonable assurance that (i) the Company does not maintain any off-the-book accounts and that the Company's assets are used only in accordance with the Company's management directives, (ii) transactions are executed with management's authorization, (iii) transactions are recorded as necessary to permit preparation of the financial statements of the Company and to maintain accountability for the Company's assets, (iv) access to the Company's assets is permitted only in accordance with management's authorization, (v) the reporting of the Company's assets is compared with existing assets at regular intervals and verified for actual amounts, and (vi) accounts, notes and other receivables and inventory are recorded accurately, and proper and adequate procedures are implemented to effect the collection of accounts, notes and other receivables on a current and timely basis. All of the financial books and records of the Company are complete and accurate in all material respects and have been maintained in the ordinary course consistent with past practice and in accordance with applicable Laws. The Company has not been subject to or involved in any material fraud that involves management or other employees who have a significant role in the internal controls over financial reporting of the Company. Since its incorporation, neither the Company nor its Representatives has received any written complaint, allegation, assertion or claim regarding the accounting or auditing practices, procedures, methodologies or methods of the Company or its internal accounting controls, including any material written complaint, allegation, assertion or claim that the Company has engaged in questionable accounting or auditing practices.

(c) The Company does not have any Indebtedness.

(d) Except as set forth on <u>Schedule 4.7(d)</u>, the Company is not subject to any Liabilities or obligations (whether or not required to be reflected on a balance sheet prepared in accordance with GAAP), except for those that are not material and that were incurred after the Interim Balance Sheet Date in the ordinary course of business consistent with past practice (other than Liabilities for breach of any Contract or violation of any Law).

24

4.8 Absence of Certain Changes. Except with respect to the negotiation and execution of this Agreement and the Ancillary Documents, or as set forth on <u>Schedule 4.8</u>, since its incorporation, the Company has (a) not conducted any business other than the negotiation and execution of this Agreement and the Ancillary Documents and actions related thereto, (b) not been subject to a Material Adverse Effect and (c) has not taken any action or committed or agreed to take any action that would be prohibited by <u>Section 6.2(b)</u> (without giving effect to <u>Schedule 6.2</u>) if such action were taken on or after the date hereof without the consent of Purchaser.

4.9 Compliance with Laws. Each of the Company and Atlas FinTech (solely with respect to the Contributed Assets) is not, and since its incorporation has not been, in material conflict or material non-compliance with, or in material default or material violation of, nor has the Company nor Atlas FinTech (solely with respect to the Contributed Assets), received, since its incorporation, any written or, to the Knowledge of the Company, oral notice of any material conflict or material non-compliance with, or material default or material violation of, any applicable Laws by which it or any of its properties, assets, employees, business, products or operations are or were bound or affected (including the Contributed Assets in the case of Atlas FinTech).

4.10 Company Permits. The Company (and its employees who are legally required to be licensed by a Governmental Authority in order to perform his or her duties with respect to his or her employment with the Company) holds all material Permits necessary to lawfully conduct in all material respects its business as presently conducted and as proposed to be conducted; to own, lease and operate its assets and properties (collectively, the "*Company Permits*"). The Company has made available to Purchaser true, correct and complete copies of all material Company Permits, all of which Company Permits are listed on <u>Schedule 4.10</u>. All of the material Company Permits are in full force and effect, and no suspension or cancellation of any of the material Company Permits is pending or, to the Company's Knowledge, threatened. Except as would not be material to the Company, the Company is not in violation in any material respect of the terms of any Company Permit, and the Company has not received any written or, to the Company's Knowledge, oral notice of any Actions relating to the revocation or modification of any Company Permit.

4.11 Litigation. Except as described on <u>Schedule 4.11</u>, there is no (a) Action of any nature currently pending or, to the Company's Knowledge, threatened, and no such Action has been brought since the Company's incorporation; or (b) Order now pending or outstanding or that was rendered by a Governmental Authority since the Company's incorporation, in either case of (a) or (b) by or against (x) the Company, its current or former directors, officers or equity holders (provided, that any litigation involving the directors, officers or equity holders of the Company must be related to the Company's business, equity securities or assets), its business, equity securities or assets, or (y) by or against Atlas FinTech, its current or former directors, officers or equity holders (provided, that any litigation involving the directors, officers or equity holders of Atlas FinTech must be directly related to the Contributed Assets), and the Contributed Assets. The items listed on <u>Schedule 4.11</u>, if finally determined adversely to the Company, will not have, either individually or in the aggregate, a Material Adverse Effect on the Company. Since the Company's incorporation, none of the current or former officers, senior management or directors of the Company have been charged with, indicted for, arrested for, or convicted of any felony or any crime involving fraud.

25

4.12 Material Contracts.

(a) <u>Schedule 4.12(a)</u> sets forth a true, correct and complete list of, and the Company has made available to Purchaser (including written summaries of oral Contracts), true, correct and complete copies of, each Contract to which (x) the Company is a party or by which the Company, or any of its properties or assets are bound or affected, or (y) the Contributed Assets are bound or directly affected (each Contract required to be set forth on <u>Schedule 4.12(a)</u>, other than a Company Benefit Plan, a "*Company Material Contract*") that:

(i) contains covenants that materially limit the ability of the Company (A) to compete in any line of business or with any Person or in any geographic area or to sell, or provide any service or product or solicit any Person, including any non-competition covenants, employee and customer non-solicit covenants, exclusivity restrictions, rights of first refusal or most-favored pricing clauses or (B) to purchase or acquire an interest in any other Person;

(ii) involves any joint venture, profit-sharing, partnership, limited liability company or other similar agreement or arrangement relating to the formation, creation, operation, management or control of any partnership or joint venture;

(iii) involves any exchange traded, over the counter or other swap, cap, floor, collar, futures contract, forward contract, option or other derivative financial instrument or Contract, based on any commodity, security, instrument, asset, rate or index of any kind or nature whatsoever, whether tangible or intangible, including currencies, interest rates, foreign currency and indices;

(iv) evidences Indebtedness (whether incurred, assumed, guaranteed or secured by any asset) of the Company having an outstanding principal amount in excess of \$7,000,000;

(v) involves the acquisition or disposition, directly or indirectly (by merger or otherwise), of assets with an aggregate value in excess of \$7,000,000 (other than in the ordinary course of business consistent with past practice) or shares or other equity interests of the Company or another Person, which Contracts shall include the Broker-Dealer Acquisition Agreement and the Bank Acquisition Agreement;

(vi) relates to any merger, consolidation or other business combination with any other Person or the acquisition or disposition of any other entity or its business or material assets or the sale of the Company, its business or material assets;

(vii) by its terms, individually or with all related Contracts, calls for aggregate payments or receipts by the Company under such Contract or Contracts of at least \$7,000,000 per year or in the aggregate;

(viii) is between the Company and any directors, officers or employees of the Company (other than at-will employment arrangements with employees entered into in the ordinary course of business consistent with past practice), including all non-competition, non-solicitation, severance and indemnification agreements, or any Related Person;

(ix) obligates the Company to make any capital commitment or expenditure in excess of \$7,000,000 (including pursuant to any joint venture);

(x) relates to a material settlement entered into within two (2) years prior to the date of this Agreement or under which any the Company has outstanding obligations (other than customary confidentiality obligations); or

(xi) relates to Third Party Intellectual Property Rights or Third Party Technology and is required to be listed in <u>Schedule 4.13(d)</u>;

(xii) involving the license or grant of rights to material Company Owned Intellectual Property Rights or Company Owned Technology by the Company, but excluding any nonexclusive licenses (or sublicenses) of Company Owned Intellectual Property Rights or Company Owned Technology in the ordinary course of business;

(xiii) that materially restricts the Company in the use, development, enforcement, prosecution, maintenance, transfer, licensing or other exploitation (the "*Exploitation*") of any Company Intellectual Property Rights used in or necessary for the Exploitation of the Products and Services in the manner so done currently or the conduct of the Company's business as currently conducted and as proposed to be conducted.

(b) Except as disclosed in Schedule 4.12(b), with respect to each Company Material Contract: (i) such Company Material Contract is valid and binding and enforceable in all respects against the Company and, to the Knowledge of the Company, each other party thereto, and is in full force and effect (except, in each case, as such enforcement may be limited by the Enforceability Exceptions); (ii) the consummation of the Contemplated Transactions will not affect the validity or enforceability of any Company Material Contract; (iii) the Company is not in breach or default in any material respect, and, to the Knowledge of the Company, no event has occurred that with the passage of time or giving of notice or both would constitute a material breach or material default by the Company, or permit termination or acceleration by the other party thereto, under such Company Material Contract; (iv) to the Knowledge of the Company, no other party to such Company Material Contract is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a material breach or material default by such other party, or permit termination or acceleration by the Company, under such Company Material Contract; (v) the Company has not received written or, to the Company's Knowledge, notice of an intention by any party to any such Company Material Contract that provides for a continuing obligation by any party thereto to terminate such Company Material Contract or amend the terms thereof, other than modifications in the ordinary course of business. The representations and warranties of the Target Broker-Dealer and its stockholders set forth in the Broker-Dealer Acquisition Agreement (as modified by the disclosure schedules delivered by the Target Broker-Dealer to the Company pursuant thereto) are true and correct as of the date the Broker-Dealer Acquisition Agreement was executed, and the representations and warranties of the Target Bank set forth in the Bank Acquisition Agreement (as modified by the disclosure schedules delivered by the Target Broker-Dealer to the Company pursuant thereto), are true and correct as of the date the Broker-Dealer Acquisition Agreement was executed, except, in each case, for those representations and warranties that address matters only as of a particular date (which representations and warranties shall have been accurate as of such date).

4.13 Intellectual Property.

(a) The Company or one of its Affiliates (i) has sole title to and ownership of all Company Owned Intellectual Property Rights and Company Owned Technology, free and clear of any Liens, other than Permitted Liens, and other rights or claims of others, and (ii) possesses legally enforceable rights, under valid and subsisting written Contracts, all Third Party Intellectual Property Rights and Third Party Technology.

(b) <u>Schedule 4.13(b)(i)</u> lists all Patents, Trademark and Copyright applications and registrations (x) owned by, purported to be owned by, or licensed exclusively to the Company or any of its Affiliates or (y) owned by, purported to be

owned by, or licensed exclusively to Atlas FinTech as a Contributed Asset ("Company Registered Intellectual Property"), including, in each case, the jurisdiction in which each item of such Company Registered Intellectual Property has been issued or registered or in which any such application for such issuance or registration has been filed, the title, application number, filing date, jurisdiction and registration number. Schedule 4.13(b)(ii) lists all Products and material Software and Trademarks that are unregistered Company Owned Intellectual Property Rights and Company Owned Technology. Each registration of Company Owned Intellectual Property Rights, and each Company Internet Name, was properly registered and is in good standing and enforceable under applicable Laws, and no renewal, fee, payment or other action is required to be taken with respect to any such registration, or otherwise. With respect to each item of Company Registered Intellectual Property, to the Company's Knowledge, the Company and its Affiliate have not misrepresented, or failed to disclose, any facts or circumstances in any application for any Company Registered Intellectual Property that would constitute fraud or a misrepresentation with respect to such application or that would otherwise adversely affect the enforceability of any Company Registered Intellectual Property.

(c) All Contracts relating to Third Party Intellectual Property Rights or Third Party Technology to which the Company or any of its Affiliates is, or purports to be, a party shall continue as such following the Closing on terms identical to those in effect immediately prior to Closing, without obtaining the consent or permission of, or giving notice to, any party to such Contract. Neither the Company nor any Affiliate is in default under any Contract relating to Third Party Intellectual Property Rights or Third Party Technology, and to the Company's Knowledge, no counter party to any Contract relating to Third Party Intellectual Property Rights or Third Party Technology is in default under any such Contract. No proceeding is pending or is being or has been threatened in writing, nor has any claim or demand been made in writing, which challenges the legality, validity, or enforceability of any Contract relating to Third Party Intellectual Property Rights or Third Party Technology to which the Company or any of its Affiliates is, or purports to be, a party. To the Company's Knowledge, no underlying item of Third Party Intellectual Property Rights or Third Party Technology is subject to any Lien that materially interferes with or may reasonably be expected to materially interfere with the rights granted to the Company or any of its Affiliates with respect to such

(d) <u>Schedule 4.13(d)</u> lists all Contracts relating to Third Party Intellectual Property Rights or Third Party Technology, other than any Contract for a non-exclusive license for off-the-shelf software or Technology that is generally available on standard terms for an aggregate fee of less than Two Hundred Fifty Thousand Dollars (\$250,000). None of Atlas Fintech (with respect to the Company Intellectual Property or Company Technology) or the Company or any of its Affiliates is a party to any oral Contract with respect to any Intellectual Property Rights or Technology. None of Atlas Fintech or the Company or any of its Affiliates (A) has provided or made available (or agreed to provide or make available) any source code for any Product or any other Software owned or purported to be owned by the Company or any of its Affiliates, to any Person, and (B) is now or ever has been a member of or party to any patent pool, industry standards body, trade association or other organization under the rules of which it is obligated to license any existing or future Intellectual Property Rights of the Company or any of its Affiliates to any Person or to refrain from enforcing any such rights.

28

(e) There are no material errors, omissions, issues or defects in the Products or any Internal Systems. To the Company's Knowledge, the Products and Internal Systems do not contain any disabling device, virus, worm, back door, cancel bot, remote control, "time bomb," "lock," "drop-dead device," spyware, keyboard logger, Trojan horse, access control or other disruptive or malicious code that may or is intended to impair the performance of or otherwise permit unauthorized access to, hamper, delete or damage the Product, any Internal Systems, or any other computer system, software, network or data. Atlas Fintech and the Company and its Affiliates have taken commercially reasonable actions to maintain, protect and police the integrity and security of the Products and Internal Systems, including such actions designed to protect and police against all unauthorized use of, access to, or "hacking" into the Products, Internal Systems, and all other software, communications, network or computer systems (whether mobile, desktop or enterprise), customer data files, databases, servers and other equipment utilized by the Company or any of its Affiliates. No Person has any right to access or use such source code owned by the Company other than the employees or contractors of the Company's Knowledge, no event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time, or both) will, or would reasonably be expected to, result in the disclosure or release of such source code by the Company or any of its Affiliates, or any other Person to any third party. Without limiting the generality of the foregoing, (i) there have been and are no defects, malfunctions or nonconformities in any Product; (ii)

there have been, and are, no claims asserted against Atlas Fintech or Company or any of its Affiliates or any of its or their customers or distributors related to any Product; and (iii) Atlas Fintech and the Company and its Affiliates have not been or are not required to recall any Products. The Company and its Affiliates maintain commercially reasonable disaster recovery and procedures sufficient for the continued operations of the Company and its Affiliates in the manner as presently conducted and as proposed to be conducted.

(f) Atlas Fintech (with respect to the business of the Company and its Affiliates) and the Company and its Affiliates have used commercially reasonable efforts to protect and enforce their Trade Secrets and otherwise to safeguard and maintain the secrecy and confidentiality of all Company Intellectual Property Rights and Company Technology which by their nature are, or reasonably should be held, confidential, and Atlas Fintech (with respect to the business of the Company and its Affiliates) and the Company and its Affiliates have complied in all material respects with all applicable contractual and legal requirements pertaining to information privacy and security. All current and previous stockholders, officers, employees, independent contractors and consultants of Atlas Fintech and the Company and its Affiliates who have had access to Trade Secrets that are material to the Company's business as currently conducted and as proposed to be conducted or who have otherwise been involved in the development of any Company Owned Intellectual Property Right or Company Owned Technology, have executed and delivered to Atlas Fintech or the Company or its Affiliates (as applicable) agreements that require such Persons to maintain the confidentiality of Trade Secrets and that assign to Atlas Fintech or the Company or its Affiliates, as applicable, all Intellectual Property Rights arising from the services performed for Atlas Fintech or the Company or its Affiliates, as applicable, by such Persons that are material to the business of the Company as currently conducted and proposed to be conducted, except to the extent any such Intellectual Property Rights are owned by the Company or its Affiliates by operation of law. To the Company's Knowledge, there has been no violation of Atlas Fintech's (with respect to the business of the Company and its Affiliates) or the Company's or any of its Affiliates' policies or practices related to the protection of Trade Secrets or of any confidentiality or nondisclosure agreement relating to the Company Intellectual Property Rights or Company Technology, nor has there been any authorized disclosure of any such information or security breach that has resulted in any third party Person obtaining access to any such information, and Atlas Fintech (with respect to the business of the Company and its Affiliates) and the Company and its Affiliates have not received any complaint relating to an improper use or disclosure of, or breach in the security of, any such information. To the Company's Knowledge, no employee, consultant or contractor is in breach of any such agreement described in this Schedule 4.13(f).

29

(g) Except as set forth in <u>Schedule 4.13(g)</u>, no government funding, facilities or resources of a university, college, other educational institution or research center, or funding from third parties was used in the development of any Products, Company Owned Intellectual Property Rights, or Company Owned Technology, and no Governmental Authority, university, college or other educational institution or research center has any claim or right in or to the Products, Company Owned Intellectual Property Rights or Company Owned Technology. To the Company's Knowledge, no current or former employee, consultant or independent contractor of Atlas Fintech or the Company or any of its Affiliates who was involved in, or who contributed to, the creation or development of any material element of any Products, Company Owned Intellectual Property Rights or Company Owned Technology, has performed services for a Governmental Authority, university, college or other educational institution, or a research center, during a period of time during which such employee, consultant or independent contractor was also performing services for Atlas Fintech or the Company or any of its Affiliates.

(h) To the Company's Knowledge, the Exploitation of the Products, the performance of the Services, and the conduct of the Company's or any of its Affiliates' businesses as previously conducted (including as conducted by Atlas Fintech), currently conducted, or as it is proposed to be conducted does not and will not infringe, misappropriate, or otherwise violate, and has not infringed, misappropriated, or otherwise violated, any Intellectual Property Rights of any Person. No action, suit, investigation or claim is pending or, to the Company or any of its Affiliates with regard to any person's right in any Company Intellectual Property Right or Company Technology, including any offer to license the Intellectual Property Rights of any person, allegation of infringement, unauthorized use or misappropriation or of any breach or default of any license or other agreement, or challenge of the ownership, right to Exploit, license or sublicense, validity or enforceability of any Company Intellectual Property Rights or Company Technology.

(i) To the Company's Knowledge, none of the Company Intellectual Property Rights has been or is being infringed, misappropriated, or otherwise violated by any other person. None of Atlas Fintech or the Company or any of its Affiliates has

brought, or threatened to bring, any claims, actions or lawsuits alleging (i) infringement, misappropriation, or other violation of any Intellectual Property Rights or (ii) breach of any Contract that includes a grant of license to or right to access or use any of the Company Intellectual Property Rights or Company Technology, and, to the Company's Knowledge, there do not exist any facts which could form the basis of any such claim, action or lawsuit.

(j) To the Company's Knowledge, at no time during the conception or reduction of any Company Owned Intellectual Property Rights or Company Owned Technology to practice was any developer, inventor or other contributor to any such Intellectual Property Rights or Technology subject to any employment agreement or invention assignment or nondisclosure agreement with or other obligation to any third person. To the Company's Knowledge, none of the activities of current or former employees of Atlas Fintech or the Company or any of its Affiliates are violating or have violated any Contract or other arrangement which any such employee has or had with a former employer.

30

(k) Set forth in <u>Schedule 4.13(k)</u> are all Internet Names used in the Company's or any of its Subsidiaries' business as presently conducted and as proposed to be conducted ("*Company Internet Names*"). The Company is the sole registrant or account holder (as applicable) of all Company Internet Names, and all registrations or accounts of Company Internet Names are in good standing and, with respect to Domain Names included in the Company Internet Names, registered until the dates set forth on <u>Schedule 4.13(k)</u>. No facts or circumstances exist which could reasonably form the basis of a challenge relating to the unencumbered use of the Company Internet Names or any part thereof.

(l) None of Atlas Fintech or the Company or any of its Affiliates have permitted any person to modify, improve or create derivative works of the Products or own any Intellectual Property Rights therein.

(m) No Product is sold, licensed, distributed or otherwise made available by the Company as Open Source Software. Atlas Fintech and the Company and its Affiliates have not incorporated, linked, combined, interfaced, embedded, included, distributed, made available, or otherwise used any Open Source Software in a manner that requires (1) that the Company or any of its Affiliates grant a license under or refrain from asserting or enforcing any of its Patent rights or (2) any proprietary portion of any Product or other Software that is a Company Intellectual Property Right to be (x) disclosed or distributed in source code form, (y) licensed for the purpose of making derivative works, or (z) redistributable at no charge. Except as would not, individually or in the aggregate, have a Material Adverse Effect, Atlas Fintech and the Company and its Affiliates are and have been in compliance with the terms and conditions of all licenses for the Open Source Software listed in (or required to be listed in) Schedule 4.13(a).

4.14 Taxes and Returns.

(a) The Company has or will have timely filed, or caused to be timely filed, all income and other material Tax Returns required to be filed by it (taking into account all available extensions), which Tax Returns are true, accurate, correct and complete in all material respects, and has paid, collected or withheld, or caused to be paid, collected or withheld, all material Taxes required to be paid, collected or withheld, other than such Taxes for which adequate reserves in Company Financials have been established in accordance with GAAP.

(b) There is no Action currently pending or, to the Knowledge of the Company, threatened against the Company by a Governmental Authority in a jurisdiction where the Company does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.

(c) There are no claims, assessments, audits, examinations, investigations or other Actions pending against the Company in respect of any Tax, and the Company has not been notified in writing of any proposed Tax claims or assessments against it (other than, in each case, claims or assessment for which adequate reserves in Company Financials have been established in accordance with GAAP).

(d) There are no Liens with respect to any Taxes upon the Company's assets, other than Permitted Liens.

(e) The Company has no outstanding waivers or extensions of any applicable statute of limitations to assess any amount of Taxes. There are no outstanding requests by the Company for any extension of time (other than automatic extensions) within which to file any Tax Return or within which to pay any Taxes shown to be due on any Tax Return.

(f) The Company has not made any change in accounting method (except as required by a change in Law) or received a ruling from, or signed an agreement with, any taxing authority that would reasonably be expected to have a material impact on its Taxes following the Closing.

(g) The Company has not engaged in any "listed transaction," as defined in U.S. Treasury Regulation section 1.6011-4(b)(2).

(h) The Company has no Liability for the Taxes of another Person (i) under any applicable Tax Law, (ii) as a transferee or successor, or (iii) by contract or indemnity (excluding commercial agreements entered into in the ordinary course of business the primary purpose of which is not the sharing of Taxes). The Company is not a party to or bound by any Tax indemnity agreement, Tax sharing agreement or Tax allocation agreement or similar agreement, arrangement or practice (excluding commercial agreements, arrangements or practices entered into in the ordinary course of business the primary purpose of which is not the sharing of Taxes) with respect to Taxes (including an advance pricing agreement, closing agreement or other agreement relating to Taxes with any Governmental Authority) that will be binding on the Company with respect to any period following the Closing Date.

(i) The Company has not requested, nor is it the subject of or bound by, any private letter ruling, technical advice memorandum, closing agreement or similar ruling, memorandum or agreement with any Governmental Authority with respect to any Taxes, nor is any such request outstanding.

(j) The Company: (i) has not constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of securities (to any Person or entity that is not a member of the consolidated group of which the Company is the common parent corporation) qualifying for, or intended to qualify for, Tax-free treatment under Section 355 of the Code within the two-year period ending on the date hereof; and (ii) is not nor has ever been (A) a U.S. real property holding corporation within the meaning of Section 897(c)(2) of the Code, or (B) a member of any consolidated, combined, unitary or affiliated group of corporations for any Tax purposes other than a group of which the Company is or was the common parent corporation.

(k) To the Knowledge of the Company, there is no fact or circumstance that would reasonably be expected to prevent the Contemplated Transactions from qualifying for the Intended Tax Treatment.

4.15 Real Property. <u>Schedule 4.15</u> contains a complete and accurate list of all premises currently leased or subleased or otherwise used or occupied by the Company for the operation of the business of the Company, and of all current leases, lease guarantees, agreements and documents related thereto, including all amendments, terminations and modifications thereof or waivers thereto (collectively, the "*Company Real Property Leases*"), as well as the current annual rent and term under each Company Real Property Lease. The Company Real Property Lease, a written summary of the material terms of such Company Real Property Leases, and in the case of any oral Company Real Property Lease, a written summary of the material terms of such Company Real Property Lease. The Company Real Property Leases are valid, binding and enforceable in accordance with their terms and are in full force and effect, subject to Enforceability Exceptions. To the Knowledge of the Company, no event has occurred which (whether with or without notice, lapse of time or both or the happening or occurrence of any other event) would constitute a material default on the part of the Company or, to the Knowledge of the Company does not own and has never owned any real property or any interest in real property (other than the leasehold interests in the Company Real Property Leases).

4.16 Personal Property. Each item of Personal Property which is currently owned, used or leased by the Company, or included in the Contributed Assets, with a book value or fair market value of greater than Fifty Thousand Dollars (\$50,000) is set forth on <u>Schedule</u>

<u>4.16</u>, along with, to the extent applicable, a list of lease agreements, lease guarantees, security agreements and other agreements related thereto, including all amendments, terminations and modifications thereof or waivers thereto ("*Company Personal Property Leases*"). Except as would not be material to the Company, or as set forth in <u>Schedule 4.16</u>, all such items of Personal Property are in good operating condition and repair (reasonable wear and tear excepted consistent with the age of such items) and are suitable for their intended use in the business of the Company. The operation of the Company's business as it is now conducted and as proposed to be conducted is not in any material respect dependent upon the right to use the Personal Property of Persons other than the Company, except for such Personal Property that is owned, leased or licensed by or otherwise contracted to the Company. The Company Personal Property Lease, a written summary of the material terms of such Company Personal Property Lease. The Company Personal Property Leases are valid, binding and enforceable in accordance with their terms and are in full force and effect. To the Knowledge of the Company, no event has occurred which (whether with or without notice, lapse of time or both or the happening or occurrence of any other event) would constitute a material default on the part of the Company or, to the Knowledge of the Company, any other party under any of the Company Personal Property Leases, and the Company and other event) would constitute a material default on the part of the Company or, to the Knowledge of the Company, any other party under any of the Company Personal Property Leases, and the Company has not received written notice of any such condition.

4.17 Title to and Sufficiency of Assets. The Company has good and marketable title to, or a valid leasehold interest in or right to use, all of its assets, and Atlas FinTech has good and marketable title to, or a valid leasehold interest in or right to use the Contributed Assets, in each case, free and clear of all Liens other than (a) Permitted Liens, (b) the rights of lessors under leasehold interests, (c) Liens specifically identified on the consolidated balance sheet of the Company as of the Interim Balance Sheet Date, (d) Liens set forth on Schedule 4.17, or (e) as would not be material to the Company. The assets (including Intellectual Property Rights and contractual rights) owned or leased by the Company constitute all of the material assets, rights and properties that are used in the operation of the businesses of the Company as it is now conducted and as it is proposed to be conducted or that are used or held by the Company for use in the operation of the businesses of the Company, and taken together, are adequate and sufficient for the operation of the businesses of the Company as currently conducted and as proposed to be conducted. The Company Intellectual Property Rights and Company Technology constitute all of the Intellectual Property Rights and Technology required for (i) the Exploitation of the Products and Services in the manner so done currently, and (ii) the continued conduct of the business as conducted by Atlas Fintech prior to the Contribution, as presently conducted by the Company and its Affiliates, and as proposed to be conducted by the Company and its Affiliates. The consummation of the Contemplated Transactions hereby will not result in the loss or impairment of the Company's or its Affiliate's right to own or use any Company Intellectual Property Rights or Company Technology. Immediately subsequent to the Closing, the Company Intellectual Property Rights and Company Technology will be owned or available for use on terms and conditions under which the Company or its Affiliate owns or uses (or proposed to use) the Company Intellectual Property Rights and Company Technology immediately prior to Closing, without payment of additional fees. Upon consummation of the Contemplated Transactions, Atlas FinTech will have no right, title or interest in or to any Company Intellectual Property or Company Technology.

33

4.18 Employee Matters.

(a) Except as set forth in <u>Schedule 4.18(a)</u>, the Company is not a party to any collective bargaining agreement or other Contract covering any group of employees, labor organization or other representative of any of the employees of the Company, and the Company has no Knowledge of any activities or proceedings of any labor union or other party to organize or represent such employees. There has not occurred or, to the Knowledge of the Company, been threatened any strike, slow-down, picketing, work-stoppage, or other similar labor activity with respect to any such employees. <u>Schedule 4.18(a)</u> sets forth all unresolved labor controversies (including unresolved grievances and age or other discrimination claims other than any workers' compensation or unemployment claims), if any, that are pending or, to the Knowledge of the Company. No current officer or employee of the Company has provided the Company written or, to the Knowledge of the Company, oral notice of his or her plan to terminate his or her employment with the Company.

(b) Except as set forth in <u>Schedule 4.18(b)</u>, the Company (i) is and has been in compliance in all material respects with all applicable Laws respecting employment and employment practices, terms and conditions of employment, health and safety and wages and hours, and other Laws relating to discrimination, disability, labor relations, hours of work, payment of wages and overtime wages, pay equity, immigration, workers compensation, working conditions, employee scheduling, occupational safety and health, family and medical leave, and employee terminations, and has not received written or, to the Knowledge of the Company, oral notice that there is any pending Action involving unfair labor practices against the Company, (ii) is not liable for any material past due arrears of wages or any material penalty for failure to comply with any of the foregoing, and (iii) is

not liable for any material payment to any Governmental Authority with respect to unemployment compensation benefits, social security or other benefits or obligations for employees, independent contractors or consultants (other than routine payments to be made in the ordinary course of business and consistent with past practice). Except as set forth in <u>Schedule 4.18(b)</u>, there are no material Actions pending or, to the Knowledge of the Company, threatened against the Company brought by or on behalf of any applicant for employment, any current or former employee, any Person alleging to be a current or former employee, or any Governmental Authority, relating to any such Law or regulation, or alleging material breach of any express contract of employment, wrongful termination of employment, or alleging any other discriminatory, wrongful or tortious conduct in connection with the employment relationship.

(c) <u>Schedule 4.18(c)</u> hereto sets forth a complete and accurate list as of the date hereof of all employees of the Company showing for each as of such date (i) the employee's name, job title or description, employer, location, salary or hourly rate; and (ii) wages, bonus, commission or other compensation paid during the fiscal year ending December 31, 2021. Except as set forth on <u>Schedule 4.18(c)</u>, (A) no employee is a party to a written employment Contract with the Company that is not terminable "at will," and (B) the Company has paid in full to all its employees all wages, salaries, commission, bonuses and other compensation due to their employees, including overtime compensation, and the Company has no material obligation or Liability (whether or not contingent) with respect to severance payments to any such employees under the terms of any written or, to the Company's Knowledge, oral agreement, or any applicable Law, custom, trade or practice. Except as set forth in <u>Schedule 4.18(c)</u>, each Company employee has entered into the Company's standard form of employee non-disclosure, inventions and restrictive covenants agreement with the Company (whether pursuant to a separate agreement or incorporated as part of such employee's overall employment agreement), a copy of which has been made available to Purchaser by the Company.

34

(d) <u>Schedule 4.18(d)</u> contains a list of all independent contractors (including consultants) currently engaged by the Company, along with a description of the general nature of the work performed, date of retention and rate of remuneration, most recent increase (or decrease) in remuneration and amount thereof, for each such Person. Except as set forth on <u>Schedule 4.18(d)</u>, all of such independent contractors are a party to a written Contract with the Company. Except as set forth on <u>Schedule 4.18(d)</u>, each such independent contractor has entered into customary covenants regarding confidentiality and assignment of inventions and copyrights in such Person's agreement with the Company, a copy of which has been provided to Purchaser by the Company. For the purposes of applicable Law, including the Code, all independent contractors who are currently, or since the Company's incorporation, engaged by the Company are bona fide independent contractors and not employees of the Company. Each independent contractor is terminable on fewer than thirty (30) days' notice, without any obligation of the Company to pay severance or a termination fee.

4.19 Benefit Plans.

(a) Set forth on <u>Schedule 4.19(a)</u> is a true and complete list of each Benefit Plan of the Company (each, a "*Company Benefit Plan*"). With respect to each Company Benefit Plan, there are no funded benefit obligations for which contributions have not been made or properly accrued and there are no unfunded benefit obligations. The Company is not nor has since its incorporation been a member of a "controlled group" for purposes of Section 414(b), (c), (m) or (o) of the Code, nor does the Company have any Liability with respect to any collectively-bargained for plans, whether or not subject to the provisions of ERISA.

(b) Each Company Benefit Plan is and has been operated, administered, and funded at all times in compliance with all applicable Laws in all material respects, including, but not limited to, ERISA and the Code. Each Company Benefit Plan which is intended to be "qualified" within the meaning of Section 401(a) of the Code (i) has been determined by the IRS to be so qualified (or is based on a prototype plan which has received a favorable opinion letter) during the period from its adoption to the date of this Agreement and (ii) its related trust has been determined to be exempt from taxation under Section 501(a) of the Code or the Company has requested an initial favorable IRS determination of qualification and/or exemption within the period permitted by applicable Law. To the Company's Knowledge, no fact exists which could adversely affect the qualified status of such Company Benefit Plans or the exempt status of such trusts.

(c) With respect to each Company Benefit Plan which covers any current or former officer, director, consultant or employee (or beneficiary thereof) of the Company, the Company has provided to Purchaser accurate and complete copies, if applicable, of: (i) all Company Benefit Plan documents and agreements and related trust agreements or annuity Contracts

(including any amendments, modifications or supplements thereto); (ii) all summary plan descriptions and summary of material modifications thereto; (iii) the three (3) most recent Forms 5500, if applicable, and annual report, including all schedules thereto; (iv) the most recent annual and periodic accounting of plan assets; (v) the three (3) most recent nondiscrimination testing reports; (vi) the most recent determination letter received from the IRS, if any; (vii) the most recent actuarial valuation; and (viii) all material communications with any Governmental Authority within the last three (3) years.

(d) With respect to each Company Benefit Plan: (i) such Company Benefit Plan has been administered and enforced in all material respects in accordance with its terms, the Code and ERISA; (ii) no material breach of fiduciary duty has occurred; (iii) no material Action is pending, or to the Company's Knowledge, threatened (other than routine claims for benefits arising in the ordinary course of administration); (iv) no prohibited transaction, as defined in Section 406 of ERISA or Section 4975 of the Code, has occurred, excluding transactions effected pursuant to a statutory or administration exemption; and (v) all contributions and premiums due through the Closing Date have been made in all material respects as required under ERISA.

35

(e) No Company Benefit Plan is currently, or is reasonably expected to become, a "defined benefit plan" (as defined in Section 414(j) of the Code), a "multiemployer plan" (as defined in Section 3(37) of ERISA) or a "multiple employer plan" (as described in Section 413(c) of the Code) or is otherwise subject to Title IV of ERISA or Section 412 of the Code, and the Company has not incurred any material Liability or otherwise could have any material Liability, contingent or otherwise, under Title IV of ERISA and no condition presently exists that is expected to cause such material Liability to be incurred. No Company Benefit Plan will become a multiple employer plan with respect to the Company immediately after the Closing Date. The Company currently does not maintain, nor has ever maintained, nor is required currently or has ever been required to contribute to or otherwise participate in, a multiple employer welfare arrangement or voluntary employees' beneficiary association as defined in Section 501(c)(9) of the Code.

(f) No arrangement exists pursuant to which the Company will be required to "gross up" or otherwise compensate any person because of the imposition of any excise tax on a payment to such person.

(g) With respect to each Company Benefit Plan which is a "welfare plan" (as described in Section 3(1) of ERISA): (i) no such plan provides medical or death benefits with respect to current or former employees of the Company beyond their termination of employment (other than coverage mandated by Law, which is paid solely by such employees); and (ii) there are no reserves, assets, surplus or prepaid premiums under any such plan. The Company has complied in all material respects with the provisions of Section 601, *et seq.* of ERISA and Section 4980B of the Code.

(h) Except as set forth on <u>Schedule 4.19(h)</u>, the consummation of the Contemplated Transactions will not: (i) entitle any individual to material severance pay, unemployment compensation or other benefits or compensation (except as set forth on <u>Schedule 4.19(h)</u>); (ii) accelerate the time of payment or vesting, or increase the amount of any material compensation due, or in respect of, any individual; or (iii) result in or satisfy a condition to the payment of material compensation that would, in combination with any other payment, result in an "excess parachute payment" within the meaning of Section 280G of the Code. The Company has not incurred any material Liability for any Tax imposed under Chapter 43 of the Code or civil liability under Section 502(i) or (l) of ERISA.

(i) All Company Benefit Plans can be terminated at any time prior to the Closing Date without resulting in any material Liability to the Second Surviving Corporation or Purchaser or their respective Affiliates for any additional and material contributions, penalties, premiums, fees, fines, excise taxes or any other charges or liabilities.

(j) Each Company Benefit Plan that is subject to Section 409A of the Code (each, a "Section 409A Plan") as of the Closing Date is indicated as such on <u>Schedule 4.19(j)</u>. No options or other equity-based awards have been issued or granted by the Company that are, or are subject to, a Section 409A Plan. Each Section 409A Plan has been administered in compliance, and is in documentary compliance, in all material respects, with the applicable provisions of Section 409A of the Code, the regulations thereunder and other official guidance issued thereunder. The Company has no obligation to any employee or other service provider with respect to any Section 409A Plan that may be subject to any Tax under Section 409A of the Code. No payment to be made under any Section 409A Plan is, or to the Knowledge of the Company will be, subject to the penalties of Section 409A(a)(1) of the Code. There is no Contract or plan to which the Company is a party or by which it is bound to compensate any employee, consultant or director for penalty taxes paid pursuant to Section 409A of the Code.

4.20 Environmental Matters.

(a) The Company is and has been in compliance in all material respects with all applicable Environmental Laws, including obtaining, maintaining in good standing, and complying in all material respects with all Permits required for its business and operations by Environmental Laws ("*Environmental Permits*"), no material Action is pending or, to the Company's Knowledge, threatened to revoke, modify, or terminate any such Environmental Permit, and, to the Company's Knowledge, no facts, circumstances, or conditions currently exist that could adversely and materially affect such continued compliance with Environmental Laws and Environmental Permits or require capital expenditures to achieve or maintain such continued compliance with Environmental Laws and Environmental Permits.

(b) The Company is not the subject of any outstanding material Order or Contract with any Governmental Authority or other Person in respect of any (i) Environmental Laws, (ii) Remedial Action, or (iii) Release or threatened Release of a Hazardous Material. The Company has not assumed, contractually or by operation of Law, any material Liabilities or material obligations under any Environmental Laws.

(c) No material Action has been made or is pending, or to the Company's Knowledge, threatened against the Company or any material assets of the Company alleging either or both that the Company may be in material violation of any Environmental Law or Environmental Permit or may have any material Liability under any Environmental Law.

(d) The Company has not manufactured, treated, stored, disposed of, arranged for or permitted the disposal of, generated, handled or released any Hazardous Material, or owned or operated any property or facility, in a manner that has given or would reasonably be expected to give rise to any material Liability or material obligation under applicable Environmental Laws. No fact, circumstance, or condition exists in respect of the Company or any property currently or formerly owned, operated, or leased by the Company or any property to which the Company arranged for the disposal or treatment of Hazardous Materials that could reasonably be expected to result in the Company incurring any material Environmental Liabilities.

(e) There is no investigation of the business, operations, or currently owned, operated, or leased property of the Company or, to the Company's Knowledge, previously owned, operated, or leased property of the Company pending or, to the Company's Knowledge, threatened that could lead to the imposition of any Liens under any Environmental Law or material Environmental Liabilities.

(f) To the Knowledge of the Company, there is not located at any of the properties of the Company any (i) underground storage tanks, (ii) asbestos-containing material, or (iii) equipment containing polychlorinated biphenyls.

(g) The Company has provided to Purchaser all environmentally related site assessments, audits, studies, reports, analysis and results of investigations that have been performed in respect of the currently or previously owned, leased, or operated properties of the Company.

37

4.21 Transactions with Related Persons. Except as set forth on <u>Schedule 4.21</u>, neither the Company nor any of its Affiliates, nor any officer, director, manager, employee, trustee or beneficiary of the Company or any of its Affiliates, nor any immediate family member of any of the foregoing (whether directly or indirectly through an Affiliate of such Person) (each of the foregoing, a "*Related Person*") is presently, or since the Company's incorporation, has been, a party to any transaction with the Company, including any Contract or other arrangement (a) providing for the furnishing of services by (other than as officers, directors or employees of the Company), (b) providing for the rental of real property or Personal Property from or (c) otherwise requiring payments to any Related Person or any Person in which any Related Person has an interest as an owner, officer, manager, director, trustee or partner or in which any Related Person has any direct or indirect interest, in each case, other than (i) the ownership of securities representing no more than two percent (2%) of the outstanding voting power or economic interest of a publicly traded company or (ii) for services or expenses as

directors, officers or employees of the Company in the ordinary course of business consistent with past practice. Except as set forth on <u>Schedule 4.21</u>, the Company has no outstanding Contract or other arrangement or commitment with any Related Person, and no Related Person owns any real property or Personal Property, or right, tangible or intangible (including Intellectual Property Rights) which is used in the business of the Company. The assets of the Company do not include any material receivable or other obligation from a Related Person, and the liabilities of the Company do not include any material payable or other obligation or commitment to any Related Person. Each Contract or other arrangement listed on <u>Schedule 4.21</u> was entered into at arms' length and in the ordinary course of business on commercially reasonable terms and is reasonably necessary for the operation of the business as presently conducted and as currently contemplated to be conducted.

4.22 Insurance. <u>Schedule 4.22</u> lists all material insurance policies (by policy number, insurer, coverage period, coverage amount, annual premium and type of policy) held by the Company relating to the Company or its business, properties, assets, directors, officers and employees, copies of which have been provided to Purchaser. All premiums due and payable under all such insurance policies have been timely paid and the Company are otherwise in material compliance with the terms of such insurance policies. Each such insurance policy (i) is legal, valid, binding, enforceable and in full force and effect and (ii) will continue to be legal, valid, binding, enforceable, and in full force and effect on identical terms following the Closing. The Company has no self-insurance or co-insurance programs. Since incorporation, the Company has not received any written notice from, or on behalf of, any insurance carrier relating to or involving any adverse change or any change other than in the ordinary course of business, in the conditions of insurance, any refusal to issue an insurance policy or non-renewal of a policy.

4.23 Books and Records. All of the financial books and records of the Company, or of Atlas FinTech with respect to the Contributed Assets, are complete and accurate in all material respects and have been maintained in the ordinary course of business consistent with past practice and in accordance with applicable Laws.

4.24 Certain Business Practices.

(a) Neither the Company, nor any of its Representatives acting on their behalf has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 or any other local or foreign anti-corruption or bribery Law or (iii) made any other unlawful payment. Neither the Company, nor any of its Representatives acting on its behalf has directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder the Company or assist the Company in connection with any actual or proposed transaction.

38

(b) The operations of the Company are and have been conducted at all times in compliance with money laundering statutes (including but not limited to the USA PATRIOT Act of 2011, the United States Bank Secrecy Act of 1970, the United States Money Laundering Control Act of 1986, and the International Money Laundering Abatement and Anti-Terrorist Financing Act of 2001, the laws and executive orders administered by state agencies, and any regulations promulgated thereunder (collectively, the "*Anti-Money Laundering Laws*")) in all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority, and no Action involving the Company with respect to any of the foregoing is pending or, to the Knowledge of the Company, threatened.

(c) Neither the Company nor any of its shareholders, directors, or officers, nor, to the Knowledge of the Company, any other Representative acting on behalf of the Company is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any U.S. sanctions administered by OFAC, and the Company has not, since its incorporation, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any Subsidiary, joint venture partner or other Person, in connection with any sales or operations in Cuba, Iran, North Korea, Syria, the Donetsk People's Republic, Luhansk People's Republic, and Crimea regions of Ukraine, or any other country sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC.

(d) Except as specifically authorized by a governmental license, license exception, or other permit or applicable authorization of a Governmental Authority, the Company has not exported, reexported, transferred, facilitated or brokered the sale of any goods, services, technology, or technical data to or from, or entered into any transaction or had any dealing with, any person or entity for whom a license or other authorization is required under the U.S. Export Administration Regulations (the "EAR," 15 C.F.R. § 730 et seq.), the International Traffic in Arms Regulations (the "ITAR," 22 C.F.R. § 120 et seq.) or any other U.S. or non-U.S. export control regime, nor has the Company entered into any transaction prohibited by such laws.

(e) During the past five (5) years, the Company has not received any written or oral communication from any Governmental Authority that alleges that the Company or any of its agents is in violation of, or has, or may have any liability under, any International Trade Laws or Anti-Money Laundering Laws. Further, the Company has not, during the past five (5) years, made any voluntary or involuntary disclosure to a Governmental Authority or conducted any internal investigation or audit concerning any actual or potential violation or wrongdoing related to International Trade Laws or Anti-Money Laundering Laws.

4.25 Compliance with Privacy Laws, Privacy Policies and Certain Contracts.

(a) Neither the Company, nor, to the Knowledge of the Company, any officer, director, manager, employee, agent, subcontractor or vendor to whom the Company has given access to Personal Data or Protected Health Information, is now or has since its incorporation acted or been in material violation of any applicable Privacy Laws.

(b) Except as would not, individually or in the aggregate, have a Material Adverse Effect, to the Knowledge of the Company, the Company has not experienced any material loss, damage or unauthorized access, use, disclosure or modification, or breach of security of Personal Data or Protected Health Information maintained by or on behalf of the Company (including, to the Knowledge of the Company, by any agent, subcontractor or vendor of the Company).

39

(c) Except as would not, individually or in the aggregate, have a Material Adverse Effect, to the Knowledge of the Company, (i) no Person, including any Governmental Authority, has made any written claim or commenced any Proceeding with respect to any violation of any Privacy Law by the Company (ii) the Company has not been given written notice of any criminal, civil or administrative violation of any Privacy Law, in any case including any claim or action with respect to any loss, damage or unauthorized access, use, disclosure, modification, or breach of security, of Personal Data or Protected Health Information maintained by or on behalf of the Company (including by any agent, subcontractor or vendor of the Company).

(d) Neither the Company nor, to the Knowledge of the Company, any subcontractor agent or vendor of the Company, has incurred any material breach of "unsecured protected health information" (as defined in 45 C.F.R. Part 164, Subpart D) requiring reporting to any Governmental Authority.

(e) To the Knowledge of the Company, all activities conducted by the Company with respect to any Protected Health Information or Personal Data are permitted under the Contracts relating to Personal Data or Protected Health Information.

(f) To the Knowledge of the Company, each Contract between the Company and any customer of the Company contains all the terms and conditions that the Company is required to include therein under the Company's Contracts with its vendors and suppliers.

(g) The Company is not an "investment company" or a Person directly or indirectly "controlled" by or acting on behalf of an "investment company," or required to register as an "investment company," in each case within the meaning of the Investment Company Act of 1940, as amended.

4.26 Compliance with Broker Dealer Laws and Regulations.

(a) Since the Company's incorporation, except as would not, individually or in the aggregate, have a Material Adverse Effect, the Company is and has been, duly registered, licensed or qualified as a broker-dealer under the Exchange Act and in each state where the conduct of its business requires such registration, licensing or qualification. The Company is a member in good standing of FINRA, the Securities Investor Protection Corporation (SIPC) and each other Governmental Authority where

membership or association with such Governmental Authority is required in order for the Company to conduct its business as presently conducted and as proposed to be conducted. Except as would not, individually or in the aggregate, have a Material Adverse Effect, the Company is, and has been since the Company's incorporation, in compliance with Law applicable to it, including rules and regulations of FINRA, the SEC and any state securities authorities the jurisdiction of which the Company is subject. As of the date hereof, the Company has not entered into any agreement or arrangement with any Governmental Authority to increase its regulatory capital above the amount required to be maintained under Rule 15c3-1 of the Exchange Act.

(b) Since the Company's incorporation, except as would not, individually or in the aggregate, have a Material Adverse Effect, the Company has filed on a timely basis all regulatory reports, schedules, forms, registrations, financial statements, notices, and other documents, together with any material amendments that were required to be filed by it with FINRA or the SEC or otherwise required to be filed under the Securities Act, the Exchange Act, or FINRA's rules or its Bylaws, including all FOCUS reports, Form BD and all such reports required pursuant to Rule 17a-5 under the Exchange Act. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, all such filings, including the Form BD of the Company, were, at the time they were filed, in compliance with applicable Law.

40

(c) The Company has adopted and, at all times since its incorporation, has had in place and have been in compliance with written supervisory control policies and procedures, which have been implemented pursuant to and in accordance with Law applicable to them, including but limited to Rule 30(a) of Regulation S-P (17 C.F.R. § 248.30(a)), FINRA Rule 3110(b) and FINRA Rule 3120(a), (the "*Broker-Dealer Compliance Policies*") except in each case where the failure to adopt, have in place, implement or comply with such Broker-Dealer Compliance Policies would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. Except as set forth in any of the Company's annual or other period compliance with the Broker-Dealer Compliance Policies, except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

4.27 Finders and Brokers. Except as set forth in <u>Schedule 4.27</u>, the Company has not incurred nor will incur any Liability for any brokerage, finder's or other fee or commission in connection with the Contemplated Transactions.

4.28 Independent Investigation. The Company has conducted its own independent investigation, review and analysis of the business, results of operations, prospects, condition (financial or otherwise) or assets of Purchaser, and acknowledges that it has been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of Purchaser for such purpose. The Company acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the Contemplated Transactions, it has relied solely upon its own investigation and the express representations and warranties of Purchaser set forth in Agreement (including the related portions of Purchaser Disclosure Schedules) and in any certificate delivered to the Company pursuant hereto; and (b) neither Purchaser nor any of its Representatives have made any representation or warranty as to Purchaser or this Agreement, except as expressly set forth in <u>Article III</u> of this Agreement (including the related portions of Purchaser Disclosure Schedules) or in any certificate delivered to the Company pursuant hereto.

4.29 Information Supplied. None of the information supplied or to be supplied by the Company expressly for inclusion or incorporation by reference: (a) in any current report on Form 8-K, and any exhibits thereto or any other report, form, registration or other filing made with any Governmental Authority or stock exchange with respect to the transactions Contemplated Transactions; (b) in the Registration Statement/ Proxy Statement; or (c) in the mailings or other distributions to Purchaser's stockholders and/or prospective investors with respect to the consummation of the Contemplated Transactions or in any amendment to any of documents identified in (a) through (c), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. None of the information supplied or to be supplied by the Company expressly for inclusion or incorporation by reference in any of the Signing Press Release, the Signing Filing, the Closing Press Release and the Closing Filing will, when filed or distributed, as applicable, contain any untrue statement of a material fact required to be stated therein or necessary in order to a material fact or omit to state any material fact required to make the statements therein, in light of the closing Filing will, when filed or distributed, as applicable, contain any untrue statement of a material fact or omit to state any material fact required to make the statements therein, in light of the circumstances under which they are made, not misleading. Notwithstanding the foregoing, the Company makes no representation, warranty or covenant with respect to any information supplied by or on behalf of Purchaser, either Acquisition Target or any of their respective Affiliates.

ARTICLE V. REPRESENTATIONS AND WARRANTIES OF THE STOCKHOLDERS

Each Company Stockholder hereby represents and warrants to Purchaser with respect to itself, as of the date of this Agreement, as follows:

5.1 Organization and Standing. In the case of Atlas FinTech, such Company Stockholder is duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted.

5.2 Authorization; Binding Agreement. Such Company Stockholder has all requisite power (or corporate power in the case of Atlas FinTech) and authority to execute and deliver this Agreement and each Ancillary Document to which it is or is required to be a party, to perform such Company Stockholder's obligations hereunder and thereunder and to consummate the Contemplated Transactions. In the case of Atlas FinTech, the execution and delivery of this Agreement and each Ancillary Document to which such Company Stockholder is or is required to be a party and the consummation of the Contemplated Transactions, (a) have been duly and validly authorized by such Company Stockholder's board of directors in accordance with such Company Stockholder's Organizational Documents, the DGCL, any other applicable Law or any Contract to which such Company Stockholder or any of its stockholders is a party or by which it or its securities are bound and (b) no other corporate proceedings on the part of such Company Stockholder are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party or to consummate the Contemplated Transactions. This Agreement has been, and each Ancillary Document to which such Company Stockholder is or is required to be a party shall be when delivered, duly and validly executed and delivered by such Company Stockholder and assuming the due authorization, execution and delivery of this Agreement and any such Ancillary Document by the other parties hereto and thereto, constitutes, or when delivered shall constitute, the legal, valid and binding obligation of such Company Stockholder, enforceable against such Company Stockholder in accordance with its terms, subject to the Enforceability Exceptions. In the case of Atlas FinTech, such Company Stockholder's board of directors, by resolutions duly adopted at a meeting duly called and held or by action by unanimous written consent in accordance with such Stockholder's Organizational Documents (i) determined that this Agreement and the Second Merger and the Contemplated Transactions by are advisable, fair to, and in the best interests of, such Company Stockholder and its stockholders, and (ii) approved this Agreement and the Second Merger and the Contemplated Transactions in accordance with the WBCA.

5.3 Title to Equity Interests. Such Company Stockholder is the sole and lawful record and beneficial owner of the Company Stock listed as owned by such Company Stockholder on <u>Schedule 4.3(a)</u>, and such Company Stockholder has good, valid and marketable title to such Company Stock, free and clear of all Liens other than restrictions on transfer under federal and state securities Laws. The Contemplated Transactions will result in Purchaser obtaining good and valid title, free and clear of all Liens, to such Company Stockholder's Company Stock. There are no Contracts with respect to the acquisition, disposition or voting of, or any other matters pertaining to, any of such Company Stockholder's Company Stock or any other equity interests of the Company. Other than the Company Stock, such Company Stockholder does not, directly or indirectly hold any equity interest in the Company or any option, warrant or other security convertible into or exercisable for any equity interest in the Company.

5.4 Governmental Approvals. Except as may be required under federal and state securities Laws, no Consent of or with any Governmental Authority on the part of such Company Stockholder is required to be obtained or made in connection with the execution, delivery or performance by such Company Stockholder of this Agreement or any Ancillary Documents or the consummation by such Company Stockholder of the Contemplated Transactions.

5.5 Non-Contravention. The execution and delivery by such Company Stockholder of this Agreement and each Ancillary Document to which such Company Stockholder is or is required to be a party or otherwise bound, and the consummation by such Company Stockholder of the Contemplated Transactions and compliance by such Company Stockholder with any of the provisions hereof and thereof, will not (a) conflict with or violate any provision of the Atlas FinTech's Organizational Documents (in the case of Atlas FinTech), (b) conflict with or violate any Law, Order or Consent applicable to such Company Stockholder or any of its material properties

or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by such Company Stockholder under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien upon any of the properties or assets of such Company Stockholder under (other than Permitted Liens), (viii) give rise to any obligation to obtain any third party Consent or provide any notice to any Person or (ix) give any Person the right to declare a default, exercise any remedy, claim a rebate, chargeback, penalty or change in delivery schedule, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of any material Contract to which such Company Stockholder is a party.

5.6 Finders and Brokers. Such Company Stockholder has not incurred nor will incur any Liability for any brokerage, finder's or other fee or commission in connection with the Contemplated Transactions.

ARTICLE VI. COVENANTS

6.1 Access and Information.

(a) During the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement in accordance with Section 9.1 or the Closing (the "Interim Period"), subject to Section 6.15, the Company shall give and shall cause its Representatives to give Purchaser and its Representatives, at reasonable times during normal business hours and upon reasonable intervals and notice, reasonable access to Contracts, agreements, books and records, financial and operating data and other information (including Tax Return, internal working papers, client Contracts and director service agreements), of or pertaining to the Target Companies and the Contributed Assets, as Purchaser or its Representatives may reasonably request regarding the Target Companies, the Contributed Assets and their respective businesses, assets, Liabilities, financial condition, prospects, operations, management, employees and other aspects (including unaudited quarterly financial statements, including a consolidated quarterly balance sheet and income statement, a copy of each material report, schedule and other document filed with or received by a Governmental Authority pursuant to the requirements of applicable securities Laws, and independent public accountants' work papers (subject to the consent or any other conditions required by such accountants, if any)) and cause the Company's Representatives to reasonable best efforts to cause Atlas Fintech and each of the other Target Company's respective Representatives to reasonable cooperate with Purchaser and its Representatives in such a manner as not to unreasonably interfere with the business or operations of Atlas Fintech or the Target Companies.

43

(b) During the Interim Period, subject to <u>Section 6.15</u>, Purchaser shall give, and shall cause its Representatives to give, the Company and its Representatives, at reasonable times during normal business hours and upon reasonable intervals and notice, reasonable access to all offices and other facilities and to all employees, properties, Contracts, agreements, commitments, books and records, financial and operating data and other information (including Tax Returns, internal working papers, client files, client Contracts and director service agreements), of or pertaining to Purchaser or its Subsidiaries, as the Company or its Representatives may reasonably request regarding Purchaser, its Subsidiaries and their respective businesses, assets, Liabilities, financial condition, prospects, operations, management, employees and other aspects (including unaudited quarterly financial statements, including a consolidated quarterly balance sheet and income statement, a copy of each material report, schedule and other document filed with or received by a Governmental Authority pursuant to the requirements of applicable securities Laws, and independent public accountants' work papers (subject to the consent or any other conditions required by such accountants, if any)) and cause each of Purchaser's Representatives to reasonably cooperate with the Company and its Representatives in their investigation; *provided, however*; that the Company and its Representatives shall conduct any such activities in such a manner as not to unreasonably interfere with the business or operations of Purchaser or any of its Subsidiaries.

6.2 Conduct of Business of the Company.

(a) Unless Purchaser shall otherwise consent in writing (such consent not to be unreasonably withheld, conditioned or delayed), during the Interim Period, except as expressly contemplated by this Agreement or the Ancillary Documents or as set forth on <u>Schedule 6.2</u>, the Company shall (i) conduct its business, in all material respects, in the ordinary course of business

consistent with past practice, (ii) comply with all Laws applicable to it and its business, assets and employees, and (iii) take all commercially reasonable measures necessary or appropriate to preserve intact, in all material respects, its business organization, to keep available the services of its managers, directors, officers, employees and consultants, and to preserve the possession, control and condition of its material assets, all as consistent with past practice.

(b) Without limiting the generality of <u>Section 6.2(a)</u> and except as contemplated by the terms of this Agreement or the Ancillary Documents as set forth on <u>Schedule 6.2</u>, during the Interim Period, without the prior written consent of Purchaser (such consent not to be unreasonably withheld, conditioned or delayed), the Company shall not:

(i) amend, waive or otherwise change, in any respect, its Organizational Documents, except as required by applicable Law;

(ii) authorize for issuance, issue, grant, sell, pledge, dispose of or propose to issue, grant, sell, pledge or dispose of any of its equity securities or any options, warrants, commitments, subscriptions or rights of any kind to acquire or sell any of its equity securities, or other securities, including any securities convertible into or exchangeable for any of its shares or other equity securities or securities of any class and any other equity-based awards;

(iii) split, combine, recapitalize or reclassify any of its shares or other equity interests or issue any other securities in respect thereof or pay or set aside any dividend or other distribution (whether in cash, equity or property or any combination thereof) in respect of its equity interests, or directly or indirectly redeem, purchase or otherwise acquire or offer to acquire any of its securities (except for the repurchase of Company Common Stock from former employees, non-employee directors and consultants in accordance with agreements as in effect on the date hereof providing for the repurchase of shares in connection with any termination of service);

44

(iv) incur, create, assume, prepay or otherwise become liable for any Indebtedness (directly, contingently or otherwise) in excess of \$100,000 individually or \$250,000 in the aggregate, make a loan or advance to or investment in any third party (other than advancement of expenses to employees in the ordinary course of business), or guarantee or endorse any Indebtedness, Liability or obligation of any Person in excess of \$100,000 individually or \$250,000 in the aggregate;

(v) increase the wages, salaries or compensation of its employees other than in the ordinary course of business, consistent with past practice, and in any event not in the aggregate by more than five percent (5%), or make or commit to make any bonus payment (whether in cash, property or securities) other than in the ordinary course of business consistent with past practice, to any employee, or materially increase other benefits of employees generally other than in the ordinary course of business consistent with past practice, or enter into, establish, materially amend or terminate any Company Benefit Plan with, for or in respect of any current consultant, officer, manager director or employee, in each case other than as required by applicable Law, pursuant to the terms of any Company Benefit Plans or in the ordinary course of business consistent with past practice;

(vi) make or rescind any material election relating to Taxes, settle any claim, action, suit, litigation, proceeding, arbitration, investigation, audit or controversy relating to material Taxes, file any amended Tax Return or claim for refund, or make any material change in its accounting or Tax policies or procedures, in each case except as required by applicable Law or in compliance with GAAP;

(vii) transfer or license to any Person or otherwise extend, materially amend or modify, permit to lapse or fail to preserve any material registered patents, trademarks, domain names and copyrights and applications for any of the foregoing that have been filed with the applicable governmental authority that are owned or purported to be owned by the Company and its Subsidiaries, all Intellectual Property rights owned or purported to be owned by a third party and licensed to a Company or its Subsidiaries or to which a Company or its Subsidiaries and Intellectual Property rights owned or purported to be owned by a third party and licensed to a Company or its Subsidiaries or to which a third party and licensed to a Company or its Subsidiaries or to which a Company or its Subsidiaries otherwise has a right to use ("Company IP") (excluding non-exclusive licenses

of Company IP to customers in the ordinary course of business consistent with past practice), or disclose to any Person who has not entered into a confidentiality agreement any Trade Secrets;

(viii) terminate, or waive or assign any material right under, any Company Material Contract or enter into any Contract that would be a Company Material Contract, in any case outside of the ordinary course of business consistent with past practice;

(ix) fail to maintain its books, accounts and records in all material respects in the ordinary course of business consistent with past practice;

(x) establish any Subsidiary or enter into any new line of business;

45

(xi) fail to use reasonable best efforts to keep in force material insurance policies or replacement or revised policies providing insurance coverage with respect to its assets, operations and activities in such amount and scope of coverage substantially similar to that which is currently in effect;

(xii) revalue any of its material assets or make any material change in accounting methods, principles or practices, except to the extent required to comply with GAAP and after consulting with the Company's outside auditors;

(xiii) waive, release, assign, settle or compromise any claim, action or proceeding (including any suit, action, claim, proceeding or investigation relating to this Agreement or the Contemplated Transactions), other than waivers, releases, assignments, settlements or compromises that involve only the payment of monetary damages (and not the imposition of equitable relief on, or the admission of wrongdoing by, the Company or its Affiliates) not in excess of \$50,000 (individually or \$100,000 in the aggregate), or otherwise pay, discharge or satisfy any Actions, Liabilities or obligations;

(xiv) close or materially reduce its activities, or effect any layoff or other personnel reduction or change, at any of its facilities;

(xv) other than the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement and the contribution of Contributed Assets, acquire, including by merger, consolidation, acquisition of equity interests or assets, or any other form of business combination, any corporation, partnership, limited liability company, other business organization or any division thereof, or any material amount of assets outside the ordinary course of business consistent with past practice;;

(xvi) make capital expenditures in excess of \$100,000 (individually for any project (or set of related projects) or \$250,000 in the aggregate);

(xvii) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization;

(xviii) voluntarily incur any Liability or obligation (whether absolute, accrued, contingent or otherwise) in excess of \$100,000 individually or \$250,000 in the aggregate other than pursuant to the terms of a Company Material Contract or Company Benefit Plan;

(xix) sell, lease, license, transfer, exchange or swap, mortgage or otherwise pledge or encumber (including securitizations), or otherwise dispose of any material portion of its properties, assets or rights outside of the ordinary course of business consistent with past practice;

(xx) enter into any agreement, understanding or arrangement with respect to the voting of equity securities of the Company;

(xxi) take any action that would reasonably be expected to significantly delay or impair the obtaining of any Consents of any Governmental Authority to be obtained in connection with this Agreement;

(xxii) accelerate the collection of any trade receivables or delay the payment of trade payables or any other liabilities other than in the ordinary course of business consistent with past practice;

(xxiii) enter into, amend, waive or terminate (other than terminations in accordance with their terms) any transaction with any Related Person (other than compensation and benefits and advancement of expenses, in each case, provided in the ordinary course of business consistent with past practice);

(xxiv) consent to the taking of any of the foregoing actions with respect to the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, waive compliance with any provision thereof or consent to the taking of any action otherwise prohibited thereunder; or

(xxv) authorize or agree to do any of the foregoing actions;

provided, that any actions reasonably taken in good faith by the Company or its Subsidiaries to the extent reasonably believed to be necessary to comply with Laws (including orders of Governmental Authorities) related to COVID-19 shall be deemed not to constitute a breach of the requirements set forth under this <u>Section 6.2</u>. The Company shall notify Purchaser in writing of any such actions taken in accordance with the foregoing proviso and shall use reasonable best efforts to mitigate any negative effects of such actions on the business of the Company, in consultation with Purchaser whenever practicable.

6.3 Conduct of Business of Purchaser.

(a) Unless the Company shall otherwise consent in writing (such consent not to be unreasonably withheld, conditioned or delayed), during the Interim Period, except as expressly contemplated by this Agreement or the Ancillary Documents or as set forth on <u>Schedule 6.3</u>, Purchaser shall, and shall cause its Subsidiaries to, (i) conduct their respective businesses, in all material respects, in the ordinary course of business consistent with past practice, (ii) comply with all Laws applicable to Purchaser and its Subsidiaries and their respective businesses, assets and employees, and (iii) take all commercially reasonable measures necessary or appropriate to preserve intact, in all material respects, their respective business organizations, to keep available the services of their respective managers, directors, officers, employees and consultants, and to preserve the possession, control and condition of their respective material assets, all as consistent with past practice. Notwithstanding anything to the contrary in this <u>Section 6.3</u>, nothing in this Agreement shall prohibit or restrict Purchaser from extending, in accordance with the Purchaser's Organizational Documents and the IPO Prospectus (including soliciting its stockholders to approve such amendment) to extend the deadline by which it must complete its Business Combination (an "*Extension*"), and no consent of any other Party shall be required in connection therewith.

(b) Without limiting the generality of <u>Section 6.3(a)</u> and except as contemplated by the terms of this Agreement or the Ancillary Documents or as set forth on <u>Schedule 6.3</u>, during the Interim Period, without the prior written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed), Purchaser shall not, and shall cause its Subsidiaries to not:

(i) amend, waive or otherwise change, in any respect, its Organizational Documents except as required by applicable Law;

(ii) authorize for issuance, issue, grant, sell, pledge, dispose of or propose to issue, grant, sell, pledge or dispose of any of its equity securities or any options, warrants, commitments, subscriptions or rights of any kind to

acquire or sell any of its equity securities, or other securities, including any securities convertible into or exchangeable for any of its equity securities or other security interests of any class and any other equity-based awards, other than (x) the issuance of Purchaser securities issuable upon conversion or exchange of outstanding Purchaser securities in accordance with their terms, (y) engaging in any hedging transaction with a third Person with respect to such securities, or (z) issuing Purchaser Private Warrants to Purchaser's Affiliates or stockholders in satisfaction of the Working Capital Loans incurred in compliance with this Agreement;

(iii) split, combine, recapitalize or reclassify any of its shares or other equity interests or issue any other securities in respect thereof or pay or set aside any dividend or other distribution (whether in cash, equity or property or any combination thereof) in respect of its shares or other equity interests, or directly or indirectly redeem, purchase or otherwise acquire or offer to acquire any of its securities;

(iv) incur, create, assume, prepay or otherwise become liable for any Indebtedness (directly, contingently or otherwise) in excess of \$100,000 individually or \$250,000 in the aggregate, make a loan or advance to or investment in any third party, or guarantee or endorse any Indebtedness, Liability or obligation of any Person (provided, that this Section 6.3(b)(iv) shall not prevent Purchaser from (x) borrowing funds necessary to finance its ordinary course administrative costs and expenses and any Purchaser Transaction Expenses incurred in connection with the consummation of the Mergers and the Contemplated Transactions (including the costs and expenses necessary for an Extension (such expenses, "*Extension Expenses*") or (y) incurring Working Capital Loans);

(v) make or rescind any material election relating to Taxes, settle any claim, action, suit, litigation, proceeding, arbitration, investigation, audit or controversy relating to material Taxes, file any amended Tax Return or claim for refund, or make any material change in its accounting or Tax policies or procedures, in each case except as required by applicable Law or in compliance with GAAP;

(vi) amend, waive or otherwise change the Trust Agreement in any manner adverse to Purchaser;

(vii) terminate, waive or assign any material right under any Purchaser Material Contract;

(viii) fail to maintain its books, accounts and records in all material respects in the ordinary course of business consistent with past practice;

(ix) establish any Subsidiary or enter into any new line of business;

(x) fail to use commercially reasonable efforts to keep in force insurance policies or replacement or revised policies providing insurance coverage with respect to its assets, operations and activities in such amount and scope of coverage substantially similar to that which is currently in effect;

48

(xi) revalue any of its material assets or make any material change in accounting methods, principles or practices, except to the extent required to comply with GAAP and after consulting Purchaser's outside auditors;

(xii) waive, release, assign, settle or compromise any claim, action or proceeding (including any suit, action, claim, proceeding or investigation relating to this Agreement or the Contemplated Transactions), other than waivers, releases, assignments, settlements or compromises that involve only the payment of monetary damages (and not the imposition of equitable relief on, or the admission of wrongdoing by, Purchaser or its Subsidiary) not in excess of \$100,000 individually, or \$250,000 in the aggregate, or otherwise pay, discharge or satisfy any Actions, Liabilities or obligations, unless such amount has been reserved in Purchaser Financials;

(xiii) acquire, including by merger, consolidation, acquisition of equity interests or assets, or any other form of business combination, any corporation, partnership, limited liability company, other business organization or any division thereof, or any material amount of assets outside the ordinary course of business;

(xiv) make capital expenditures in excess of \$100,000 (individually for any project (or set of related projects) or \$250,000 in the aggregate);

(xv) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization;

(xvi) voluntarily incur any Liability or obligation (whether absolute, accrued, contingent or otherwise) in excess of \$100,000 individually or \$250,000 in the aggregate (excluding the incurrence of any Purchaser Transaction Expenses) other than pursuant to the terms of a Contract in existence as of the date of this Agreement or entered into in the ordinary course of business or in accordance with the terms of this <u>Section 6.3</u> during the Interim Period;

(xvii) sell, lease, license, transfer, exchange or swap, mortgage or otherwise pledge or encumber (including securitizations), or otherwise dispose of any material portion of its properties, assets or rights;

(xviii) enter into any agreement, understanding or arrangement with respect to the voting of Purchaser Securities;

(xix) take any action that would reasonably be expected to significantly delay or impair the obtaining of any Consents of any Governmental Authority to be obtained in connection with this Agreement; or

(xx) authorize or agree to do any of the foregoing actions; provided, that any actions reasonably taken in good faith by Purchaser or its Subsidiaries to the extent reasonably believed to be necessary to comply with Laws (including orders of Governmental Authorities) related to COVID-19 shall be deemed not to constitute a breach of the requirements set forth under this <u>Section 6.3</u>. The Purchaser shall notify the Company in writing of any such actions taken in accordance with the foregoing proviso and shall use reasonable best efforts to mitigate any negative effects of such actions on Purchaser and its Subsidiaries.

49

6.4 Annual and Interim Financial Statements.

(a) Atlas Fintech and the Company shall deliver to Purchaser, as promptly as reasonably practicable following the date of this Agreement and in no event later than December 15, 2022, all the financial statements of the Company, the Target Broker-Dealer and the Target Bank that are required by the applicable accounting requirements and other rules and regulations of the SEC to be included in the Registration Statement/ Proxy Statement (including pro forma financial information) in such form as required by the rules and regulations of the SEC (the "*Required Company Financial Statements*").

(b) Following any "staleness" date (as determined in accordance with the applicable rules and regulations of the SEC) applicable to the financial statements that are required by the applicable accounting requirements and other rules and regulations of the SEC to be included in the Registration Statement/ Proxy Statement (including pro forma financial information) that occurs prior to the Closing Date, the Company shall deliver as promptly as practicable, any financial statements of the Company, the Target-Broker Dealer and the Target Bank that are required by the applicable accounting requirements and other rules and regulations of the SEC to be included in the Registration Statement/ Proxy Statement (including pro forma financial information) that occurs are regulations of the SEC to be included in the Registration Statement/ Proxy Statement (including pro forma financial information) (such audited or unaudited financial statements, the "*Additional Financial Statements*").

(c) The Required Company Financial Statements and the Additional Financial Statements (i) will be prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except, in the case of any audited financial statements, as may be specifically indicated in the notes thereto and subject, in the case of any unaudited financial statements, to normal year-end audit adjustments (none of which is expected to be individually or in the aggregate material) and the absence of notes thereto), (ii) will fairly present, in all material respects, the financial position, results of operations and comprehensive loss, shareholders' deficit and cash flows of the applicable entity as of the dates thereof and for the periods indicated therein (subject, in the case of any unaudited financial statements, to normal year-end audit adjustments (none of which is expected to be individually or in the aggregate material)), (iii) in the case of any audited financial statements, will be audited to the extent required by applicable rules and regulations of the SEC in accordance with the standards of the PCAOB and will contain an unqualified report of the applicable entity's auditors and (iv) will comply in all material respects with the applicable

accounting requirements and with the rules and regulations of the SEC, the Exchange Act and the Securities Act in effect as of the respective dates of delivery (including Regulation S-X or Regulation S-K, as applicable).

(d) The Company shall use its reasonable best efforts (i) to assist New Pubco and Purchaser in causing to be prepared in a timely manner any other financial information or statements (including customary pro forma financial statements) that are required to be included in the Registration Statement / Proxy Statement and any other filings to be made by New Pubco or Purchaser with the SEC in connection with the Contemplated Transactions and (ii) to obtain the consents of the auditors of the Company, Target Broker-Dealer and Target Bank, as applicable, with respect thereto as may be required by applicable Law or requested by the SEC.

6.5 Purchaser Public Filings. During the Interim Period, Purchaser will use reasonable best efforts to keep current and timely file all of its public filings with the SEC and otherwise comply in all material respects with applicable securities Laws and shall use its reasonable best efforts prior to the Closing to maintain the listing of Purchaser Units, Purchaser Common Stock and Purchaser Public Warrants on NYSE; *provided*, that the Parties acknowledge and agree that from and after the Closing, the Parties intend to list on NYSE only the New Pubco Common Stock and New Pubco Public Warrants.

50

6.6 No Solicitation.

(a) For purposes of this Agreement, (i) an "*Acquisition Proposal*" means any inquiry, proposal or offer, or any indication of interest in making an offer or proposal, from any Person or group at any time relating to an Alternative Transaction, and (ii) an "*Alternative Transaction*" means (A) with respect to the Company and its Affiliates, a transaction (other than the Contemplated Transactions) concerning the sale of (x) all or any material part of the business or assets of the Company (other than in the ordinary course of business consistent with past practice) or (y) any of the shares or other equity interests, assets, merger, consolidation, issuance of debt securities, management Contract, joint venture or partnership, or otherwise and (B) with respect to Purchaser and its Affiliates, a transaction (other than the Contemplated Transactions) concerning a Business Combination involving Purchaser.

(b) During the Interim Period, in order to induce the other Parties to continue to commit to expend management time and financial resources in furtherance of the Contemplated Transactions, each Party shall not, and shall cause its Representatives to not, without the prior written consent of the Company and Purchaser, directly or indirectly, (i) solicit, assist, initiate or facilitate the making, submission or announcement of, or intentionally encourage, any Acquisition Proposal, (ii) furnish any non-public information regarding such Party or its Affiliates or their respective businesses, operations, assets, Liabilities, financial condition, prospects or employees to any Person or group (other than a Party to this Agreement or their respective Representatives) in connection with or in response to an Acquisition Proposal, (iii) engage or participate in discussions or negotiations with any Person or group with respect to, or that could reasonably be expected to lead to, an Acquisition Proposal, (iv) approve, endorse or recommend, or publicly propose to approve, endorse or recommend, any Acquisition Proposal, (v) negotiate or enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement related to any Acquisition Proposal, or (vi) release any third Person from, or waive any provision of, any confidentiality agreement to which such Party is a party.

(c) Each Party shall notify the others as promptly as practicable (and in any event within 48 hours) in writing of the receipt by such Party or any of its Representatives (and in any event within forty-eight (48) hours) of (i) any bona fide inquiries, proposals or offers, requests for information or requests for discussions or negotiations regarding or constituting any Acquisition Proposal or any bona fide inquiries, proposals or offers, requests for information Proposal, and (ii) any request for non-public information relating to such Party or its Affiliates in connection with any Acquisition Proposal, specifying in each case, the material terms and conditions thereof (including a copy thereof if in writing or a written summary thereof if oral) and the identity of the party making such inquiries, proposals, offers or requests for information. During the Interim Period, each Party shall, and shall cause its Representatives to, immediately cease and cause to be terminated any solicitations, discussions or negotiations with any Person with respect to any Acquisition Proposal and shall, and shall direct its Representatives to, cease and terminate any such solicitations, discussions or negotiations.

6.7 No Trading. Each of Company and the Company Stockholders acknowledges and agrees that he or it is aware of the restrictions imposed by U.S. federal securities laws and the rules and regulations of the SEC and NYSE promulgated thereunder or otherwise (the "*Federal Securities Laws*") and other applicable foreign and domestic Laws on a Person possessing material nonpublic information about a publicly traded company. Each of the Company and the Company Stockholders hereby agrees that, while he or it is in possession of such material nonpublic information, he or it shall not purchase or sell any securities of Purchaser (other than to engage in the Second Merger in accordance with <u>Article I</u>), communicate such information to any third party (other than its officers, directors employees, advisors and other representatives), take any other action with respect to Purchaser in violation of such Laws, or cause or encourage any third party to do any of the foregoing.

6.8 Notification of Certain Matters. During the Interim Period, each Party shall give prompt notice to the other Parties if such Party or its Affiliates: (a) fails to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it or its Affiliates hereunder; (b) receives any notice or other communication in writing from any third (3rd) party (including any Governmental Authority) alleging (i) that the Consent of such third (3rd) party is or may be required in connection with the Contemplated Transactions, or (ii) any non-compliance with any Law by such person or its Affiliates; (c) receives any notice or other communication from any Governmental Authority in connection with the Contemplated Transactions; (d) discovers any fact or circumstance that, or becomes aware of the occurrence or non-occurrence of any event the occurrence or non-occurrence of which, would reasonably be expected to cause or result in any of the conditions to the Closing set forth in this Agreement not being satisfied or the satisfaction of those conditions being materially delayed; or (e) becomes aware of the commencement or threat, in writing, of any Action against such person or any of its Affiliates, or any of their respective properties or assets, or, to the Knowledge of such person, any officer, director, partner, member or manager, in his, her or its capacity as such, of such person or of its Affiliates with respect to the consummation of the Contemplated Transactions. No such notice shall constitute an acknowledgement or admission by the Party providing the notice regarding whether or not any of the conditions to the Closing have been satisfied or in determining whether or not any of the representations, warranties or covenants contained in this Agreement have been breached.

6.9 Efforts.

(a) Subject to the terms and conditions of this Agreement, each Company Stockholders and each Party shall use his or its reasonable best efforts to, and shall cooperate fully with the other Parties to, take, or cause to be taken, all actions and do, or cause to be done, all things reasonably necessary, proper or advisable under applicable Laws and regulations to consummate the Contemplated Transactions (including the receipt of all applicable Consents of Governmental Authorities) and comply as promptly as practicable with all requirements of Governmental Authorities applicable to the Contemplated Transactions.

(b) Subject to the terms and conditions of this Agreement, the Company and each of the Company Stockholders shall use his or its respective reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary, proper or advisable under applicable Laws and regulations to consummate the Contemplated Transactions (including the receipt of all applicable Consents of Governmental Authorities) and to comply as promptly as practicable with all requirements of Governmental Authorities applicable to the Contemplated Transactions. The Company shall promptly as practicable (and in any event within 48 hours) notify Purchaser in writing of (i) any breach or default of the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, or giving of notice or both would constitute such a breach or default, or permit the termination of, the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, or (iii) the amendment of any terms of the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, or (iii) the amendment of any terms of the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, or (iii) the amendment of any terms of the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, or the Contribution Agreement, or the Contribution Agreement.

(c) In furtherance and not in limitation of <u>Section 6.9(a)</u>, to the extent required under any Laws that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade ("*Antitrust Laws*"), each Party hereto agrees to make any required filing or application under Antitrust Laws, as applicable, at such Party's sole

cost and expense (except with respect to the filing fees related thereto, which shall be paid by Purchaser), with respect to the Contemplated Transactions as promptly as practicable, to supply as promptly as reasonably practicable any additional information and documentary material that may be reasonably requested pursuant to Antitrust Laws and to take all other actions reasonably necessary, proper or advisable to cause the expiration or termination of the applicable waiting periods under Antitrust Laws as soon as practicable, including by requesting early termination of the waiting period provided for under the Antitrust Laws. Each Party shall, in connection with its efforts to obtain all requisite approvals and authorizations for the Contemplated Transactions under any Antitrust Law, use its reasonable best efforts to: (i) cooperate in all respects with each other Party or its Affiliates in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private Person; (ii) keep the other Parties reasonably informed of any communication received by such Party or its Representatives from, or given by such Party or its Representatives to, any Governmental Authority and of any communication received or given in connection with any proceeding by a private Person, in each case regarding any of the Contemplated Transactions; (iii) permit a Representative of the other Parties and their respective outside counsel to review any communication given by it to, and consult with each other in advance of any meeting or conference with, any Governmental Authority or, in connection with any proceeding by a private Person, with any other Person, and to the extent permitted by such Governmental Authority or other Person, give a Representative or Representatives of the other Parties the opportunity to attend and participate in such meetings and conferences; (iv) in the event a Party's Representative is prohibited from participating in or attending any meetings or conferences, the other Parties shall keep such Party promptly and reasonably apprised with respect thereto; and (v) use reasonable best efforts to cooperate in the filing of any memoranda, white papers, filings, correspondence or other written communications explaining or defending the Contemplated Transactions, articulating any regulatory or competitive argument, and/or responding to requests or objections made by any Governmental Authority.

(d) As soon as reasonably practicable following the date of this Agreement, the Parties shall reasonably cooperate with each other and shall use (and shall cause their respective Affiliates to use) reasonable best efforts to promptly (and in the case of the applications, notices, petitions and filings in respect of the Requisite Regulatory Approvals, within forty-five (45) days of the date of this Agreement) prepare and file with Governmental Authorities requests for approval of the Contemplated Transactions and shall use all reasonable best efforts to have such Governmental Authorities approve the Contemplated Transactions. Each Party shall give prompt written notice to the other Parties if such Party or any of its Representatives receives any notice from such Governmental Authorities in connection with the Contemplated Transactions, and shall promptly furnish the other Parties with a copy of such Governmental Authority notice. If any Governmental Authority requires that a hearing or meeting be held in connection with its approval of the Contemplated Transactions, whether prior to the Closing or after the Closing, each Party shall arrange for Representatives of such Party to be present for such hearing or meeting. If any objections are asserted with respect to the Contemplated Transactions under any applicable Law or if any Action is instituted (or threatened to be instituted) by any applicable Governmental Authority or any private Person challenging any of the Contemplated Transactions as violative of any applicable Law or which would otherwise prevent, materially impede or materially delay the Contemplated Transactions, the Parties shall use their reasonable best efforts to resolve any such objections or Actions so as to timely permit consummation of the Contemplated Transactions, including in order to resolve such objections or Actions which, in any case if not resolved, could reasonably be expected to prevent, materially impede or materially delay the consummation of the Contemplated Transactions. In the event any Action is instituted (or threatened to be instituted) by a Governmental Authority or private Person challenging the Contemplated Transactions, the Parties shall, and shall cause their respective Representatives to, reasonably cooperate with each other and use their respective reasonable best efforts to contest and resist any such Action and to have vacated, lifted, reversed or overturned any Order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the Contemplated Transactions. As used in this Agreement, the term "Requisite Regulatory Approvals" shall mean all regulatory authorizations, consents, permits, waivers, non-objections, orders and approvals (and the expiration or termination of all statutory waiting periods in respect thereof) (i) from the Federal Reserve Board, the FDIC, the Wyoming Division of Banking and any other state banking authorities, and FINRA and (ii) set forth in such Section 3.4 or Section (b) that are necessary to consummate the Contemplated Transactions or those the failure of which to be obtained would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the First Surviving Corporation or Second Surviving Corporation.

(e) Notwithstanding the immediately preceding paragraph, nothing contained herein shall be deemed to require Purchaser or the Company, or any of their respective Subsidiaries, and Purchaser and the Company, and any of their respective Subsidiaries, shall not be permitted (without the written consent of the other party), to take any action, or commit to take any action, or agree to any condition, commitment or restriction, in connection with obtaining the foregoing permits, consents, orders, approvals, waivers, non-objections and authorizations of Governmental Authorities that would reasonably be expected to be materially financially burdensome to the business, operations, financial condition or results of operations on the business of the Company and its Subsidiaries (taken as a whole), or on the business of Purchaser and its Subsidiaries (taken as a whole) (which restriction, commitment, or condition could include materially increasing capital, divesting or reducing lines of businesses or asset classes, entering into compliance or remediation programs, and making material lending or investment commitments) (a "*Materially Burdensome Regulatory Condition*").

(f) Prior to the Closing, each Party shall use its reasonable best efforts to obtain any Consents of Governmental Authorities or other third Persons as may be necessary for the consummation by such person or its Affiliates of the Contemplated Transactions, the Broker-Dealer Acquisition Agreement and the Bank Acquisition Agreement, or required as a result of the execution or performance of, or consummation of the Contemplated Transactions and as may be necessary for the contemplated conduct of the business of New Pubco following the Closing and set forth on <u>Schedule 6.9</u> hereto, by such Party or its Affiliates, and the other Parties shall provide reasonable cooperation in connection with such efforts.

6.10 Tax Matters. Each of the Parties shall use its reasonable best efforts to cause the Contemplated Transactions to qualify for the Intended Tax Treatment. None of the Parties shall (and each of the Parties shall cause their respective Subsidiaries not to) take any action, or fail to take any action, that could reasonably be expected to cause the Contemplated Transactions to fail to qualify for the Intended Tax Treatment. The Parties intend to report and, except to the extent otherwise required by Law, shall report, for federal income tax purposes, the Contemplated Transactions in accordance with the Intended Tax Treatment.

54

6.11 Further Assurances. The Parties hereto shall further cooperate with each other and use their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable on their part under this Agreement and applicable Laws to consummate the Contemplated Transactions as soon as reasonably practicable, including preparing and filing as soon as practicable all documentation to effect all necessary notices, reports and other filings.

6.12 The Registration Statement/ Proxy Statement.

(a) As promptly as practicable after the delivery by Atlas FinTech and the Company of all financial statements that are required by the applicable accounting requirements and other rules and regulations of the SEC to be included in the Registration Statement/ Proxy Statement (including pro forma financial information), Purchaser shall prepare with the reasonable assistance of the Company, and cause New Pubco to file with the SEC a registration statement on Form S-4 with respect to the New Pubco Common Stock and New PubCo Warrants to be offered and issued in connection with the First Merger (the "Registration Statement"), which shall include a proxy statement for the purpose of soliciting proxies from Purchaser stockholders for the matters to be acted upon at Purchaser Special Meeting (as amended, the "Proxy Statement", and together with the Registration Statement, the "Registration Statement/ Proxy Statement") and providing the Public Stockholders an opportunity in accordance with the Purchaser's Organizational Documents and the IPO Prospectus to have their Purchaser Common Stock redeemed (the "Redemption") in conjunction with the stockholder vote on the approval of this Agreement. The Proxy Statement shall include proxy materials for the purpose of soliciting proxies from Purchaser stockholders to vote, at a special meeting of Purchaser stockholders to be called and held for such purpose (the "Purchaser Special Meeting"), in favor of resolutions approving (i) the adoption and approval of this Agreement and the Contemplated Transactions, including the Mergers, by the holders of Purchaser Common Stock in accordance with the Purchaser's Organizational Documents and IPO Prospectus, the Securities Act, the DGCL and the rules and regulations of the SEC and NYSE, (ii) the adoption and approval of the New Pubco Charter and New Pubco Bylaws, including the change of name of New Pubco, (iii) adoption and approval of a new equity incentive plan by New Pubco substantially in a form to be agreed to by the Parties (the "Incentive Plan"), and which will provide for awards for a number of shares of New Pubco Common Stock equal to ten percent (10%) of the aggregate number of shares of New Pubco Common Stock issued and outstanding immediately after the Closing (giving effect to the Redemption), (iv) the appointment of the members of the New Pubco Board in accordance with Section 6.16 hereof, (v) such other matters as the Company and Purchaser shall hereafter mutually determine to be necessary or appropriate in order to effect the Mergers and the Contemplated Transactions (the approvals described in foregoing clauses (i) through (v), collectively, the "Purchaser Stockholder Approval Matters"), and (vi) the adjournment of Purchaser Special Meeting, if necessary or desirable in the reasonable determination of Purchaser. If on the date for which Purchaser Special Meeting is scheduled, Purchaser has not received proxies representing a sufficient number of shares to obtain the Required Purchaser Stockholder Approval, whether or not a quorum is present, Purchaser may make one or more successive postponements or adjournments of Purchaser Special

Meeting. In connection with the Registration Statement/ Proxy Statement, New Pubco and Purchaser will file with the SEC financial and other information about the Contemplated Transactions in accordance with applicable Law and applicable proxy solicitation and registration statement rules set forth in the Purchaser's Organizational Documents, the Securities Act, the DGCL and the rules and regulations of the SEC and NYSE. Purchaser shall cooperate and provide the Company (and its counsel) with a reasonable opportunity to review and comment on the Registration Statement/ Proxy Statement and any amendment or supplement thereto prior to filing the same with the SEC, and Purchaser shall consider any such comments timely made in good faith. The Company shall provide Purchaser with such information concerning the Target Companies and Contributed Assets and, to the extent available to the Company, the Target Companies and their stockholders, officers, directors, employees, assets, Liabilities, condition (financial or otherwise), business and operations that may be required or appropriate for inclusion in the Registration Statement/ Proxy Statement, or in any amendments or supplements thereto.

55

(b) Purchaser shall take any and all commercially reasonable and necessary actions required to satisfy the requirements of the Securities Act, the Exchange Act and other applicable Laws in connection with the Registration Statement/ Proxy Statement, Purchaser Special Meeting and the Redemption. Each of Purchaser and the Company shall, and shall cause each of its Subsidiaries to, make their respective directors, officers and employees, upon reasonable advance notice, available to the Company, Purchaser and their respective Representatives in connection with the drafting of the public filings with respect to the Contemplated Transactions, including the Registration Statement/ Proxy Statement, and responding in a timely manner to comments from the SEC. Each Party shall promptly correct any information provided by it for use in the Registration Statement/ Proxy Statement (and other related materials) if and to the extent that such information is determined to have become false or misleading in any material respect or as otherwise required by applicable Laws. New Pubco and Purchaser shall amend or supplemented, to be filed with the SEC and to be disseminated to Purchaser stockholders, in each case as and to the extent required by applicable Laws and subject to the terms and conditions of this Agreement and the Purchaser's Organizational Documents; provided, however, that Purchaser shall not amend or supplement the Registration Statement/ Proxy Statement without prior written consent of the Company, not to be unreasonably withheld, conditioned, or delayed.

(c) New Pubco and Purchaser, with the assistance of the other Parties, shall promptly respond to any SEC comments on the Registration Statement/ Proxy Statement and shall otherwise use its reasonable best efforts to cause the Proxy Statement to "clear" comments from the SEC and the Registration Statement to be declared effective. New Pubco and Purchaser shall provide the Company with copies of any written comments, and shall inform the Company of any material oral comments, that New Pubco, Purchaser or its Representatives receive from the SEC or its staff with respect to the Registration Statement/ Proxy Statement, Purchaser Special Meeting and the Redemption promptly after the receipt of such comments and shall give the Company and its counsel a reasonable opportunity under the circumstances to review and comment on any proposed written or material oral responses to such comments, and Purchaser shall consider any such comments timely made in good faith under the circumstances.

(d) As soon as practicable following the Proxy Statement "clearing" comments from the SEC and the Registration Statement being declared effective and the expiration of the waiting period in Rule 14a-6(a) under the Exchange Act, New Pubco and Purchaser shall distribute the Registration Statement/ Proxy Statement to Purchaser's stockholders, and, pursuant thereto, Purchaser shall call the Purchaser Special Meeting in accordance with the DGCL and Purchaser's Organizational Documents for a date no later than thirty (30) days following the mailing of the Registration Statement/ Proxy Statement, or the earliest date thereafter as may be required by law or regulation.

(e) Purchaser shall comply with all applicable Laws, any applicable rules and regulations of NYSE, the Purchaser's Organizational Documents and this Agreement in the preparation, filing and distribution of the Proxy Statement, any solicitation of proxies thereunder, the calling and holding of the Purchaser Special Meeting and the Redemption. Purchaser and New Pubco shall apply for, and shall take commercially reasonable actions to cause, the New Pubco Common Stock to be issued in connection with the First Merger to be approved for listing on NYSE as of the Closing.

(f) New Pubco may, at its election, include in the Registration Statement, the shares of New Pubco Common Stock to offered and issued in connection with the transactions contemplated by the Bank Acquisition Agreement.

6.13 Company Stockholders Written Consent. Concurrently with the execution and delivery of this Agreement, the Company shall deliver to Purchaser the Company Stockholders Written Consent, which such Company Stockholders Written Consent shall represent the Required Company Stockholder Approval in favor of the approval and adoption of this Agreement, the Second Merger, and the Contemplated Transactions.

6.14 Public Announcements.

(a) The Parties agree that during the Interim Period no public release, filing or announcement concerning this Agreement or the Ancillary Documents or the Contemplated Transactions shall be issued by any Party or any of their Affiliates without the prior written consent of Purchaser and the Company (which consent shall not be unreasonably withheld, conditioned or delayed), except as such release or announcement may be required by applicable Law or the rules or regulations of any securities exchange, in which case the applicable Party shall use reasonable best efforts to allow the other Parties reasonable time to comment on, and arrange for any required filing with respect to, such release or announcement in advance of such issuance. The Company shall use reasonable best efforts to cause the Acquisition Targets to comply with this <u>Section 6.14</u>.

(b) The Parties shall mutually agree upon and, as promptly as practicable after the execution of this Agreement (but in any event within four (4) Business Days thereafter), issue a press release announcing the execution of this Agreement (the "Signing Press Release"). Promptly after the issuance of the Signing Press Release, Purchaser shall file a current report on Form 8-K (the "Signing Filing") with the Signing Press Release and a description of this Agreement as required by Federal Securities Laws, which the Company shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing (with the Company reviewing, commenting upon and approving such Signing Filing in any event no later than the third (3rd) Business Day after the execution of this Agreement); provided that Purchaser provides the Company with a reasonable amount of time to complete such review, comment and approval prior to the third (3rd) Business Day after the date thereof. The Parties shall mutually agree upon and, as promptly as practicable after the Closing (but in any event within four (4) Business Days thereafter), issue a press release announcing the consummation of the Contemplated Transactions (the "Closing Press Release"). Promptly after the issuance of the Closing Press Release, Purchaser shall file a current report on Form 8-K (the "Closing Filing") with the Closing Press Release and a description of the Closing as required by Federal Securities Laws which the Company shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing. In connection with the preparation of the Signing Press Release, the Signing Filing, the Closing Filing, the Closing Press Release, or any other report, statement, filing notice or application made by or on behalf of a Party to any Governmental Authority or other third party in connection with the Contemplated Transactions, each Party shall, upon request by any other Party, furnish the Parties with all information concerning themselves, their respective directors, officers and equity holders, and such other matters as may be reasonably necessary or advisable in connection with the Contemplated Transactions, or any other report, statement, filing, notice or application made by or on behalf of a Party to any third party and/ or any Governmental Authority in connection with the Contemplated Transactions.

57

6.15 Confidential Information. All Purchaser Confidential Information obtained by the Company and its Affiliates and Company Confidential Information obtained by Purchaser and its Affiliates shall be kept confidential in accordance with the Confidentiality Agreement, dated August 14, 2022, between Purchaser and the Company (the "Confidentiality Agreement"). The Confidentiality Agreement shall survive the execution and delivery of this Agreement and shall apply to all information furnished thereunder or hereunder and any other activities contemplated thereby. The Company acknowledges that, in connection with the Potential Financing, Purchaser shall be entitled to disclose and file with the SEC any information contained in any presentation made to the potential equity investors, which information may include Confidential Information (as defined in the Confidentiality Agreement); provided that, Company shall have the right to review and approve (which approval shall not be unreasonably conditioned, withheld or delayed) any materials prior to their distribution.

6.16 Post-Closing Board of Directors and Officers; Employment Agreements.

(a) The Parties shall mutually agree upon and, as promptly as practicable after the execution of this Agreement (but in any event prior to the initial filing of the Proxy Statement), (i) the officers of New Pubco and the Second Surviving Corporation as of the Second Effective Time, and (ii) the members of the board of directors of New Pubco and the Second Surviving Corporation as of the Second Effective Time (the "*New Pubco Board*"); *provided, however*, that in making such designation at least a majority of the directors must qualify as independent under NYSE requirements.

(b) The Parties shall take all action necessary, including causing all of the officers of the Company to resign, so that effective as of the Second Effective Time, the New Pubco's officers and the New Pubco Board will consist of the individuals contemplated by <u>Section 6.16(a)</u>. At or prior to the Closing, New Pubco will provide each member of the New Pubco Board with a customary director indemnification agreement.

(c) The Parties shall negotiate in good faith to finalize employment agreements with the individuals identified on Schedule 8.3(d)(v).

6.17 Indemnification of Directors and Officers; Tail Insurance.

(a) The Parties agree that all rights to exculpation, indemnification and advancement of expenses existing in favor of the current or former directors and officers of Purchaser Parties and each Person who served as a director, officer, member, trustee or fiduciary of another corporation, partnership, joint venture, trust, pension or other employee benefit plan or enterprise at the request of Purchaser Parties (the "*D&O Indemnified Persons*") as provided in their respective Organizational Documents or under any indemnification, employment or other similar agreements between any D&O Indemnified Person and Purchaser Parties, in each case as in effect on the date of this Agreement, shall survive the Closing and continue in full force and effect in accordance with their respective terms to the extent permitted by applicable Law and, as of the Closing, New Pubco shall be jointly and severally liable for all such obligations. For a period of six (6) years after the First Effective Time, New Pubco shall cause the Organizational Documents of New Pubco, the First Surviving Corporation and the Second Surviving Corporation to contain provisions no less favorable with respect to exculpation and indemnification of and advancement of expenses to D&O Indemnified Persons than are set forth as of the date of this Agreement in the Organizational Documents of the Purchaser Parties to the extent permitted by applicable Law. The provisions of this <u>Section 6.17</u> shall survive the consummation of the Mergers and are intended to be for the benefit of, and shall be enforceable by, each of the D&O Indemnified Persons and their respective heirs and representatives.

58

(b) For the benefit of Purchaser Parties' directors and officers, Purchaser and New Pubco shall be permitted prior to the First Effective Time to obtain and fully pay the premium for a "tail" insurance policy that provides coverage for up to a six-year period from and after the First Effective Time for events occurring prior to the First Effective Time (the "**D&O Tail Insurance**") that is substantially equivalent to and in any event not less favorable in the aggregate than Purchaser's existing policy or, if substantially equivalent insurance coverage is unavailable, the best available coverage. If obtained, New Pubco shall, or shall cause the First Surviving Corporation and the Second Surviving Corporation to, maintain the D&O Tail Insurance in full force and effect, and continue to honor the obligations thereunder, and New Pubco shall, or shall cause the First Surviving Corporation to, timely pay or caused to be paid all premiums with respect to the D&O Tail Insurance.

6.18 Trust Account Proceeds. Upon satisfaction or waiver of the conditions set forth in <u>Article VIII</u> and provision of notice thereof to the Trustee (which notice Purchaser shall provide to the Trustee in accordance with the terms of the Trust Agreement), (a) in accordance with and pursuant to the Trust Agreement, at the Closing, Purchaser (i) shall cause the documents, opinions and notices required to be delivered to the Trustee pursuant to the Trust Agreement to be so delivered, including providing the Trustee with the Trust Termination Letter; and (ii) shall cause the Trustee to, and the Trustee shall thereupon be obligated to, distribute the Trust Account as directed in the Trust Termination Letter, including all amounts payable (A) to Purchaser stockholders who have elected to have their Purchaser Common Stock redeemed in connection with the Redemption, (B) for income tax or other tax obligations of Purchaser prior to Closing and (b) thereafter, the Trust Account shall terminate, except as otherwise provide therein, and the proceeds therefrom shall be used to pay the Purchaser Transaction Expenses and the Company Transaction Expenses.

6.19 Financing. During the Interim Period, New Pubco or Purchaser may execute subscription agreements or enter into other arrangements (including, without limitation, non-redemption agreements, backstop agreements for the trust account, etc.) with potential sources of debt, equity or other financing in order to satisfy the condition set forth in <u>Section 8.2(d)</u>. If Purchaser desires to seek such

financing (the "Potential Financing") and the Company consents in writing to such financing (such consent not to be unreasonably conditioned, withheld or delayed), the Company and each Company Stockholder agrees, and shall cause the appropriate officers and employees thereof, to use commercially reasonable efforts to cooperate in connection with the arrangement of such Potential Financing (including the satisfaction of the conditions precedent set forth therein) as may be reasonably requested by Purchaser, including by (a) participating in a reasonable number of meetings, presentations, due diligence sessions, drafting sessions and sessions with rating agencies at mutually agreeable times and locations and upon reasonable advance notice, (b) assisting with the preparation of customary materials for actual and potential investors, rating agency presentations, offering documents, private placement memoranda, bank information memoranda, prospectuses and similar documents required in connection with such financing (which shall not include pro forma financial information); provided, that, the Company shall have the right to review and approve (which approval shall not be unreasonably conditioned, withheld or delayed) any such materials prior to their distribution, (c) executing and delivering any pledge and security documents, other definitive financing documents, or other certificates, or documents as may be reasonably requested by Purchaser or otherwise reasonably facilitating the pledging of collateral, provided, that, such documents will not take effect until the Closing, (d) taking or appointing a representative of Purchaser to take all corporate actions, subject to the occurrence of the Closing, reasonably requested by Purchaser to permit the consummation of the Potential Financing immediately prior to or following the Closing Date, (e) providing the Company Financials and such other financial information regarding the Company that is readily available or within the Company's possession and as is reasonably requested in connection with arrangement of such financing, (f) executing and delivering reasonable and customary certificates and other documentation required by any such equity investor and the subscription agreement, provided, that no such certificates, letters or other documentation shall be effective prior to the consummation of the transactions contemplated by the subscription agreement, (g) furnishing Purchaser, with all documentation and other information required by bank regulatory authorities under applicable "know-your-customer" and anti-money laundering rules and regulations, including the Patriot Act and (h) otherwise reasonably cooperating in Purchaser's efforts to obtain such Potential Financing. If Purchaser desires to seek Potential Financing, Atlas FinTech further agrees, and shall cause the appropriate officers and employees thereof, to cause the applicable Sponsors to transfer up to 1,279,426.82 Purchaser Common Stock and up to 1,657,578.65 Purchaser Warrants that are held, directly or indirectly, by Atlas Fintech to potential sources of debt, equity or other financing in connection with the Potential Financing, which transfer shall be effectuated in connection with, and as of immediately prior to, the Closing. Atlas Fintech agrees that any Purchaser Common Stock or Purchaser Warrant that remains held, directly or indirectly, by Atlas Fintech as of immediately prior to the Closing shall automatically be forfeited for no consideration and all such shares of Purchaser Common Stock or Purchaser Warrants shall be deemed to be cancelled and no longer outstanding.

59

6.20 Pacsquare Acquisition Agreement. The Company and each Company Stockholder will, and will cause his, her or its respective Representatives, as applicable, to, (i) enter into a definitive agreement to reflect the terms and condition of the Pacsquare LOI as promptly as reasonably practicable after the date of this Agreement (the "*Pacsquare Acquisition Agreement*", and (ii) consummate the transactions contemplated by the Pacsquare Acquisition Agreement as promptly as reasonably practicable after the date of doubt, with respect to any amendments to the Pacsquare LOI and the execution of the Pacsquare Acquisition Agreement, the Company shall consult with Purchaser in good faith with respect to such actions, which shall require the prior written consent of Purchaser.

ARTICLE VII. NO SURVIVAL

7.1 No Survival. Representations and warranties of the Company, Purchaser and the Company Stockholders contained in this Agreement or in any certificate or instrument delivered by or on behalf of the Company, Purchaser or a Company Stockholder pursuant to this Agreement shall not survive the Closing, and from and after the Closing, the Company, Purchaser and the Company Stockholders and their respective Representatives shall not have any further obligations, nor shall any claim be asserted or action be brought against the Company, Purchaser, the Company Stockholders or their respective Representatives with respect thereto. The covenants and agreements made by the Company, Purchaser and the Company Stockholders in this Agreement or in any certificate or instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such covenants or agreements, shall not survive the Closing, except for those covenants and agreements contained herein and therein that by their terms apply or are to be performed in whole or in part after the Closing (which such covenants shall survive the Closing and continue until fully performed in accordance with their terms).

ARTICLE VIII. CLOSING CONDITIONS

8.1 Conditions to Each Party's Obligations. The obligations of each Party to consummate the Mergers and the other transactions described herein shall be subject to the satisfaction or written waiver (where permissible) by the Company and Purchaser of the following conditions:

(a) *Required Purchaser Stockholder Approval*. The Purchaser Stockholder Approval Matters that are submitted to the vote of the stockholders of Purchaser at Purchaser Special Meeting in accordance with the Proxy Statement shall have been approved by the requisite vote of the stockholders of Purchaser at Purchaser at Purchaser Special Meeting in accordance with the Purchaser's Organizational Documents, applicable Law and the Proxy Statement (the "*Required Purchaser Stockholder Approval*").

60

(b) *Required Company Stockholder Approval*. The Company Stockholders Written Consent shall have been delivered to Purchaser, and remain in full force and effect, representing the requisite vote of the Company Stockholders, as necessary, to authorize, approve and consent to, the execution, delivery and performance of this Agreement and each of the Ancillary Documents to which the Company is or is required to be a party or bound, and the consummation of the Contemplated Transactions, including the Second Merger (the "*Required Company Stockholder Approval*").

(c) *Antitrust Laws*. Any waiting period (and any extension thereof) applicable to the consummation of this Agreement under any Antitrust Laws shall have expired or been terminated.

(d) *Requisite Regulatory Approvals*. All Consents and approvals required to be obtained from or made with any Governmental Authority in order to consummate the Contemplated Transactions, including the Requisite Regulatory Approvals, shall have been obtained or made and shall remain in full force and effect and all statutory waiting periods in respect thereof shall have expired or been terminated and no such Consents shall have resulted in the imposition of any Materially Burdensome Regulatory Condition.

(e) *No Adverse Law or Order*. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any Law (whether temporary, preliminary or permanent) or Order that is then in effect and which has the effect of making the Contemplated Transactions illegal or which otherwise prevents or prohibits consummation of the Contemplated Transactions.

(f) *Net Tangible Assets Test.* Upon the Closing, after giving effect to the Redemption and any equity investments in Purchaser made in connection with the Closing, Purchaser shall have net tangible assets of at least Five Million One Dollar (\$5,000,001).

(g) Appointment to the New Pubco Board. The members of the New Pubco Board shall have been elected or appointed as of the Closing consistent with the requirements of <u>Section 6.16</u>.

(h) *NYSE/Nasdaq Listing*. The shares of New Pubco Common Stock issued as Merger Consideration shall have been approved for listing on NYSE or, if such listing is not practicable, The Nasdaq Stock Market LLC, subject to official notice of issuance.

(i) Adoption and Approval of the Incentive Plan. The Incentive Plan shall have been adopted and approved consistent with the requirements of Section 6.12(a)(iii).

(j) *Minimum Cash Condition*. Upon the Closing, the Available Closing Cash shall not be less than Forty Million U.S. Dollars (\$40,000,000).

8.2 Conditions to Obligations of the Company. In addition to the conditions specified in <u>Section 8.1</u>, the obligations of the Company to consummate the Second Merger and the Contemplated Transactions are subject to the satisfaction or written waiver (by the Company) of the following conditions:

(a) *Representations and Warranties*. All of the representations and warranties of Purchaser Parties set forth in this Agreement and in any certificate delivered by or on behalf of Purchaser Parties pursuant hereto shall be true and correct on and as of the Closing Date as if made on the Closing Date, except for (i) those representations and warranties that address matters only as of a particular date (which representations and warranties shall have been accurate as of such date), and (ii) any failures

to be true and correct that (without giving effect to any qualifications or limitations as to materiality or Material Adverse Effect), individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on Purchaser Parties.

(b) Agreements and Covenants. The Purchaser Parties shall have performed in all material respects all of Purchaser Parties' obligations and complied in all material respects with all of Purchaser Parties' agreements and covenants under this Agreement to be performed or complied with by it on or prior to the Closing Date.

(c) No Purchaser Material Adverse Effect. No Material Adverse Effect shall have occurred with respect to Purchaser.

(d) Closing Deliveries.

(i) OFFICER CERTIFICATE. The Purchaser shall have delivered to the Company a certificate, dated the Closing Date, signed by an executive officer of Purchaser in such capacity, certifying as to the satisfaction of the conditions specified in Sections 8.2(a), 8.2(b) and 8.2(c).

(ii) SECRETARY CERTIFICATE. The Purchaser shall have delivered to the Company a certificate from its secretary or other executive officer certifying as to, and attaching, (A) copies of the Purchaser's Organizational Documents as in effect as of the Closing Date, (B) the resolutions of Purchaser's board of directors authorizing and approving the execution, delivery and performance of this Agreement and each of the Ancillary Documents to which it is a party or by which it is bound, and the consummation of the Contemplated Transactions, and recommending the approval and adoption of the same by the stockholders of Purchaser, (C) evidence that the Required Purchaser Stockholder Approval has been obtained, (D) the incumbency of officers authorized to execute this Agreement or any Ancillary Document to which Purchaser is or is required to be a party or otherwise bound, and (E) the composition of New Pubco Board being in compliance with Section 6.16(a).

(iii) GOOD STANDING. The Purchaser shall have delivered to the Company a good standing certificate (or similar documents applicable for such jurisdictions) for Purchaser certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of Purchaser's jurisdiction of organization and from each other jurisdiction in which Purchaser is qualified to do business as a foreign entity as of the Closing, in each case to the extent that good standing certificates or similar documents are generally available in such jurisdictions.

(iv) REGISTRATION RIGHTS AND LOCK-UP AGREEMENT. The Company shall have received a copy of the Registration Rights and Lock-Up Agreement, duly executed by New Pubco and Purchaser's stockholders party thereto.

(v) CERTIFICATION AND NOTICE TO THE IRS. Purchaser shall have delivered to New Pubco a properly executed certification that shares of Purchaser Common Stock are not "U.S. real property interests" in accordance with the U.S. Treasury Regulations under Sections 897 and 1445 of the Code, together with a notice to the IRS (which shall be filed by New Pubco with the IRS following the Closing) in accordance with the provisions of Section 1.897-2(h)(2) of the U.S. Treasury Regulations.

62

8.3 Conditions to Obligations of Purchaser Parties. In addition to the conditions specified in <u>Section 8.1</u>, the obligations of Purchaser Parties to consummate the Mergers and the Contemplated Transactions are subject to the satisfaction or written waiver (by Purchaser) of the following conditions:

(a) Representations and Warranties.

(i) All of the representations and warranties of the Company and the Company Stockholders set forth in this Agreement and in any certificate delivered by or on behalf of the Company or the Company Stockholders pursuant hereto shall be true and correct on and as of the Closing Date as if made on the Closing Date, except for (A) those representations and warranties that address matters only as of a particular date (which representations and warranties shall have been accurate as of such date), and (B) any failures to be true and correct that (without giving effect to any qualifications or limitations as to materiality or Material Adverse Effect), individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on the Company.

(ii) The representations and warranties of the Target Broker-Dealer and its stockholders set forth in the Broker-Dealer Acquisition Agreement (as modified by the disclosure schedules delivered by the Target Broker-Dealer to the Company pursuant thereto), and the representations and warranties of the Target Bank set forth in the Bank Acquisition Agreement (as modified by the disclosure schedules delivered by the Target Bank to the Company pursuant thereto), shall be true and correct in all material respects on and as of the Closing Date as if made on the Closing Date.

(b) Agreements and Covenants. Each of the Company and the Company Stockholders shall have performed in all material respects all of its obligations and complied in all material respects with all of its agreements and covenants under this Agreement to be performed or complied by it on or prior to the Closing Date.

(c) No Material Adverse Effect. No Material Adverse Effect shall have occurred with respect to the Company.

(d) Closing Deliveries.

(i) OFFICER CERTIFICATE. The Purchaser shall have received a certificate from the Company and from each of the Company Stockholders, dated as the Closing Date, signed by an executive officer of the Company in such capacity, certifying as to the satisfaction of the conditions specified in Sections 8.3(a), 8.3(b) and 8.3(c).

(ii) SECRETARY CERTIFICATE. The Company shall have delivered to Purchaser a certificate executed by an executive officer of the Company certifying as to the validity and effectiveness of, and attaching, (A) a copy of the Company's Organizational Documents as in effect as of the Closing Date (immediately prior to the Second Effective Time), (B) the requisite resolutions of the board of directors of the Company authorizing and approving the execution, delivery and performance of this Agreement and each Ancillary Document to which the Company is or is required to be a party or bound, and the consummation of the Second Merger and the Contemplated Transactions, and recommending the approval and adoption of the same by the Company Stockholders, and (C) the incumbency of officers of the Company authorized to execute this Agreement or any Ancillary Document to which the Company is or is required to be a party or otherwise bound.

63

(iii) GOOD STANDING. The Company shall have delivered to Purchaser a good standing certificate (or similar documents applicable for such jurisdiction) for the Company certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of the Company's jurisdiction of organization.

(iv) CERTIFIED CHARTER. The Company shall have delivered to Purchaser a copy of the Company Charter, as in effect as of immediately prior to the Second Effective Time, certified by the Secretary of State of the State of Wyoming as of a date no more than ten (10) Business Days prior to the Closing Date.

(v) EMPLOYMENT AGREEMENTS. The Purchaser shall have received employment agreements, in each case effective as of the Closing, in form and substance reasonably acceptable to the Company and Purchaser, between each of the persons set forth in <u>Schedule 8.3(d)(v)</u> hereto and the applicable Target Company or New Pubco, as noted in <u>Schedule 8.3(d)(v)</u>, each such employment agreement duly executed by the parties thereto.

(vi) REGISTRATION RIGHTS AND LOCK-UP AGREEMENT. The Company shall have delivered to Purchaser copies of the Registration Rights and Lock-Up Agreement duly executed by all of the Company Stockholders.

(vii) RESIGNATIONS. Notwithstanding the provision set forth in Section 6.17, Purchaser shall have received written resignations, effective as of the Closing, of each of the directors and officers of the Company and each other Target Company listed on Schedule 8.3(d)(vii), or as otherwise mutually agreed by Purchaser and the Company prior to the Closing.

(viii) CERTIFICATION AND NOTICE TO THE IRS. The Company shall have delivered to New Pubco a properly executed certification that shares of Company Common Stock are not "U.S. real property interests" in accordance with the U.S. Treasury Regulations under Sections 897 and 1445 of the Code, together with a notice to the IRS (which shall be filed by New Pubco with the IRS following the Closing) in accordance with the provisions of Section 1.897-2(h)(2) of the U.S. Treasury Regulations.

(e) *Pre-Closing Transactions*. The transactions contemplated by the Broker-Dealer Acquisition Agreement, and the Contribution Agreement the Pacsquare Acquisition Agreement (unless such agreement is not entered into at Purchaser's request pursuant to <u>Section 6.20</u>) shall have been duly consummated, or shall be consummated substantially simultaneously with the Closing, in each case, in accordance with the terms thereof. The terms and conditions of the Bank Acquisition Agreement shall be in full force and effect as of the Closing Date.

8.4 Frustration of Conditions. Notwithstanding anything contained herein to the contrary, no Party may rely on the failure of any condition set forth in this <u>Article VIII</u> to be satisfied if such failure was caused by the failure of such Party or its Affiliates failure to comply with or perform any of its covenants or obligations set forth in this Agreement.

64

ARTICLE IX. TERMINATION AND EXPENSES

9.1 Termination. This Agreement may be terminated and the Contemplated Transactions may be abandoned at any time prior to the Closing as follows:

(a) by mutual written consent of Purchaser and the Company;

(b) by written notice by Purchaser or the Company if any of the conditions to the Closing set forth in <u>Article VIII</u> have not been satisfied or waived by February 9, 2023 (provided, that if Purchaser seeks and obtains an Extension, Purchaser shall have the right by providing written notice thereof to the Company to extend the Outside Date for up to the shorter of (i) up to two additional periods equal to three (3) additional months each (for a total of not more than six (6) months in aggregate), (ii) the period ending on the last date for Purchaser to consummate its Business Combination pursuant to such Extension and (iii) such period as determined by Purchaser) (the "*Outside Date*"); *provided, however*, the right to terminate this Agreement under this <u>Section 9.1(b)</u> shall not be available to a Party if the breach or violation by such Party or its Affiliates of any representation, warranty, covenant or obligation under this Agreement was the cause of, or resulted in, the failure of the Closing to occur on or before the Outside Date;

(c) by written notice by either Purchaser or the Company if a Governmental Authority of competent jurisdiction shall have issued an Order or taken any other action permanently restraining, enjoining or otherwise prohibiting the Contemplated Transactions, and such Order or other action has become final and non-appealable; *provided, however*, that the right to terminate this Agreement pursuant to this <u>Section 9.1(c)</u> shall not be available to a Party if the failure by such Party or its Affiliates to comply with any provision of this Agreement has been a substantial cause of, or substantially resulted in, such action by such Governmental Authority;

(d) by written notice by the Company to Purchaser, if (i) there has been a material breach by Purchaser of any of its representations, warranties, covenants or agreements contained in this Agreement, or if any representation or warranty of Purchaser shall have become untrue or inaccurate, in any case, which would result in a failure of a condition set forth in <u>Section 8.2(a)</u> or <u>Section 8.2(b)</u> to be satisfied (treating the Closing Date for such purposes as the date of this Agreement or, if later, the date of such breach), and (ii) the breach or inaccuracy is incapable of being cured or is not cured within the earlier of (A) twenty (20) days after written notice of such breach or inaccuracy is provided to Purchaser or (B) the Outside Date;

provided, that the Company shall not have the right to terminate this Agreement pursuant to this <u>Section 9.1(d)</u> if at such time the Company is in material uncured breach of this Agreement;

(e) by written notice by Purchaser to the Company, if (i) there has been a material breach by the Company or a Company Stockholder of any of his or its representations, warranties, covenants or agreements contained in this Agreement, or if any representation or warranty of such persons shall have become untrue or inaccurate, in any case, which would result in a failure of a condition set forth in Section 8.3(v)(a) or Section 8.3(v)(b) to be satisfied (treating the Closing Date for such purposes as the date of this Agreement or, if later, the date of such breach), and (ii) the breach or inaccuracy is incapable of being cured or is not cured within the earlier of (A) twenty (20) days after written notice of such breach or inaccuracy is provided to the Company or (B) the Outside Date; *provided, that* Purchaser shall not have the right to terminate this Agreement pursuant to this Section 9.1(e) if at such time Purchaser is in material uncured breach of this Agreement;

(f) by written notice by Purchaser to the Company, if there shall have been a Material Adverse Effect on the Company following the date of this Agreement which is uncured upon the earlier to occur of (i) the Outside Date, or (ii) for at least ten (10) business days after written notice of such Material Adverse Effect is provided by Purchaser to the Company;

(g) by written notice by Purchaser to the Company, if any of the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, the Pacsquare Acquisition Agreement or the Contribution Agreement is terminated or if any of the conditions set forth in the Broker-Dealer Acquisition Agreement or the Bank Acquisition Agreement are reasonably likely to be unsatisfied;

65

(h) by written notice by either Purchaser or the Company to the other, if Purchaser Special Meeting is held (not including any adjournment or postponement thereof in accordance with the terms of this Agreement) and has concluded, Purchaser's stockholders have duly voted, and the Required Purchaser Stockholder Approval was not obtained; or

(i) by written notice by Purchaser to the Company, if the Company shall fail to deliver the Required Company Financial Statements to Purchaser by December 15, 2022.

9.2 Effect of Termination. This Agreement may only be terminated in the circumstances described in <u>Section 9.1</u> and pursuant to a written notice delivered by the applicable Party to the other applicable Parties, which sets forth the basis for such termination, including the provision of <u>Section 9.1</u> under which such termination is made. In the event of the valid termination of this Agreement pursuant to <u>Section 9.1</u>, this Agreement shall forthwith become void, and there shall be no Liability on the part of any Party or any of their respective Representatives, and all rights and obligations of each Party shall cease, except: <u>Section 9.3</u> (Fees and Expenses), <u>Section 10.1</u> (Waiver of Claims Against Trust), <u>Article XI</u> (Miscellaneous) and this <u>Section 9.2</u> (Effect of Termination) shall survive the termination of this Agreement. Without limiting the foregoing, and except as provided in <u>Sections 9.3</u> and this <u>Section 9.2</u> (but subject to <u>Section 10.1</u>) and subject to the right to seek injunctions, specific performance or other equitable relief in accordance with <u>Section 11.6</u>, the Parties' sole right prior to the Closing with respect to any breach of any representation, warranty, covenant or other agreement contained in this Agreement by another Party or with respect to the Contemplated Transactions shall be the right, if applicable, to terminate this Agreement pursuant to <u>Section 9.1</u>.

9.3 Fees and Expenses. All expenses incurred in connection with this Agreement and the Contemplated Transactions shall be paid by the Party incurring such expenses, except (a) the Company shall be responsible for all required filing or similar fees with respect to any required approval from a Governmental Authority, and (b) if the Contemplated Transactions are consummated, Purchaser shall pay the Purchaser Transaction Expenses and the Company Transaction Expenses pursuant to <u>Section 6.18</u>.

ARTICLE X. WAIVERS AND RELEASES

10.1 Waiver of Claims Against Trust. Reference is made to the IPO Prospectus. The Company hereby represents and warrants that it understands that Purchaser has established the Trust Account containing the proceeds of the IPO and the overallotment shares acquired by Purchaser's underwriters and from certain private placements occurring simultaneously with the IPO (including interest accrued from time to time thereon) for the benefit of Purchaser's public stockholders (including overallotment shares acquired by Purchaser's underwriters) (the "*Public Stockholders*") and that, except as otherwise described in the IPO Prospectus, Purchaser may

disburse monies from the Trust Account only: (a) to the Public Stockholders in the event they elect to redeem their Purchaser Common Stock in connection with the consummation of its initial business combination (as such term is used in the IPO Prospectus) ("Business Combination") or in connection with an amendment to the Purchaser's Organizational Documents to extend Purchaser's deadline to consummate a Business Combination, (b) to the Public Stockholders if Purchaser fails to consummate a Business Combination within eighteen (18) months (or twenty four (24) months, as applicable) after the closing of the IPO, (c) with respect to any interest earned on the amounts held in the Trust Account, amounts necessary to pay for any taxes, and (d) to Purchaser after or concurrently with the consummation of a Business Combination. For and in consideration of Purchaser entering into this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Company hereby agrees on behalf of itself and its Affiliates that, notwithstanding anything to the contrary in this Agreement, neither the Company nor any of its Affiliates do now or shall at any time hereafter have any right, title, interest or claim of any kind in or to any monies in the Trust Account or distributions therefrom (other than any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination), or make any claim against the Trust Account (including any distributions therefrom, other than any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination), regardless of whether such claim arises as a result of, in connection with or relating in any way to, this Agreement or any proposed or actual business relationship between Purchaser or any of its Representatives, on the one hand, and the Company, or any of its Representatives, on the other hand, or any other matter, and regardless of whether such claim arises based on contract, tort, equity or any other theory of legal liability (collectively, the "Released Claims"). The Company on behalf of itself and its Affiliates hereby irrevocably waives any Released Claims that any such Party or any of its Affiliates may have against the Trust Account (including any distributions therefrom, other than any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination) now or in the future as a result of, or arising out of, any negotiations, contracts or agreements with Purchaser or its Representatives and will not seek recourse against the Trust Account (including any distributions therefrom, other than any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination) for any reason whatsoever (including for an alleged breach of this Agreement or any other agreement with Purchaser or its Affiliates); provided, however, that the foregoing waiver will not limit or prohibit the Company or its Affiliates from pursuing a claim against a Purchaser Party or any other Person for legal relief against monies or other assets of Purchaser Parties held outside of the Trust Account (including any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination) or for specific performance or other equitable relief in connection with the Contemplated Transactions. The Company agrees and acknowledges that such irrevocable waiver is material to this Agreement and specifically relied upon by Purchaser and its Affiliates to induce Purchaser to enter in this Agreement, and the Company further intends and understands such waiver to be valid, binding and enforceable against such Party and each of its Affiliates under applicable Law. To the extent that the Company or any of its Affiliates commences any Action based upon, in connection with, relating to or arising out of any matter relating to Purchaser or its Representatives, which proceeding seeks, in whole or in part, monetary relief against Purchaser or its Representatives, the Company hereby acknowledges and agrees that its and its Affiliates' sole remedy shall be against funds held outside of the Trust Account (including any amounts distributed from the Trust Account to Purchaser (or any successor entity) in connection with the consummation of a Business Combination) and that such claim shall not permit such Party or any of its Affiliates (or any Person claiming on any of their behalf or in lieu of them) to have any claim against the Trust Account (including any distributions therefrom) or any amounts contained therein. This Section 10.1 shall survive termination of this Agreement for any reason and continue indefinitely.

<u>6</u>6

ARTICLE XI. MISCELLANEOUS

11.1 Notices. All notices, consents, waivers and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered (i) in person, (ii) by facsimile or other electronic means (including e-mail), with affirmative confirmation of receipt, (iii) one Business Day after being sent, if sent by reputable, nationally recognized overnight courier service or (iv) three (3) Business Days after being mailed, if sent by registered or certified mail, pre-paid and return receipt requested, in each case to the applicable Party at the following addresses (or at such other address for a Party as shall be specified by like notice):

If to Purchaser Parties at or prior to the Closing, to:

Quantum FinTech Acquisition Corporation 4221 W. Boy Scout Blvd., Suite 300 Tampa, FL 33607 with a copy (which will not constitute notice) to:

Winston & Strawn LLP 200 Park Avenue New York, NY 10166

Attn: Carol Anne Huff; Jason D. Osborn Facsimile: 212-294-4700 Email: CHuff@winston.com; JOsborn@winston.com

If to the Company or the Second Surviving Corporation, to:

AtlasClear, Inc. 4221 W. Boy Scout Blvd., Ste 300 Tampa, FL 33607 Attn: Craig Ridenhour Telephone No.: +1 727.512.9631 E-mail: cridenhour@atlasbanc.com with a copy (which will not constitute notice) to:

Greenberg Traurig, LLP 1750 Tysons Boulevard, Suite 1000 McLean, VA 22102 Attn: Jason Simon Facsimile No.: +1 703.714.8386 Telephone No.: +1 703.749.1386 E-mail: simonj@gtlaw.com

If to First Surviving Corporation, to:

AtlasClear Holdings, Inc. 4221 W. Boy Scout Blvd., Ste 300 Tampa, FL 33607 Attn: Craig Ridenhour Telephone No.: +1-727-512-9631 E-mail: cridenhour@atlasbanc.com with a copy (which will not constitute notice) to:

Winston & Strawn LLP 200 Park Avenue New York, NY 10166 Attn: Carol Anne Huff; Jason D. Osborn Facsimile: 212-294-4700 Email: CHuff@winston.com; JOsborn@winston.com

11.2 Binding Effect; Assignment. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns. This Agreement shall not be assigned by operation of Law or otherwise without the prior written consent of Purchaser and the Company, and any assignment without such consent shall be null and void; *provided* that no such assignment shall relieve the assigning Party of its obligations hereunder.

11.3 Third Parties. Except for the rights of the D&O Indemnified Persons set forth in Section 6.17, which the Parties acknowledge and agree are express third (3rd) party beneficiaries of this Agreement, nothing contained in this Agreement or in any instrument or document executed by any Party in connection with the Contemplated Transactions shall create any rights in, or be deemed to have been executed for the benefit of, any Person that is not a Party hereto or thereto or a successor or permitted assign of such a Party.

11.4 Governing Law; Jurisdiction. This Agreement shall be governed by, construed and enforced in accordance with the Laws of the State of Delaware without regard to the conflict of laws principles thereof; provided, however, that the Second Merger shall be governed by, construed and enforced in accordance with the Laws of the State of Wyoming (including the WBCA). All Actions arising out of or relating to this Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware located in Wilmington, Delaware (and in the absence of jurisdiction, the Parties consent to be subject to the jurisdiction of the United States District Court for the District of Delaware or any other state court located in Wilmington, Delaware) (the "Specified Courts"). Each Party hereto hereby (a) submits to the exclusive jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Agreement brought by any Party hereto and (b) irrevocably waives, and agrees not to assert by way of motion, defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the Contemplated Transactions may not be enforced in or by any Specified Court. Each Party agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each Party irrevocably consents to the service of the summons and complaint and any other process in any other Action relating to the Contemplated Transactions, on behalf of itself, or its property, by personal delivery of copies of such process to such Party at the applicable address set forth in Section 11.1. Nothing in this Section 11.4 shall affect the right of any Party to serve legal process in any other manner permitted by Law.

11.5 WAIVER OF JURY TRIAL. EACH PARTY HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE CONTEMPLATED TRANSACTIONS. EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF ANY ACTION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 11.5</u>.

11.6 Specific Performance. Each Party acknowledges that the rights of each Party to consummate the Contemplated Transactions are unique, recognizes and affirms that in the event of a breach of this Agreement by any Party, money damages may be inadequate and the non-breaching Parties may have not adequate remedy at law, and agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed by an applicable Party in accordance with their specific terms or were otherwise breached. Accordingly, each Party shall be entitled to seek an injunction or restraining order to prevent breaches of this Agreement and to seek to enforce specifically the terms and provisions hereof, without the requirement to post any bond or other security or to prove that money damages would be inadequate, this being in addition to any other right or remedy to which such Party may be entitled under this Agreement, at law or in equity.

11.7 Severability. In case any provision in this Agreement shall be held invalid, illegal or unenforceable in a jurisdiction, such provision shall be modified or deleted, as to the jurisdiction involved, only to the extent necessary to render the same valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby nor shall the validity, legality or enforceability of such provision be affected thereby in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties will substitute for any invalid, illegal or unenforceable provision a suitable and equitable provision that carries out, so far as may be valid, legal and enforceable, the intent and purpose of such invalid, illegal or unenforceable provision.

11.8 Amendment. This Agreement may be amended, supplemented or modified only by execution of a written instrument signed by Purchaser and the Company.

68

11.9 Waiver. The Purchaser on behalf of itself and its Affiliates, and the Company on behalf of itself and its Affiliates, may in its sole discretion (i) extend the time for the performance of any obligation or other act of any other non-Affiliates hereto, (ii) waive any inaccuracy in the representations and warranties by such other non-Affiliates Party contained herein or in any document delivered pursuant hereto and (iii) waive compliance by such other non-Affiliates with any covenant or condition contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the Party or Parties to be bound thereby. Notwithstanding the foregoing, no failure or delay by a Party in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any other right hereunder.

11.10 Entire Agreement. This Agreement and the documents or instruments referred to herein, including any exhibits and schedules attached hereto, which exhibits and schedules are incorporated herein by reference, together with the Ancillary Documents, embody the entire agreement and understanding of the Parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred to herein or the documents or instruments referred to herein, which collectively supersede all prior agreements and the understandings among the Parties with respect to the subject matter contained herein.

11.11 Interpretation. The table of contents and the Article and Section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the Parties and shall not in any way affect the meaning or interpretation of this Agreement. In this Agreement, unless the context otherwise requires: (a) any pronoun used shall include the corresponding masculine, feminine or neuter forms, and words in the singular, including any defined terms, include the plural and vice versa; (b) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity; (c) any accounting term used

and not otherwise defined in this Agreement or any Ancillary Document has the meaning assigned to such term in accordance with GAAP; (d) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding or succeeding such term and shall be deemed in each case to be followed by the words "without limitation"; (e) the words "herein," "hereto," and "hereby" and other words of similar import shall be deemed in each case to refer to this Agreement as a whole and not to any particular Section or other subdivision of this Agreement; (f) the word "if" and other words of similar import when used herein shall be deemed in each case to be followed by the phrase "and only if"; (g) the term "or" means "and/or"; (h) any reference to the term "ordinary course" or "ordinary course of business" shall be deemed in each case to be followed by the words "consistent with past practice"; (i) any agreement, instrument, insurance policy, Law or Order defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument, insurance policy, Law or Order as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes, regulations, rules or orders) by succession of comparable successor statutes, regulations, rules or orders and references to all attachments thereto and instruments incorporated therein; (j) except as otherwise indicated, all references in this Agreement to the words "Section," "Article," "Schedule" and "Exhibit" are intended to refer to Sections, Articles, Schedules and Exhibits to this Agreement; and (k) the term "Dollars" or "\$" means United States dollars. Any reference in this Agreement to a Person's directors shall include any member of such Person's governing body and any reference in this Agreement to a Person's officers shall include any Person filling a substantially similar position for such Person. Any reference in this Agreement or any Ancillary Document to a Person's shareholders or stockholders shall include any applicable owners of the equity interests of such Person, in whatever form, including with respect to Purchaser its stockholders under the DGCL, as then applicable, or its Organizational Documents. The Parties have participated jointly in the negotiation and drafting of this Agreement. Consequently, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement. To the extent that any Contract, document, certificate or instrument is represented and warranted to by the Company to be given, delivered, provided or made available by the Company, in order for such Contract, document, certificate or instrument to have been deemed to have been given, delivered, provided and made available to Purchaser or its Representatives, such Contract, document, certificate or instrument shall have been posted to the electronic data site maintained on behalf of the Company for the benefit of Purchaser and its Representatives and Purchaser and its Representatives have been given access to the electronic folders containing such information.

11.12 Counterparts. This Agreement and each Ancillary Document may be executed and delivered (including by facsimile or other electronic transmission) in one or more counterparts, and by the different Parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.

11.13 Legal Representation.

(a) The Parties agree that, notwithstanding the fact that Winston & Strawn, LLP ("*Purchaser Counsel*") may have, prior to Closing, jointly represented Purchaser Parties in connection with this Agreement, the Ancillary Documents and the Contemplated Transactions, and has also represented Purchaser and/or its Affiliates in connection with matters other than the transaction that is the subject of this Agreement, Purchaser Counsel will be permitted in the future, after Closing, to represent Purchaser or its Affiliates in connection with matters in which such Persons are adverse to Purchaser in connection with any disputes arising out of, or related to, this Agreement. The Company hereby agrees in advance to waive any actual or potential conflict of interest that may hereafter arise in connection with Purchaser Counsel's future representation of Purchaser Parties in which the interests of such Person are adverse to the interests of Purchaser in connection with any disputes arising out of, or related to, this Agreement. The Parties acknowledge and agree that, for the purposes of the attorney-client privilege, Purchaser shall be deemed the client of Purchaser Counsel with respect to the negotiation, execution and performance of this Agreement and the Ancillary Documents. All such communications shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to Purchaser (including, after the First Effective Time, the First Surviving Corporation) of any applicable privileges or protections that can or may be asserted to prevent disclosure of any such communications to any third party.

(b) The Parties agree that, notwithstanding the fact that Greenberg Traurig, LLP ("**GT**") may have, prior to Closing, represented the Company in connection with this Agreement, the Ancillary Documents and the Contemplated Transactions, and has also represented the Company and/or its Affiliates in connection with matters other than the transaction that is the subject of this Agreement, GT will be permitted in the future, after Closing, to represent the Company or its Affiliates in connection

with matters in which such Persons are adverse to the Company in connection with any disputes arising out of, or related to, this Agreement. Each Purchaser Party hereby agrees in advance to waive any actual or potential conflict of interest that may hereafter arise in connection with GT's future representation of the Company in which the interests of such Person are adverse to the interests of Purchaser in connection with any disputes arising out of, or related to, this Agreement. The Parties acknowledge and agree that, for the purposes of the attorney-client privilege, the Company shall be deemed the client of GT with respect to the negotiation, execution and performance of this Agreement and the Ancillary Documents. All such communications shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to the Company and shall be controlled by the Company; provided, further, that nothing contained herein shall be deemed to be a waiver by the Company of any applicable privileges or protections that can or may be asserted to prevent disclosure of any such communications to any third party.

70

ARTICLE XII. DEFINITIONS

12.1 Certain Definitions. For purpose of this Agreement, the following capitalized terms have the following meanings:

"Action" means any notice of noncompliance or violation, or any claim, demand, charge, action, suit, litigation, audit, settlement, complaint, stipulation, assessment or arbitration, or any request (including any request for information), inquiry, hearing, proceeding or investigation, by or before any Governmental Authority.

"Acquisition Targets" means the Target Bank and Target Broker-Dealer.

"Affiliate" means, with respect to any Person, any other Person directly or indirectly Controlling, Controlled by, or under common Control with such Person. For further avoidance of doubt, for purposes of this Agreement *"Affiliate"* shall also mean, with respect to the Company, any Person directly or indirectly Controlling, Controlled by, or under common Control with the Company.

"Ancillary Documents" means the Purchaser Support Agreement, the Registration Rights and Lock-Up Agreement, the Contribution Agreement, the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, the First Certificate of Merger, the Second Certificate of Merger, and the other agreements, certificates and instruments to be executed or delivered by any of the Parties hereto in connection with or pursuant to this Agreement.

"Available Closing Cash" means, without duplication, an amount equal to (a) all amounts in the Trust Account (after reduction of for the aggregate amount of payments required to be made in connection with the Redemption), <u>plus</u> (b) if any Potential Financing is consummated pursuant to the terms of this Agreement, the aggregate amount of net cash proceeds received from the consummation any such Potential Financing (if any), <u>plus</u> (c) any net proceeds obtained by the Company as a result of any debt financing arrangements for borrowed money entered into by the Company following the date hereof (if any), <u>plus</u> (d) any other cash of Purchaser or the Target Companies that will be on hand at the Closing, in each case, without giving effect to payment of Company Transaction Expenses or Purchaser Transaction Expenses; *provided that*, with respect to debt financing described in clause (c), only the amounts outstanding immediately following the Closing shall be considered when determining the Available Closing Cash.

"Benefit Plans" of any Person means any and all deferred compensation, executive compensation, incentive compensation, equity purchase or other equity-based compensation plan, severance or termination pay, holiday, vacation or other bonus plan or practice, hospitalization or other medical, life or other insurance, supplemental unemployment benefits, profit sharing, pension, or retirement plan, program, agreement, commitment or arrangement, and each other employee benefit plan, program, agreement or arrangement, including each "employee benefit plan" as such term is defined under Section 3(3) of ERISA, maintained or contributed to or required to be contributed to by a Person for the benefit of any employee or terminated employee of such Person, or with respect to which such Person has any Liability, whether direct or indirect, actual or contingent, whether formal or informal, and whether legally binding or not.

"Business Day" means any day other than a Saturday, Sunday or a legal holiday on which commercial banking institutions in New York, New York are authorized to close for business, excluding as a result of "stay at home," "shelter-inplace," "non-essential employee" or any other similar orders or restrictions or the closure of any physical branch locations at the direction of any governmental authority so long as the electronic funds transfer systems, including for wire transfers, of commercially banking institutions in New York, New York are generally open for use by customers on such day.

"*Code*" means the Internal Revenue Code of 1986, as amended, and any successor statute thereto, as amended. Reference to a specific section of the Code shall include such section and any valid treasury regulation promulgated thereunder.

"*Company Charter*" means the Certificate of Incorporation of the Company, as amended and effective under the WBCA, prior to the Second Effective Time.

"Company Class A Common Stock" means the class A common stock, par value \$0.0001 per share, of the Company.

"Company Class B Common Stock" means the class B common stock, par value \$0.0001 per share, of the Company.

"Company Common Stock" means the Company Class A Common Stock and the Company Class B Common Stock.

"Company Confidential Information" means all confidential or proprietary documents and information concerning the Company or its Representatives, furnished in connection with this Agreement or the Contemplated Transactions; *provided*, *however*, that Company Confidential Information shall not include any information which, (i) at the time of disclosure by Purchaser or its Representatives, is generally available publicly and was not disclosed in breach of this Agreement or (ii) at the time of the disclosure by the Company or its Representatives to Purchaser or its Representatives was previously known by such receiving party without violation of Law or any confidentiality obligation by the Person receiving such Company Confidential Information.

"Company Convertible Securities" means, collectively, any warrants or rights to subscribe for or purchase any capital stock of the Company or securities convertible into or exchangeable for, or that otherwise confer on the holder any right to acquire any capital stock of the Company.

"*Company Equity Value*" means an amount equal to seventy-five million four hundred thousand dollars \$75,400,000 less the Broker Dealer Acquisition Purchase Price and the Bank Acquisition Purchase Price.

"Company Intellectual Property Rights" means (i) all Intellectual Property Rights that are owned, or purported to be owned, by the Company or any of its Affiliates ("Company Owned Intellectual Property Rights"), and (ii) all other Intellectual Property Rights licensed by, purported to be licensed by, used in or held for use in, necessary for or material to the conduct of the business of the Company or any of its Affiliates as currently conducted and as proposed to be conducted ("Third Party Intellectual Property Rights").

"Company Preferred Stock" means the preferred stock, par value \$0.0001 per share, of the Company.

72

"Company Sale" means, other than the Contemplated Transactions, (a) any transaction or series of related transactions that results in any Person or "group" (within the meaning of Section 13(d)(3) of the Exchange Act) acquiring equity securities that represent more than 50% of the total voting power of Purchaser, or (b) a sale or disposition of all or substantially all of the assets of Purchaser and its Subsidiaries on a consolidated basis, in each case that results in Purchaser Common Stock being converted into cash or other consideration (including equity securities of another Person) (other than a transaction where Purchaser Common Stock is converted into equity securities of a Person who has substantially similar ownership to Purchaser immediately prior to such transaction).

"Company Stock" means shares of the Company Common Stock and the Company Preferred Stock. As of the Second Effective Time, the Company's initial Articles of Incorporation authorizes a total of 22,000,000 shares of capital stock, divided into 12,000,000 shares of common stock, at par value \$0.0001 (the "Common Stock") and 10,000,000 shares of preferred stock, at par value \$0.0001 (the "Preferred Stock").

"Company Technology" means (i) all Technology that is owned, or purported to be owned, by the Company or any of its Affiliates ("Company Owned Technology"), and (ii) all other Technology licensed by, purported to be licensed by, used in or held for use in, necessary for or material to the conduct of the business of the Company or any of its Subsidiaries as currently conducted and as proposed to be conducted ("Third Party Technology").

"Company Transaction Expenses" means (a) all fees, costs and expenses of the Company and its Subsidiaries incurred prior to and through the Closing Date in connection with or related to the authorization, preparation, negotiation, execution or performance of this Agreement or any Ancillary Document related hereto and all other matters related to the consummation of this Agreement, whether paid or unpaid prior to the Closing (including the fees, costs, expenses and disbursements of counsel, accountants, advisors and consultants to the Company and its Subsidiaries), (b) any "single trigger" stay, retention, transaction, change of control or other similar bonuses, compensation or amounts paid or payable by the Company solely in connection with the consummation of the Contemplated Transactions (including the employer portion of any employment, withholding, payroll, social security, unemployment or similar Taxes imposed on such amounts, determined assuming (1) such amounts are payable as of the Closing Date, and (2) no deferral of such Taxes has occurred under any COVID-19 legislation relief program), and (c) all fees in connection with filings with Governmental Authorities, including filings or applications required under Antitrust Laws or other materials contemplated by <u>Section 6.9</u>, the Requisite Regulatory Approvals and filings with the SEC and any related printer fees.

"Consent" means any consent, approval, waiver, authorization or Permit of, or notice to or declaration or filing with any Governmental Authority or any other Person.

"Contemplated Transactions" means the consummation by the Company of the transactions contemplated in this Agreement, each Ancillary Agreement, and the other transactions contemplated to be completed in accordance with the terms and conditions of this Agreement, including the transactions contemplated by the Broker-Dealer Acquisition Agreement, the Bank Acquisition Agreement, and the Contribution Agreement.

"Contracts" means all contracts, agreements, binding arrangements, bonds, notes, indentures, mortgages, debt instruments, purchase order, licenses (and all other contracts, agreements or binding arrangements concerning Intellectual Property Rights), franchises, leases and other instruments or obligations of any kind, written or oral (including any amendments and other modifications thereto).

73

"Control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract, or otherwise. "Controlled," "Controlling" and "under common Control with" have correlative meanings. Without limiting the foregoing a Person (the "Controlled Person") shall be deemed Controlled by (a) any other Person (i) owning beneficially, as meant in Rule 13d-3 under the Exchange Act, securities entitling such Person to cast ten percent (10%) or more of the votes for election of directors or equivalent governing authority of the Controlled Person; (b) an officer, director, general partner, partner (other than a limited partner), manager, or member (other than a member having no management authority that is not a Person described in clause (a) above) of the Controlled Person; or (c) a spouse, parent, lineal descendant, sibling, aunt, uncle, niece, nephew, mother-in-law, father-in-law, or brother-in-law of an Affiliate of the Controlled Person or a trust for the benefit of an Affiliate of the Controlled Person is a trustee.

"*Copyrights*" means any works of authorship, mask works and all copyrights therein, including all renewals and extensions, copyright registrations and applications for registration and renewal, and non-registered copyrights.

"Dissenting Shares" means any shares of Company Stock for which a Company Stockholder has exercised appraisal rights pursuant under the WBCA.

"Domain Name" means any network or web site domain name or other universal resource locator (URL).

"Environmental Law" means any Law in any way relating to (a) the protection of human health and safety, (b) the protection, preservation or restoration of the environment and natural resources (including air, water vapor, surface water, groundwater, drinking water supply, surface land, subsurface land, plant and animal life or any other natural resource), or (c) the exposure to, or the use, storage, recycling, treatment, generation, transportation, processing, handling, labeling, production, release or disposal of Hazardous Materials, including the Comprehensive Environmental Response, Compensation and Liability Act, 42 USC. Section 9601 et. seq., the Resource Conservation and Recovery Act, 42 USC. Section 6901 et. seq., the Toxic Substances Control Act, 15 USC. Section 2601 et. seq., the Federal Water Pollution Control Act, 33 USC. Section 1151 et seq., the Clean Air Act, 42 USC. Section 7401 et seq., the Federal Insecticide, Fungicide and Rodenticide Act, 7 USC. Section 111 et. seq., Occupational Safety and Health Act, 29 USC. Section 651 et. seq. (to the extent it relates to exposure to Hazardous Material), the Asbestos Hazard Emergency Response Act, 15 USC. Section 2601 et. seq., the Safe Drinking Water Act, 42 USC. Section 300f et. seq., the Oil Pollution Act of 1990 and analogous state acts.

"Environmental Liabilities" means, in respect of any Person, all Liabilities, obligations, responsibilities, Remedial Actions, losses, damages, costs, and expenses (including all reasonable fees, disbursements, and expenses of counsel, experts, and consultants and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any claim or demand by any other Person or in response to any violation of Environmental Law, whether known or unknown, accrued or contingent, whether based in contract, tort, implied or express warranty, strict liability, criminal or civil statute, to the extent based upon, related to, or arising under or pursuant to any Environmental Law, Environmental Permit, Order, or Contract with any Governmental Authority or other Person, that relates to any environmental, health or safety condition, violation of Environmental Law, or a Release or threatened Release of Hazardous Materials.

74

"ERISA" means the U.S. Employee Retirement Income Security Act of 1974, as amended.

"Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended.

"*Exploit*" means develop, design, test, modify, make, use, sell, have made, used and sold, import, export, copy, reproduce, publish, market, display, perform, distribute, commercialize, license, sublicense, make available, support, maintain, correct, translate and create derivative works of, in any medium or means of storage or transmission, now know or hereafter invented.

"FINRA" means Financial Industry Regulatory Authority.

"GAAP" means generally accepted accounting principles as in effect in the United States of America.

"Governmental Authority" means any federal, state, local, foreign or other governmental, quasi-governmental or administrative body, instrumentality, department or agency, including the Federal Reserve Board, or any self-regulatory organization, including FINRA, or any court, tribunal, administrative hearing body, arbitration panel, commission, or other similar dispute-resolving panel or body.

"Hazardous Material" means any waste, gas, liquid or other substance or material that is defined, listed or designated as a "hazardous substance," "pollutant," "contaminant," "hazardous waste," "regulated substance," "hazardous chemical," or "toxic chemical" (or by any similar term) under any Environmental Law, or any other material regulated, or that could result in the imposition of Liability or responsibility, under any Environmental Law, including petroleum and its by-products, asbestos, polychlorinated biphenyls, radon, mold, and urea formaldehyde insulation.

"Indebtedness" of any Person means, without duplication, (a) all indebtedness of such Person for borrowed money (including the outstanding principal and accrued but unpaid interest), (b) any other indebtedness of such Person that is evidenced by a note, bond, debenture, credit agreement or similar instrument, (c) all obligations of such Person for the reimbursement of any obligor on any line or letter of credit, banker's acceptance, guarantee or similar credit transaction, in each case, that has been drawn or claimed against, (d) any premiums, prepayment fees or other penalties, fees, costs or expenses associated with payment

of any Indebtedness of such Person, and (e) all obligation described in clauses (a) through (e) above of any other Person which is directly or indirectly guaranteed by such Person or which such Person has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a creditor against loss.

"Intellectual Property Rights" means all world-wide intellectual property rights, including all (i) Patents, (ii) Trademarks (registered and unregistered), (iii) copyrights (registered or unregistered), works of authorship and copyrightable works and registrations and applications for registration thereof, (iv) rights in Software, (v) Trade Secrets, (vi) other intellectual property rights (including, without limitation, inventors' rights and moral rights) throughout the world relating to the foregoing (including remedies against infringement thereof and rights of protection of interest therein under the Laws of all jurisdiction).

"Internal System" means the computer, communications network and other information technology systems and any application service provider services, hosted service, "Software-as-a-Service", "Infrastructure-as-a-Service", "Platform-as-a-Service" or similar "cloud" service used for internal purposes by the Company.

"*International Trade Laws*" means all applicable U.S. and non-U.S. laws, regulations, orders, directives, and/or governmental contractual obligations relating to economic and trade sanctions, export controls, export filings, customs and imports, anti-boycott, anti-money laundering, foreign direct investment reviews, and anti-corruption and anti-bribery.

"Internet Names" means all Domain Names and Facebook, Twitter, Instagram, Snapchat, LinkedIn and other social networking names, tags and registrations.

"IPO" means the initial public offering of Purchaser Units pursuant to the IPO Prospectus.

"*IPO Prospectus*" means the final prospectus of Purchaser, dated as of February 8, 2021, and filed with the SEC on February 10, 2021 (File Nos. 333-252226 and 333-252761).

"IPO Underwriter" means Chardan Capital Markets, LLC.

"IPO Underwriter Fees" means the costs and expenses incurred by Purchaser and its Affiliates in connection with the IPO that are due and payable to the IPO Underwriter upon the consummation of the Contemplated Transactions.

"*IRS*" means the U.S. Internal Revenue Service (or any successor Governmental Authority).

"*Knowledge*" means, with respect to (i) the Company, the actual knowledge of Robert McBey and Craig Ridenhour, after reasonable inquiry or (ii) Purchaser, the actual knowledge of John Schaible, Miguel Leon and Daniel Caamano V, after reasonable inquiry.

"*Law*" means any federal, state, local, municipal, foreign or other law, statute, legislation, principle of common law, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, directive, requirement, writ, injunction, settlement, Order or Consent that is or has been issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority.

"*Liabilities*" means any and all liabilities, Indebtedness, Actions or obligations of any nature (whether absolute, accrued, contingent or otherwise, whether known or unknown, whether direct or indirect, whether matured or unmatured, whether due or to become due and whether or not required to be recorded or reflected on a balance sheet under GAAP or other applicable accounting standards), including Tax liabilities due or to become due.

"Lien" means any mortgage, pledge, security interest, attachment, right of first refusal, option, proxy, voting trust, encumbrance, lien or charge of any kind (including any conditional sale or other title retention agreement or lease in the nature thereof), restriction (whether on voting, sale, transfer, disposition or otherwise), any subordination arrangement in favor of another Person, or any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code or any similar Law.

"Material Adverse Effect' means, with respect to any specified Person, any fact, event, occurrence, change or effect that has had, or would reasonably be expected to have, individually or in the aggregate, a material adverse effect upon (a) the business, assets, Liabilities, results of operations, prospects or condition (financial or otherwise) of such Person and its Subsidiaries, taken as a whole, or (b) the ability of such Person or any of its Subsidiaries on a timely basis to consummate the Contemplated Transactions, or to perform its obligations under this Agreement or the Ancillary Documents to which it is a party or bound; provided, however, that for purposes of clause (a) above, any changes or effects directly or indirectly attributable to, resulting from, relating to or arising out of the following (by themselves or when aggregated with any other, changes or effects) shall not be deemed to be, constitute, or be taken into account when determining whether there has or may, would or could have occurred a Material Adverse Effect: (i) general changes in the financial or securities markets or general economic or political conditions in the country or region in which such Person or any of its Subsidiaries do business (including with respect to or as a result of the COVID-19 pandemic); (ii) changes, conditions or effects that generally affect the industries in which such Person or any of its Subsidiaries principally operate (including with respect to or as a result of the COVID-19 pandemic); (iii) changes in GAAP or other applicable accounting principles or mandatory changes in the regulatory accounting requirements applicable to any industry in which such Person and its Subsidiaries principally operate; (iv) conditions caused by acts of God, terrorism, war (whether or not declared) or natural disaster; (v) any failure in and of itself by such Person and its Subsidiaries to meet any internal or published budgets, projections, forecasts or predictions of financial performance for any period (provided that the underlying cause of any such failure may be considered in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent not excluded by another exception herein); (vi) changes or proposed changes in any Law or other binding directives issued by any Governmental Authority; and (vii) with respect to Purchaser, the announcement of this Agreement, the pendency or consummation of the Mergers or the performance of this Agreement, including the impact thereof on Purchaser's stock price and the consummation and effects of the Redemption (or any redemption in connection with the Extension); provided further, however, that any event, occurrence, fact, condition, or change referred to in clauses (i) - (vi) immediately above shall be taken into account in determining whether a Material Adverse Effect has occurred or could reasonably be expected to occur to the extent that such event, occurrence, fact, condition, or change has a disproportionate effect on such Person or any of its Subsidiaries compared to other participants in the industries in which such Person or any of its Subsidiaries primarily conducts its businesses. Notwithstanding the foregoing, with respect to Purchaser, the amount of the Redemption (or any redemption in connection with the Extension, if any) or the failure to obtain the Required Purchaser Stockholder Approval shall not be deemed to be a Material Adverse Effect on or with respect to Purchaser.

"*Merger Consideration*" means the number of shares of New Pubco Common Stock equal to the quotient of (a) the Company Equity Value divided by (b) \$10.00.

"Merger Sub 1 Common Stock" means the shares of common stock, par value \$0.0001 per share, of Merger Sub 1.

"Merger Sub 2 Common Stock" means the shares of common stock, par value \$0.0001 per share, of Merger Sub 2.

"New Pubco Bylaws" means the amended and restated bylaws of New Pubco in a form and substance to be agreed upon by the Parties.

"New Pubco Charter" means the amended and restated certificate of incorporation of New Pubco in a form and substance to be agreed upon by the Parties.

"New Pubco Common Stock" means the common stock, par value \$0.0001 per share, of New Pubco.

"New Pubco Private Warrant" means one of the redeemable private placement warrants entitling the holder to purchase shares of New Pubco Common Stock.

"New Pubco Public Warrant" means one of the redeemable public warrants entitling the holder to purchase shares of New Pubco Common Stock.

"New Pubco Warrants" means New Pubco Private Warrants and New Pubco Public Warrants, collectively.

"NYSE" means the New York Stock Exchange.

"Open Source Software" means any Software that is distributed as "free software" (as defined by the Free Software Foundation), "open source software" (meaning Software distributed under any license approved by the Open Source Initiative as set forth at www.opensource.org) or under a similar licensing or distribution model, including, without limitation, Software licensed or distributed under any of the following licenses or distribution models, or licenses or distribution models similar to any of the following: (i) GNU's General Public License (GPL), Lesser General Public License (LGPL) or Affero General Public License, (ii) the Artistic License, (iii) the Mozilla Public License, (iv) the Netscape Public License, (v) the Sun Community Source License (SCSL), (vi) the Sun Industry Standards Source License (SISSL), (vii) the BSD License, (viii) the Apache Software License, (ix) Business Source License (BSL), and (x) the MIT License (MIT).

"Order" means any order, decree, ruling, judgment, injunction, writ, determination, binding decision, verdict, judicial award or other action that is or has been made, entered, rendered, or otherwise put into effect by or under the authority of any Governmental Authority.

"Organizational Documents" means, with respect to any Person that is an entity, its certificate of incorporation or formation, bylaws, operating agreement, memorandum and articles of association or similar organizational documents, in each case, as amended.

"*Pacsquare LOF*" means that certain letter agreement, dated August 18, 2022, by and between Pacsquare Technologies, LLC, a Texas limited liability company ("*Pacsquare*") and Company.

"*Patents*" means all patents, patent applications, patent disclosures, records of intention, and priority rights, utility models, design registrations, certificates of invention and other governmental grants for the protection of inventions or industrial designs (including additions, provisional applications, national, regional and international applications, substitutions, continuations, continuations-in-part, divisionals, continued prosecution applications, renewals, extensions, revivals, reissues and reexaminations).

"PCAOB" means the U.S. Public Company Accounting Oversight Board (or any successor thereto).

78

"*Per Share Merger Consideration*" means, the Merger Consideration *divided by* the number of shares of outstanding shares of Company Common Stock.

"*Permits*" means all federal, state, local or foreign or other third-party permits, grants, easements, consents, approvals, authorizations, exemptions, licenses, franchises, concessions, ratifications, permissions, clearances, confirmations, endorsements, waivers, certifications, designations, ratings, registrations, qualifications or orders of any Governmental Authority or any other Person.

"Permitted Liens" means (a) Liens for Taxes or assessments and similar governmental charges or levies, which either are (i) not delinquent or (ii) being contested in good faith and by appropriate proceedings, and adequate reserves have been established with respect thereto, (b) other Liens imposed by operation of Law arising in the ordinary course of business for amounts which are not due and payable and as would not in the aggregate materially adversely affect the value of, or materially adversely interfere with the use of, the property subject thereto, (c) Liens incurred or deposits made in the ordinary course of business in connection with social security, (d) Liens on goods in transit incurred pursuant to documentary letters of credit, in each case arising in the ordinary course of business, (e) non-exclusive licenses, sublicenses or other rights to Company Intellectual Property Rights granted to any licensee in the ordinary course of business or (f) Liens arising under this Agreement or any Ancillary Document. "*Person*" means an individual, corporation, partnership (including a general partnership, limited partnership or limited liability partnership), limited liability company, association, trust or other entity or organization, including a government, domestic or foreign, or political subdivision thereof, or an agency or instrumentality thereof.

"Personal Data" means, with respect to any natural Person, such Person's name, street address, telephone number, email address, photograph, social security number, tax identification number, driver's license number, passport number, credit card number, bank account number and other financial information, customer or account numbers, account access codes and passwords, any other information that allows the identification of such Person or enables access to such Person's financial information or that is defined as "personal data," "personally identifiable information," "personal information," "protected health information" or similar term under any applicable Privacy Laws.

"*Personal Property*" means any machinery, equipment, tools, vehicles, furniture, leasehold improvements, office equipment, plant, parts and other tangible personal property.

"*Privacy Laws*" means all applicable United States state and federal Laws, and the applicable laws of any other jurisdictions, relating to privacy and protection of Personal Data, data availability, data destruction and data breach, including security incident notification.

"**Pro Rata Share**" means with respect to each Company Stockholder, a fraction expressed a percentage equal to (i) the portion of the Merger Consideration payable by Purchaser to such Company Stockholder in accordance with the terms of this Agreement, <u>divided by</u> (ii) the total Merger Consideration payable by Purchaser to all Company Stockholders in accordance with the terms of this Agreement, as set forth on the Allocation Schedule.

"*Proceeding*" means any action, suit, proceeding, complaint, claim, charge, hearing, labor dispute, inquiry or investigation before or by a Governmental Authority or an arbitrator.

79

"*Product*" means any product (including any Software, hardware, firmware, server, equipment, tool, device, software as a service, platform as a service, infrastructure as a service or other "cloud" product) now or previously made commercially available (including for use on a service or subscription basis or on a hosted service, "Software-as-a-Service," "Infrastructure-as-a-Service," "Platform-as-a-Service" or similar "cloud" service basis), deployed, marketed, distributed, provided, supported, sold, offered for sale, imported or exported for resale or licensed by or on behalf of the Company or any of its Affiliates, together with all user documentation, Software and other files and information related thereto.

"Protected Health Information" has the meaning given to such term under HIPAA, including all such information in electronic form.

"*Purchaser Certificate of Incorporation*" means the Amended and Restated Certificate of Incorporation of Purchaser, filed with the Secretary of State of the State of Delaware on February 4, 2021.

"Purchaser Common Stock" means the shares of common stock of Purchaser, par value \$0.0001 per share.

"Purchaser Confidential Information" means all confidential or proprietary documents and information concerning Purchaser or any of its Representatives; provided, however, that Purchaser Confidential Information shall not include any information which, (i) at the time of disclosure by the Company, or any of its Representatives, is generally available publicly and was not disclosed in breach of this Agreement or (ii) at the time of the disclosure by Purchaser or its Representatives to the Company, or any of its Representatives, was previously known by such receiving party without violation of Law or any confidentiality obligation by the Person receiving such Purchaser Confidential Information. For the avoidance of doubt, from and after the Closing, Purchaser Confidential Information will include the confidential or proprietary information of the Company.

"Purchaser's Organizational Documents" means the Purchaser Certificate of Incorporation, and the bylaws of Purchaser.

"Purchaser Parties" means, collectively, Purchaser, New Pubco, Merger Sub 1 and Merger Sub 2.

"Purchaser Preferred Stock" means the shares of preferred stock of Purchaser, par value \$0.0001 per share.

"Purchaser Private Warrants" means each warrant to purchase one share of Purchaser Common Stock at an exercise price of \$11.50 per share, subject to adjustment, on the terms and subject to the conditions set forth in the Warrant Agreement.

"*Purchaser Public Warrants*" means each warrant to purchase one-half of one share of Purchaser Common Stock at an exercise price of \$11.50 per share, subject to adjustment, on the terms and subject to the conditions set forth in the Warrant Agreement.

"*Purchaser Securities*" means Purchaser Units, Purchaser Common Stock, Purchaser Preferred Stock and Purchaser Warrants, collectively.

80

"Purchaser Transaction Expenses" means all fees, costs and expenses of Purchaser incurred prior to and through the Closing Date in connection with or related to the authorization, preparation, negotiation, execution or performance of this Agreement or any Ancillary Document related hereto and all other matters related to the consummation of this Agreement, whether paid or unpaid prior to the Closing (including the fees, costs, expenses and disbursements of counsel, accountants, advisors and consultants to Purchaser, and the IPO Underwriter Fees).

"*Purchaser Units*" means the units issued by Purchaser consisting of one (1) share of Purchaser Common Stock and one Purchaser Public Warrant.

"Purchaser Warrants" means Purchaser Private Warrants and Purchaser Public Warrants, collectively.

"*Release*" means any release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, or leaching into the indoor or outdoor environment, or into or out of any property.

"*Remedial Action*" means all actions to (i) clean up, remove, treat, or in any other way address any Hazardous Material, (ii) prevent the Release of any Hazardous Material so it does not endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (iii) perform pre-remedial studies and investigations or post-remedial monitoring and care, or (iv) correct a condition of noncompliance with Environmental Laws.

"*Representatives*" means, as to any Person, such Person's Affiliates and the respective managers, directors, officers, employees, independent contractors, consultants, advisors (including financial advisors, counsel and accountants), agents and other legal representatives of such Person or its Affiliates.

"SEC" means the U.S. Securities and Exchange Commission (or any successor Governmental Authority).

"Securities Act" means the Securities Act of 1933, as amended.

"Services" means any services now or previously provided by the Company or any of its Affiliates to its customers, end users and/or other persons, including without limitation, maintenance services, support services and consulting services (but excluding any Products provided on a hosted service, "Software-as-a-Service," "Infrastructure-as-a-Service," "Platform-as-a-Service" or similar "cloud" service basis).

"Share Price" means, the share price equal to the volume weighted average closing sale price of one share of New Pubco Common Stock as reported on the NYSE (or the exchange on which the shares of New Pubco Common Stock are then listed) over any twenty (20) Trading Days within any thirty (30) consecutive Trading Day period (as equitably adjusted as appropriate to reflect any stock splits, reverse stock splits, stock dividends (including any dividend or distribution of securities convertible into New Pubco Common Stock), extraordinary cash dividend, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change or transaction with respect to New Pubco Common Stock).

"Share Price Earn Out Period" means (i) with respect to Triggering Event I, the time period beginning on the Closing Date and ending on 11:59 p.m. (Eastern Time) on date that is the six (6)-month anniversary of the Closing Date, (ii) with respect to Triggering Event II, the time period beginning on the Closing Date and ending on 11:59 p.m. (Eastern Time) on date that is the twelve (12)-month anniversary of the Closing Date, and (iii) with respect to Triggering Event III, the time period beginning on the Closing Date and ending on 11:59 p.m. (Eastern Time) on date that is the twelve (12)-month anniversary of the Closing Date, and (iii) with respect to Triggering Event III, the time period beginning on the Closing Date and ending on 11:59 p.m. (Eastern Time) on date that is the eighteen (18)-month anniversary of the Closing Date.

"Share Price Earn Out Shares" means 5,944,444 shares of New Pubco Common Stock in the aggregate.

"*Software*" means computer software (including, without limitation, source code, object code, macros, scripts, objects, routines, modules, header files, and other components), data, databases and documentation thereof.

"Software Products Earn Out Shares" means, during each Software Product Earn Out Period, the shares of New Pubco Common Stock issued by New PubCo to Atlas FinTech equal to (i) Software Products Earn Out Revenue *divided by* (ii) the greater of (x) 10.00 or (y) the Share Price of New Pubco Common Stock as of 11:59 p.m. (Eastern Time) on the last day of the Software Product Earn Out Period.

"Software Products Earn Out Revenue" means, with respect to any Software Products Earn Out Period, (i) 50% *multiplied by* (ii) the Software Products Gross Revenue (earned during such Software Product Earn Out Period), which amount shall not exceed \$4,000,000.

"Software Products Earn Out Statement" means a detailed statement of the Software Products Gross Revenue attributable to the applicable Software Products Earn Out Period.

"Software Products Gross Revenue" means, with respect to any Software Products Earn Out Period, the gross revenue of New PubCo or any of its Subsidiaries recognized on account of a Software Products Earn Out Period that is earned, generated from, or produced by, in each case, directly from the ownership, use, sale and operation of the Transferred Software Products (as defined in the Contribution Agreement), which (i) shall be calculated in accordance with GAAP, consistently applied by New PubCo, (ii) shall be based on the audited financial statements of New Pubco and its Subsidiaries for such Software Product Earn Out Period and (iii) shall not include any intercompany revenue line items.

"SOX" means the U.S. Sarbanes-Oxley Act of 2002, as amended.

"Sponsors" means Quantum Ventures LLC, a Delaware limited liability company, and Chardan Quantum LLC, a Delaware limited liability company, collectively.

"Subsidiary" means, with respect to any Person, any corporation, partnership, association or other business entity of which (i) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a partnership, association or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons will be deemed to have a majority of partnership, association or other business entity if such Person or Persons will be allocated a majority of partnership, association or other business entity are son or persons will be allocated a majority of partnership, association or other business entity are son of such partnership, association or other business entity if such Person or Persons will be allocated a majority of partnership, association or other business entity are son of such partnership, association or other business entity. A Subsidiary of a Person will also include any variable interest entity which is consolidated with such Person under applicable accounting rules.

"*Target Companies*" means each of the Company, the Target Broker-Dealer and the Target Bank.

"*Tax Return*" means any return, declaration, report, claim for refund, information return or other documents (including any related or supporting schedules, statements or information) filed or required to be filed in connection with the determination, assessment or collection of any Taxes or the administration of any Laws or administrative requirements relating to any Taxes and including any amendment thereof.

"Taxes" means all direct or indirect federal, state, local, foreign and other net income, gross income, gross receipts, sales, use, value-added, ad valorem, transfer, franchise, profits, license, lease, service, service use, withholding, payroll, employment, social security and related contributions due in relation to the payment of compensation to employees, excise, severance, stamp, occupation, premium, property, windfall profits, alternative minimum, estimated, customs, duties or other taxes, fees, assessments or charges of any kind in the nature of (or similar to) taxes whatsoever, together with any interest and any penalties, additions to tax or additional amounts with respect thereto.

"*Technology*" means all computer hardware and equipment, electronic, electrical, and mechanical equipment, telecommunication networks, equipment, and infrastructure, and all other forms of technology and embodiments of Intellectual Property Rights, including improvements, modifications, works in process, derivatives, or changes, in each case in tangible form.

"*Trade Secrets*" means any trade secrets, confidential business information, and proprietary know-how, including concepts, ideas, designs, plans, research or development information and results, processes, procedures, techniques, technical information, specifications, operating and maintenance manuals, engineering drawings, methods, data (including business, marketing and technical data and customer, prospect and supplier lists), inventions (whether or not patentable and whether or not reduced to practice), and modifications, extensions, and/or improvements of any of the foregoing.

"Trademarks" means any registered or common law trademarks, service marks, trade dress, trade names, logos, Internet Names, corporate names, doing business as designations (DBAs), fictitious names, service marks, and common law trademarks and service marks and trade dress, together with all of the goodwill associated therewith, and any applications for registration of the foregoing.

"*Trading Day*" means any day on which shares of New Pubco Common Stock are actually traded on the principal securities exchange or securities market on which shares of New Pubco Common Stock are then traded.

"Triggering Event I" means if at any time following the Closing but prior to the expiration of the Share Price Earn Out Period, the Share Price of New Pubco Common Stock is greater than or equal to \$12.50.

"Triggering Event II" means if at any time following the Closing but prior to the expiration of the Share Price Earn Out Period, the Share Price of New Pubco Common Stock is greater than or equal to \$15.00.

83

"*Triggering Event III*" means if at any time following the Closing but prior to the expiration of the Share Price Earn Out Period, the Share Price of New Pubco Common Stock is greater than or equal to \$17.50.

"Triggering Events" shall mean collectively, Triggering Event I, Triggering Event II and Triggering Event III, and "Triggering Event" shall mean any one such individual event.

"*Trust Account*" means the trust account established by Purchaser with the proceeds from the IPO pursuant to the Trust Agreement in accordance with the IPO Prospectus.

"*Trust Agreement*" means that certain Investment Management Trust Agreement, dated as of February 4, 2021, as it may be amended, by and between Purchaser and the Trustee, as well as any other agreements entered into related to or governing the Trust Account.

"*Trust Termination Letter*" means the termination letter to the Trustee substantially in the applicable form attached to the Trust Agreement.

"Trustee" means Continental Stock Transfer & Trust Company, in its capacity as trustee under the Trust Agreement.

"*Warrant Agreement*" means that certain Warrant Agreement, dated as of February 4, 2021, as it may be amended, by and between Purchaser and Continental.

"Working Capital Loans" mean any Indebtedness of Purchaser from its Affiliates and stockholders to provide working capital to Purchaser.

12.2 Section References. The following capitalized terms, as used in this Agreement, have the respective meanings given to them in the Section as set forth below adjacent to such terms:

Term	Section
Acquisition Proposal	6.6(a)(i)
Agreement	Preamble
Allocation Schedule	2.4
Alternative Transaction	6.6(a)(ii)
Antitrust Laws	6.9(c)
Atlas FinTech	Recitals
Audited Financials	6.12(a)
Bank Acquisition Agreement	Recitals
Bank Acquisition Purchase Price	Recitals
Broker-Dealer Acquisition Agreement	Recitals
Broker Dealer Acquisition Purchase Price	Recitals
Broker-Dealer Compliance Policies	4.26(c)
Business Combination	10.1
Closing	2.1
Closing Date	2.1
Closing Filing	6.14(b)
Closing Press Release	6.14(b)
Company	Preamble
Company Benefit Plan	4.19(a)
Company Certificate(s)	1.8(b)
Company Directors	6.16(a)
Company Closing Statement	2.2
Company Disclosure Schedules	Article IV
Company Financials	4.7(a)
Company Internet Names	4.13(k)
Company IP	6.2(b)(vii)
Company Material Contract	4.12(a)
Company Permits	4.10
Company Personal Property Leases	4.16
Company Real Property Leases	4.15
Company Registered Intellectual Property	4.13(b)
Company Stockholder(s)	Recitals
Company Stockholders Written Consent	Recitals
Confidentiality Agreement	6.15
Contributed Assets	Recitals
Conversion	1.8

Term	Section
D&O Indemnified Persons	6.17(a)
D&O Tail Insurance	6.17(b)

DGCL	3.5(a)
Enforceability Exceptions	3.2
Environmental Permits	4.20(a)
Exchange Agent	1.8(a)
Exploitation	4.12(a)(xiii)
Extension	6.3(a)
Extension Expenses	6.3(b)(iv)
Federal Reserve Board	3.3
Federal Securities Laws	6.7
First Certificate of Merger	1.2(a)
First Effective Time	1.2(a)
First Merger	Recitals
Firs Surviving Corporation	1.1(a)
Incentive Plan	6.12(a)
Independent Accountant	1.11
Intended Tax Treatment	Recitals
Interim Balance Sheet Date	4.7(a)
Interim Period	6.1(a)
Registration Rights and Lock-Up Agreement	Recitals
Materially Burdensome Regulatory Condition	6.9(d)
McBey	Preamble
Mergers	Recitals
Merger Sub 1	Preamble
Merger Sub 2	Preamble
Merger Subs	Preamble
New Pubco	Preamble
OFAC	3.19(c)
Outside Date	9.1(b)
Potential Financing	6.19
Proxy Statement	6.12(a)
Public Certifications	3.6(a)
Purchaser Material Contract	3.13
Public Stockholders	10.1
Purchaser	Preamble
Purchaser Counsel	11.13(a)
Purchaser Disclosure Schedules	Article III
Purchaser Special Meeting	6.12(a)
Purchaser Stockholder Approval Matters	6.12(a)
Purchaser Support Agreement	Recitals
Redemption	6.12(a)
Related Person	4.21
Released Claims	10.1
Registration Rights and Lock-Up Agreement	Recitals
Required Company Stockholder Approval	8.1(b)
Required Purchaser Stockholder Approval	8.1(a)
Requisite Regulatory Approvals	6.9(c)
SEC Reports	3.6(a)
Second Certificate of Merger	1.2(b)
Second Effective Time	1.2(b)
Second Merger	Recitals
Second Surviving Corporation	1.1(b)
Section 409A Plan	4.19(j)
Share Price Earn Out Recipients	1.11
Signing Filing	6.14(b)
Signing Press Release	6.14(b)
Software Products Earn Out Period	1.11
Target	Recitals
Target Bank	Recitals
Target Broker-Dealer	Recitals

WBCA	1.1
\$12.50 Earn Out Shares	1.11
\$15 Earn Out Shares	1.11
\$17 Earn Out Shares	1.11

{REMAINDER OF PAGE INTENTIONALLY LEFT BLANK; SIGNATURE PAGE FOLLOWS}

85

IN WITNESS WHEREOF, each Party hereto has caused this Agreement to be signed and delivered as of the date first written above.

The Purchaser:

QUANTUM FINTECH ACQUISITION CORPORATION

By: /s/ John Schaible

Name: John Schaible Title: Chief Executive Officer

New Pubco:

CALCULATOR NEW PUBCO, INC.

By: <u>/s/ Steven J. Carlson</u>

Name: Steven J. Carlson Title: Steven J. Carlson

Merger Sub 1:

CALCULATOR MERGER SUB 1, INC.

By: /s/ Steven J. Carlson Name: Steven J. Carlson Title: Steven J. Carlson

Merger Sub 2:

CALCULATOR MERGER SUB 2, INC.

By: /s/ Steven J. Carlson

Name: Steven J. Carlson Title: Steven J. Carlson

[Signature Page to Business Combination Agreement]

IN WITNESS WHEREOF, each Party hereto has caused this Agreement to be signed and delivered as of the date first written above.

The Company:

ATLASCLEAR, INC.

By: /s/ Craig Ridenhour

Name: Craig Ridenhour Title: Chief Business Development Officer

The Company Stockholders

ATLAS FINTECH HOLDINGS, CORP.

- By: /s/ Craig Ridenhour Name: Craig Ridenhour Title: Chief Business Development Officer
- By: /s/ Robert McBey Name: ROBERT MCBEY

[Signature Page to Business Combination Agreement]

Exhibit A

Form of Purchaser Support Agreement

[Intentionally Omitted]

<u>Exhibit B</u>

Form of Registration Rights and Lock-Up Agreement

[Intentionally Omitted]

Exhibit C

Broker-Dealer Acquisition Agreement

[Intentionally Omitted]

<u>Exhibit D</u>

Bank Acquisition Agreement

[Intentionally Omitted]

<u>Exhibit E</u>

Contribution Agreement

[Intentionally Omitted]

Exhibit 10.1

EXECUTION VERSION

PURCHASER SUPPORT AGREEMENT

This Purchaser Support Agreement (this "<u>Agreement</u>") is effective as of November 16, 2022 by and among (i) Quantum Ventures LLC, a Delaware limited liability company (the "<u>Sponsor Holdco</u>"), (ii) the Persons set forth on <u>Schedule I</u> hereto (together with the Sponsor Holdco, each, a "<u>Sponsor</u>" and, together, the "<u>Sponsors</u>"), (iii) Quantum FinTech Acquisition Corporation, a Delaware corporation ("<u>Purchaser</u>"), and (iv) AtlasClear, Inc., a Wyoming corporation (the "<u>Company</u>"). The Sponsors, Purchaser and the Company and the are sometimes referred to herein individually as a "<u>Party</u>" and, collectively, as the "<u>Parties</u>".

RECITALS

WHEREAS, each Sponsor is the holder of record and the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of the shares of Purchaser Common Stock and the Purchaser Warrants as set forth opposite such Sponsor's name on <u>Schedule I</u> attached hereto;

WHEREAS, concurrently herewith, Purchaser, the Company, Calculator New Pubco., Inc., a Delaware corporation and a wholly-owned Subsidiary of Purchaser ("New Pubco"), Calculator Merger Sub 1, Inc., a Delaware corporation and a wholly-owned Subsidiary of New Pubco ("Merger Sub 1"), Calculator Merger Sub 2, Inc., a Wyoming corporation and a wholly-owned Subsidiary of New Pubco ("Merger Sub 2" and, collectively with Merger Sub 1, the "Merger Subs"), Atlas FinTech Holdings Corp. ("Atlas Fintech") and Robert McBey are entering into a Business Combination Agreement (as amended, supplemented, restated or otherwise modified from time to time, the "BCA"; capitalized terms used but not otherwise defined in this Agreement shall have the meanings ascribed to them in the BCA), pursuant to which (and subject to the terms and conditions set forth therein), (a) Purchaser will merge with and into Merger Sub 1 (the "First Merger"), with Purchaser being the surviving entity and a wholly owned subsidiary of New Pubco, and (b) following the First Merger, Merger Sub 2 will merge with and into the Company (the "Second Merger" and together with the First Merger, the "Mergers"), with the Company being the surviving entity and becoming a wholly owned subsidiary of New Pubco;

WHEREAS, concurrently herewith, the Sponsor Holdco shall have delivered to the Company a duly executed copy of the Registration Rights and Lock-up Agreement, by and among New PubCo, the Company, the Sponsor Holdco and certain of the Company's other shareholders or their respective affiliates, as applicable.

WHEREAS, as an inducement to Purchaser and the Company to enter into the BCA and to consummate the transactions contemplated therein, the Parties desire to agree to certain matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and intending to be legally bound hereby, the Parties hereby agree as follows:

ARTICLE I PURCHASER SUPPORT AGREEMENT; COVENANTS; VESTING

Section 1.1 <u>Binding Effect of BCA</u>. Each Sponsor hereby acknowledges that it has read the BCA and this Agreement and has had the opportunity to consult with its tax, legal and other advisors with respect thereto and hereto. Each Sponsor hereby agrees to be bound by and comply with Section 6.6 (*No Solicitation*), Section 6.14 (*Public Announcements*) and Section 6.15 (*Confidentiality*) of the BCA (and any relevant definitions contained in such section) as if each such Sponsor was an original signatory to the BCA with respect to such provisions. Sponsor Holdco further (a) agrees to transfer up to the Purchaser Stock and/or Purchaser Warrants that are held, directly or indirectly, by Atlas Fintech to potential sources of debt, equity or other financing pursuant to Section 6.19 (Financing) of the BCA, which transfer shall be effectuated in connection with, and as of immediately prior to, the Closing, and (b) acknowledges that any Purchaser Common Stock or Purchaser Warrants that remain held, directly or indirectly, by Atlas Fintech as of immediately prior to the Closing shall automatically be forfeited by Atlas Fintech for no consideration and that Sponsor Holdco shall cause such Purchaser Common Stock and Purchaser Warrants to be cancelled effective as of the Closing.

Section 1.2 <u>No Interim Period Transfers</u>. Each Sponsor shall not, during the Interim Period (except, in each case, pursuant to the BCA or otherwise consented to by the Company), (i) sell, transfer, hypothecate, pledge, encumber, assign, hedge, swap, convert or otherwise dispose of (including by merger (including by conversion into securities or other consideration), by tendering into any tender or exchange offer, by testamentary disposition, by operation of Law or otherwise), either voluntarily or involuntarily, any of its Purchaser Common Stock or Purchaser Warrants, (ii) enter into any Contract or option with respect to any transaction specified in clause (i) or any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any of its Purchaser Common Stock or Purchaser Warrants, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii) (any transaction specified in clause (i), (ii) or (iii), a "Transfer"); provided, however, that the foregoing shall not prohibit a Transfer to any Permitted Transferee (as defined below); provided, further, that any such Transfer shall be permitted only if, as a precondition to such Transfer, such Permitted Transferee agrees in a writing, reasonably satisfactory in form and substance to the Company, to assume all of the obligations of the transferor under, and be bound by all of the terms of, this Agreement. Any Transfer in violation of this <u>Section 1.2</u> shall be null and void.

Section 1.3 <u>New Shares</u>. In the event that (a) any Purchaser Common Stock, Purchaser Warrants or other equity securities of Purchaser are issued to a Sponsor during the Interim Period pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of Purchaser Common Stock or Purchaser Warrants of, on or affecting the Purchaser Common Stock or Purchaser Warrants or other warrants or other equity securities of Purchaser during the Interim Period, or (c) a Sponsor acquires the right to vote or share in the voting of any Purchaser Common Stock or other equity securities of Purchaser, collectively the "<u>New Securities</u>"), then such New Securities acquired or purchased by such Sponsor shall be subject to the terms of this Agreement to the same extent as if they constituted the Purchaser Common Stock or Purchaser Warrants owned by such Sponsor shall cause, or permit any Affiliate of Sponsor to cause, any New Securities to be created or issued except as expressly permitted by the BCA or otherwise consented to by the Company.

2

Section 1.4 Reserved.

Section 1.5 Sponsor Agreements.

(a) During the Interim Period, at any meeting of the shareholders of Purchaser, however called, or at any adjournment thereof, or in any other circumstance in which the vote, consent or other approval of the shareholders of Purchaser is sought, in each case, as contemplated by the BCA, each Sponsor shall (i) appear at each such meeting or otherwise cause all of its Purchaser Common Stock to be counted as present thereat for purposes of calculating a quorum and (ii) vote (or cause to be voted), or execute and deliver a written consent (or cause a written consent to be executed and delivered) covering, all of its Purchaser Common Stock:

(i) in favor of the transactions contemplated by the BCA, including the Mergers;

(ii) against any Acquisition Proposal or any proposal relating to an Acquisition Proposal (in each case, other than the transactions contemplated by the BCA, including the Mergers);

(iii) against any merger or business combination agreement or merger (other than the BCA and the Mergers), consolidation, combination, sale of substantial assets, reorganization, recapitalization, dissolution, liquidation or winding up of or by Purchaser; and

(iv) against any proposal, action or agreement that would or would reasonably be expected to (A) impede, frustrate, prevent or nullify any provision of this Agreement, the BCA or the Mergers, (B) result in a breach in any respect of any covenant, representation, warranty or any other obligation or agreement of any party under the BCA or (C) result in any of the conditions set forth in Article VIII of the BCA not being fulfilled.

(b) Each Sponsor shall comply with, and fully perform all of its obligations, covenants and agreements set forth in, (i) Section 1 of the D&O Insider Letter and (ii) the Sponsor Insider Letter (as applicable) (each as defined below).

(c) During the Interim Period, each Sponsor shall not modify or amend any Contract between or among such Sponsor, anyone related by blood, marriage or adoption to such Sponsor or any Affiliate of such Sponsor (other than Purchaser), on the one hand, and Purchaser, on the other hand, including, for the avoidance of doubt, the D&O Insider Letter and the Sponsor Insider Letter (except as expressly contemplated hereby or by the BCA or the Ancillary Documents).

Section 1.6 <u>Reasonable Best Efforts</u>. Without limiting <u>Section 1.5</u>, during the Interim Period, each Sponsor shall (i) take any action as may reasonably be necessary to satisfy the conditions of the Purchaser set forth in Article VIII of the BCA and (ii) not take any action that would reasonably be expected to prevent or delay the satisfaction of any of the conditions to the Mergers and the transactions contemplated by the BCA set forth in Article VIII of the BCA.

3

Section 1.7 <u>Further Assurances</u>. During the Interim Period, at the Company's request and without further consideration, each Sponsor shall execute and deliver, or cause to be executed or delivered, such additional documents and take all such further action as may be reasonably necessary or required to effect the actions and consummate the transactions contemplated by the BCA and the transactions contemplated hereby.

Section 1.8 <u>No Inconsistent Agreement</u>. Each Sponsor hereby agrees that such Sponsor shall not enter into any agreement that would restrict, limit, interfere or otherwise be inconsistent with the performance of such Sponsor's obligations hereunder.

Section 1.9 Defined Terms. For purposes of this Agreement:

(a) the term "<u>D&O Insider Letter</u>" means that certain Letter Agreement, dated February 4, 2021, by and among Purchaser and directors and officers of Purchaser parties thereto;

(b) the term "<u>Existing Registration Rights Agreement</u>" means that certain Registration and Stockholder Rights Agreement, dated as of February 4, 2021, by and among Purchaser, Sponsor Holdco, Chardan Quantum LLC, a Delaware limited liability company ("<u>Chardan</u>") and other Persons parties thereto;

(c) the term "Sponsor Insider Letter" means that certain Letter Agreement, dated February 4, 2021, by and among Purchaser, Sponsor Holdco and Chardan; and

Section 1.10 Waiver of Certain Rights. Each Sponsor hereby irrevocably and unconditionally agrees:

(a) (i) not to (A) demand that Purchaser redeem its or their shares of Purchaser Common Stock in connection with the transactions contemplated by the BCA, including the Mergers, or (B) otherwise participate in any such redemption by tendering or submitting any of its shares of Purchaser Common Stock for redemption and (ii) to waive the provisions of Sections 9.2 of the Purchaser Certificate of Incorporation relating to Redemption Rights (as defined therein);

(b) to waive any anti-dilution or similar protection with respect to any of their Purchaser Common Stock (which for the avoidance of doubt, shall not include any adjustments to the Private or Public Warrants pursuant to any adjustment mechanism set forth in the Warrant Agreement);

(c) (i) that no Working Capital Warrant (as defined in the Warrant Agreement) shall be issued as repayment of any outstanding Working Capital Loans and (ii) not to demand conversion of any outstanding Working Capital Loans into Working Capital Warrants (as defined in the Warrant Agreement); and

(d) not to commence or participate in, and to take all actions necessary to opt out of any class in any class action with respect to, any claim, derivative or otherwise, against Purchaser, the Company, the Company's or Purchaser's Affiliates or any of their respective successors, assigns relating to the negotiation, execution or delivery of this Agreement, the BCA or the consummation of the transactions contemplated by the BCA, including the Mergers.

Section 1.11 <u>Consent to Disclosure</u>. Each Sponsor hereby consents to the publication and disclosure in any announcement or disclosure required by applicable securities Laws, the SEC or any other securities authorities of such Sponsor's identity and ownership of such Sponsor's Purchaser Common Stock (and New Pubco Common Stock after the Closing) and Purchaser Warrants and the nature of such Sponsor's obligations under this Agreement; <u>provided</u> that, prior to any such publication or disclosure the Company and Purchaser will consider in good faith; <u>provided</u>, <u>further</u>, that the foregoing proviso shall not apply to any such publication or disclosure the content of which concerning the foregoing does not substantially differ from any prior such publication or disclosure. Each Sponsor shall promptly provide any information reasonably requested by the Company or Purchaser for any regulatory application or filing made or approval sought in connection with the transactions contemplated by the BCA, including filings with the SEC, except for any information that is subject to attorney-client privilege or confidentiality obligations (<u>provided</u>, that with respect to any confidentiality obligations, such Sponsor, Purchaser and the Company shall cooperate in good faith to enable disclosure of such information to the maximum extent possible in a manner that complies with such confidentiality obligation).

Section 1.12 <u>Termination of Existing Registration Rights</u>. The Registration Rights and Lock-Up Agreement, when executed, shall supersede the Existing Registration Rights Agreement, which shall be of no further force or effect upon (but subject to the consummation of) the Closing.

ARTICLE II <u>REPRESENTATIONS AND WARRANTIES</u>

Section 2.1 <u>Representations and Warranties of the Sponsors</u>. Each Sponsor represents and warrants as of the date hereof to the Company (severally and not jointly and solely with respect to itself, himself or herself and not with respect to any other Sponsor) as follows:

(a) <u>Organization; Due Authorization</u>. If such Sponsor is not an individual, it is duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby are within such Sponsor's corporate, limited liability company or organizational powers and have been duly authorized by all necessary corporate, limited liability company or organizational actions on the part of such Sponsor. If such Sponsor is an individual, such Sponsor has full legal capacity, right and authority to execute and deliver this Agreement and to perform his or her obligations hereunder. This Agreement has been duly executed and delivered by such Sponsor and, assuming due authorization, execution and delivery of this Agreement by each of the Company and Purchaser, this Agreement constitutes a legally valid and binding obligation of such Sponsor, enforceable against such Sponsor in accordance with the terms hereof (except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies). If this Agreement is being executed in a representative or fiduciary capacity, the Person signing this Agreement has full power and authority to enter into this Agreement on behalf of the applicable Sponsor.

5

(b) <u>Ownership</u>. Such Sponsor is the record and beneficial owner (as defined in the Securities Act) of, and has good title to, all of such Sponsor's Purchaser Common Stock and Purchaser Warrants, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose of such Purchaser Common Stock or Purchaser Warrants (other than transfer restrictions under the Securities Act)) affecting any such Purchaser Common Stock or Purchaser Warrants, other than Liens pursuant to (i) this Agreement, (ii) the Purchaser's Organizational Documents, (iii) the BCA or (iv) any applicable securities Laws. Such Sponsor's Purchaser Common Stock and Purchaser Warrants as set forth opposite such Sponsor's Affiliates on the date of this Agreement, and none of such Sponsor's Purchaser Common Stock or Purchaser Warrants are subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of such Purchaser Common Stock or Purchaser Warrants, except as provided hereunder and under the D&O Insider Letter and the Sponsor Insider Letter. Other than the Purchaser Warrants, such Sponsor

does not hold or own any rights to acquire (directly or indirectly) any equity securities of Purchaser or any equity securities convertible into, or which can be exchanged for, equity securities of Purchaser.

(c) <u>No Conflicts</u>. The execution and delivery of this Agreement by such Sponsor does not, and the performance by such Sponsor of his, her or its obligations hereunder will not, (i) if such Sponsor is not an individual, conflict with or result in a violation of the organizational documents of such Sponsor or if such Sponsor is an individual, conflict with the rights of such Sponsor's spouse or domestic partner, as applicable, or (ii) require any consent or approval that has not been given or other action that has not been taken by any Person (including under any Contract binding upon such Sponsor or such Sponsor's Purchaser Common Stock or Purchaser Warrants), in each case, to the extent the absence of such consent, approval or other action would prevent, enjoin or materially delay the performance by such Sponsor of its, his or her obligations under this Agreement.

(d) <u>Litigation</u>. As of the date hereof, there are no Actions pending against such Sponsor, or to the knowledge of such Sponsor threatened against such Sponsor, before (or, in the case of threatened Actions, that would be before) any arbitrator or any Governmental Authority, which in any manner challenges or seeks to prevent, enjoin or materially delay the performance by such Sponsor of its, his or her obligations under this Agreement.

(e) <u>Brokerage Fees</u>. Except as described on Schedule 3.17 of the Purchaser Disclosure Schedules, no broker, finder, investment banker or other Person is entitled to any brokerage fee, finders' fee or other commission in connection with the transactions contemplated by the BCA based upon arrangements made by such Sponsor, for which Purchaser or any of its Affiliates may become liable.

(f) <u>Affiliate Arrangements</u>. Except as set forth on <u>Schedule II</u> attached hereto, no Sponsor nor any anyone related by blood, marriage or adoption to such Sponsor or, to the knowledge of such Sponsor, any Person in which such Sponsor has a direct or indirect legal, contractual or beneficial ownership of 5% or greater is party to, or has any rights with respect to or arising from, any Contract with Purchaser or its Subsidiaries that has not been publicly disclosed in Purchaser's filings with the SEC.

6

(g) <u>Acknowledgment</u>. Such Sponsor understands and acknowledges that each of Purchaser and the Company is entering into the BCA in reliance upon such Sponsor's execution and delivery of this Agreement.

(h) Certain Securities Law Representations and Warranties.

(i) Such Sponsor has never been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked;

(ii) In the case of such Sponsor who is an individual, its biographical information furnished to the Company, if any (including any such information to be included in the Registration Statement), is (or will be when furnished and thereafter) true and accurate in all material respects and does not (or will not when furnished and thereafter) omit any material information with respect to such Sponsor's background;

(iii) Such Sponsor's questionnaire furnished to the Company or Purchaser, if any, is (or will be when furnished and thereafter) true and accurate in all material respects;

(iv) Such Sponsor is not subject to or a respondent in any legal action for any injunction, ceaseand-desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction; and

(v) Such Sponsor has never been convicted of, or pleaded guilty to, any crime (A) involving any fraud, (B) relating to any financial transaction or handling of funds of another Person or (C) pertaining to any dealings in any securities, and it is not currently a defendant in any such criminal proceeding.

Section 2.2 <u>No Actions to Breach Agreement</u>. Each Sponsor shall not take any action that would make any representation or warranty of such Sponsor contained herein untrue or incorrect or have the effect of preventing or disabling such Sponsor from performing its obligations under this Agreement.

Section 2.3 <u>Update of Schedule I</u>. If any Sponsor acquires record or beneficial ownership of any Purchaser Common Stock or Purchaser Warrants during the Interim Period (or becomes aware, during the Interim Period, of its record or beneficial ownership of any Purchaser Common Stock or Purchaser Warrants as of the date hereof, which securities are not already set forth on Schedule I), such Sponsor shall promptly notify the Company and Purchaser in writing, and Schedule I shall be updated to reflect such Sponsor's ownership of such additional Purchaser Common Stock or Purchaser Warrants, as applicable.

7

ARTICLE III MISCELLANEOUS

Section 3.1 Termination; Non-Survival of Representations and Warranties.

(a) This Agreement shall terminate upon the earlier to occur of (i) the termination of the BCA in accordance with its terms in circumstances where the Closing does not occur and (ii) the expiration of the Lock-up Period, and upon such termination, this Agreement shall forthwith become void and have no further force or effect, without any liability on the part of any Party; provided, that (1) no such termination shall relieve any Party of any liability for fraud or intentional and willful breach of this Agreement prior to its termination, and (2) Section 1.1, Section 2.2, this Section 3.1, and Sections 3.2 through 3.9 shall survive any such termination.

(b) None of the representations or warranties contained in this Agreement or in any certificate or other writing delivered pursuant hereto shall survive the Closing.

Section 3.2 <u>Capacity</u>. Notwithstanding anything herein to the contrary, each Sponsor is signing this Agreement solely in such Sponsor's capacity as a shareholder of Purchaser, and not in any other capacity, and this Agreement shall not limit, prevent or otherwise affect the actions of such Sponsor or any Affiliate or Representative thereof, or any of their respective Affiliates, in his, her or its capacity, if applicable, as an officer or director of Purchaser or any other Person, including in the exercise of his, her or its fiduciary duties as a director or officer of Purchaser.

Section 3.3 <u>Notices</u>. All notices, consents, waivers and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered (i) in person, (ii) by facsimile or other electronic means (including e-mail), with affirmative confirmation of receipt, (iii) one Business Day after being sent, if sent by reputable, nationally recognized overnight courier service or (iv) three (3) Business Days after being mailed, if sent by registered or certified mail, pre-paid and return receipt requested, in each case to the applicable Party at the following addresses (or at such other address for a Party as shall be specified by like notice):

If to the Purchaser at or prior to the Closing, to:	with a copy (which will not constitute notice) to:
Quantum FinTech Acquisition Corporation 4221 W. Boy Scout Blvd., Suite 300 Tampa, FL 33607 Attn: Sandip I. Patel E-mail: spatel@qventllc.com	Winston & Strawn LLP 200 Park Avenue New York, NY 10166 Attn: Carol Anne Huff; Jason D. Osborn Facsimile: 212-294-4700 Email: CHuff@winston.com; JOsborn@winston.com
If to a Sponsor, to such Sponsor's address set forth in Schedule I	with a copy to (which will not constitute notice): Greenberg Traurig, LLP 1750 Tysons Boulevard, Suite 1000 McLean, VA 22102 Attn: Jason Simon Facsimile No.: +1 703.714.8386

If to the Company or the Surviving Corporation, to:

AtlasClear, Inc. 4221 W. Boy Scout Blvd., Ste 300 Tampa, FL 33607 Attn: Craig Ridenhour Telephone No.: +1 727.512.9631 E-mail: cridenhour@atlasbanc.com with a copy (which will not constitute notice) to:

Greenberg Traurig, LLP 1750 Tysons Boulevard, Suite 1000 McLean, VA 22102 Attn: Jason Simon Facsimile No.: +1 703.714.8386 Telephone No.: +1 703.749.1386 E-mail: simonj@gtlaw.com

If to the Purchaser after the Closing, to:

Quantum FinTech Acquisition Corporation 4221 W. Boy Scout Blvd., Suite 300 Tampa, FL 33607 Attn: Sandip I. Patel E-mail: spatel@qventllc.com with a copy (which will not constitute notice) to:

Winston & Strawn LLP 200 Park Avenue New York, NY 10166 Attn: Carol Anne Huff; Jason D. Osborn Facsimile: 212-294-4700 Email: CHuff@winston.com; JOsborn@winston.com

Section 3.4 <u>Binding Effect; Assignment</u>. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns. This Agreement shall not be assigned by operation of Law or otherwise without the prior written consent of the Purchaser and the Company, and any assignment without such consent shall be null and void; *provided that* no such assignment shall relieve the assigning Party of its obligations hereunder.

Section 3.5 <u>Third Parties</u>. Nothing contained in this Agreement or in any instrument or document executed by any party in connection with the transactions contemplated hereby shall create any rights in, or be deemed to have been executed for the benefit of, any Person that is not a Party hereto or thereto or a successor or permitted assign of such a Party.

Section 3.6 <u>Governing Law</u>; Jurisdiction. This Agreement shall be governed by, construed and enforced in accordance with the Laws of the State of New Delaware without regard to the conflict of laws principles thereof. All Actions arising out of or relating to this Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware located in Wilmington, Delaware (and in the absence of jurisdiction, the Parties consent to be subject to the jurisdiction of the United States District Court for the District of Delaware or any other state court located in Wilmington, Delaware) (the "<u>Specified Courts</u>"). Each Party hereto hereby (a) submits to the exclusive jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Agreement brought by any Party hereto and (b) irrevocably waives, and agrees not to assert by way of motion, defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any Specified Court. Each Party agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each Party irrevocably consents to the service of the summons and complaint and any other process in any other Action relating to the transactions contemplated by this Agreement, on behalf of itself, or its property, by personal delivery of copies of such process to such Party at the applicable address set forth in <u>Section 3.3</u>. Nothing in this <u>Section 3.6</u> shall affect the right of any Party to serve legal process in any other manner permitted by Law.

Section 3.7 <u>WAIVER OF JURY TRIAL</u>. EACH PARTY HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF ANY ACTION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 3.7</u>.

Section 3.8 Specific Performance. Each Party acknowledges that the rights of each Party to consummate the transactions contemplated hereby are unique, recognizes and affirms that in the event of a breach of this Agreement by any Party, money damages may be inadequate and the non-breaching Parties may have not adequate remedy at law, and agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed by an applicable Party in accordance with their specific terms or were otherwise breached. Accordingly, each Party shall be entitled to seek an injunction or restraining order to prevent breaches of this Agreement and to seek to enforce specifically the terms and provisions hereof, without the requirement to post any bond or other security or to prove that money damages would be inadequate, this being in addition to any other right or remedy to which such Party may be entitled under this Agreement, at law or in equity.

Section 3.9 <u>Severability</u>. In case any provision in this Agreement shall be held invalid, illegal or unenforceable in a jurisdiction, such provision shall be modified or deleted, as to the jurisdiction involved, only to the extent necessary to render the same valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby nor shall the validity, legality or enforceability of such provision be affected thereby in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties will substitute for any invalid, illegal or unenforceable provision a suitable and equitable provision that carries out, so far as may be valid, legal and enforceable, the intent and purpose of such invalid, illegal or unenforceable provision.

Section 3.10 <u>Amendment</u>. This Agreement may be amended, supplemented or modified only by execution of a written instrument signed by the Purchaser and the Company.

Section 3.11 <u>Waiver</u>. The Purchaser on behalf of itself and its Affiliates, and the Company on behalf of itself and its Affiliates, may in its sole discretion (i) extend the time for the performance of any obligation or other act of any other non-Affiliates hereto, (ii) waive any inaccuracy in the representations and warranties by such other non-Affiliates Party contained herein or in any document delivered pursuant hereto and (iii) waive compliance by such other non-Affiliates with any covenant or condition contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the Party or Parties to be bound thereby. Notwithstanding the foregoing, no failure or delay by a Party in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any other right hereunder.

10

Section 3.12 <u>Entire Agreement</u>. This Agreement, the BCA and the documents or instruments referred to herein, including any exhibits and schedules attached hereto, which exhibits and schedules are incorporated herein by reference, together with the Ancillary Documents, embody the entire agreement and understanding of the Parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred to herein or the documents or instruments referred to herein, which collectively supersede all prior agreements and the understandings among the Parties with respect to the subject matter contained herein.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY BLANK]

IN WITNESS WHEREOF, the Sponsors, Purchaser, and the Company have each caused this Agreement to be duly executed as of the date first written above.

SPONSORS:

QUANTUM VENTURES LLC

By: /s/ John Schaible Name: John Schaible Title: Chief Executive Officer

> /s/ John Schaible Name: John Schaible

/s/ Miguel Leon Name: Miguel Leon

/s/ Sandip I. Patel Name: Sandip I. Patel

/s/ Thomas J. Hammond Name: Thomas J. Hammond

/s/ Richard Korhammer Name: Richard Korhammer

/s/ Steven J. Carlson Name: Steven J. Carlson

/s/ Michael Devlin Name: Michael Devlin

[Signature Page to Purchaser Support Agreement]

PURCHASER:

QUANTUM FINTECH ACQUISITION CORPORATION

By: /s/ John Schaible

Name: John Schaible Title: Chief Executive Officer

[Signature Page to Purchaser Support Agreement]

COMPANY:

ATLASCLEAR, INC.

By: /s/ Craig Ridenhour

Name: Craig Ridenhour Title: Chief Business Development Officer

[Signature Page to Sponsor Support Agreement]

Purchaser Purchaser Common Sponsor Warrants Stock Quantum Ventures LLC 4,922,500 3,796,335 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Chardan Quantum LLC 949,084 1,230,625 17 State Street, New York, NY 10004 John Schaible 40,833 0 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Miguel Leon 40,833 0 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Daniel Caamano, V 41,233 0 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Sandip I. Patel 0 41,334 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Thomas J. Hammond 40,833 0 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 **Richard Korhammer** 40,833 0 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607 Steven J. Carlson 0 40,833 c/o Quantum Ventures LLC 4221 W. Boy Scout Blvd., Suite 300, Tampa, FL 33607

<u>Schedule I</u> Sponsor-held Purchaser Common Stock and Purchaser Warrants

[Schedule I to Sponsor Support Agreement]

Schedule II

Affiliate Agreements

- 1. Registration and Stockholder Rights Agreement, dated February 4, 2021, by and between Quantum, Sponsor and each of the executive officers and directors of Quantum
- 2. Stock Escrow Agreement, dated February 4, 2021, by and between Quantum, Continental Stock Transfer & Trust Company, Sponsor and each of the executive officers and directors of Quantum
- 3. Subscription Agreement, dated October 23, 2020, by and between Quantum and Quantum Ventures LLC
- 4. Promissory Note, dated October 1, 2020, issued to Quantum Ventures LLC
- 5. Business Combination Marketing Agreement, dated February 4, 2021, by and between Quantum and Chardan Capital Markets LLC
- 6. Insider Letter (Co-Sponsors), dated as of February 4, 2021, by and between Quantum and Sponsor
- 7. Insider Letter, dated as of February 4, 2021, by and between Quantum and each of the executive officers and directors of Quantum
- Private Placement Warrants Purchase Agreement, dated February 4, 2021, by and between Quantum and Quantum Ventures LLC
- 9. Private Placement Warrants Purchase Agreement, dated February 4, 2021, by and between Quantum, Chardan Capital Markets, LLC and B. Riley Securities, Inc.
- 10. Underwriting Agreement, dated as of February 4, 2021, by and between Quantum, Chardan Capital Markets, LLC and B. Riley Securities, Inc.
- 11. Administrative Service Agreement, dated as of February 4, 2021, by and between Quantum and Quantum Ventures LLC
- 12. Co-Placement Joinder dated September 13, 2021 by and between PSC, Quantum and Chardan Capital Markets LLC, to the Letter agreement for lead placement agent services, dated August 20, 2021, by and between Quantum and PSC.
- 13. Letter Agreement, dated November 9, 2021, between Quantum and Chardan.
- 14. Indemnification Agreements with each of the executive officers and directors of Quantum
 - a. Indemnity Agreement, dated February 4, 2021, by and between Quantum and Miguel Leon

- b. Indemnity Agreement, dated February 4, 2021, by and between Quantum and Richard Korhammer
- c. Indemnity Agreement, dated February 4, 2021, by and between Quantum and Sandip I. Patel
- d. Indemnity Agreement, dated February 4, 2021, by and between Quantum and Steven Carlson

- e. Indemnity Agreement, dated February 4, 2021, by and between Quantum and John M. Schaible
- f. Indemnity Agreement, dated February 4, 2021, by and between Quantum and Thomas Hammond

Exhibit 10.2

EXECUTION VERSION

REGISTRATION RIGHTS AND LOCK-UP AGREEMENT

This Registration Rights and Lock-Up Agreement (this "Agreement") is made and entered into as of November 16, 2022 and effective as of the Second Effective Time (as defined in the Business Combination Agreement (as defined below)) by and among Calculator New Pubco, Inc., a Delaware corporation (the "Company"), and the parties listed on <u>Schedule A</u> hereto (each, a "Holder" and collectively, the "Holders"). Any capitalized term used but not defined herein will have the meaning ascribed to such term in the Business Combination Agreement.

RECITALS

WHEREAS, the Company is party to that certain Business Combination Agreement, dated as of the date hereof (the "Business Combination Agreement"), by and among the Company, Quantum FinTech Acquisition Corporation, a Delaware corporation (together with its successors, the "Purchaser"), Calculator Merger Sub 1, Inc., a Delaware corporation and a wholly-owned subsidiary of the Company ("Merger Sub 1"), Calculator Merger Sub 2, Inc., a Wyoming corporation and a wholly-owned subsidiary of the Company ("Merger Sub 2"), AtlasClear, Inc., a Wyoming corporation ("AtlasClear"), Atlas FinTech Holdings Corp, a Delaware corporation, and Robert McBey, pursuant to which, among other things, (a) Purchaser will merge with and into Merger Sub 1 (the "First Merger"), with Purchaser being the surviving entity and a wholly owned subsidiary of the Company, and (b) following the First Merger, Merger Sub 2 will merge with and into AtlasClear, with AtlasClear being the surviving entity and becoming a wholly owned subsidiary of the Company (the "Second Merger");

WHEREAS, the Purchaser and certain of the Holders designated as Original Holders on Schedule A hereto (the "**Original Holders**") are parties to that certain Registration Rights Agreement, dated as of February 4, 2021 (the "**Prior Agreement**");

WHEREAS, the Original Holders currently hold an aggregate of 5,031,250 shares of the Purchaser's common stock, par value \$0.0001 per share (the "**Purchaser Common Stock**") (excluding shares of Purchaser Common Stock underlying the Purchaser Private Placement Warrants (as defined below));

WHEREAS, certain of the Original Holders currently hold an aggregate of 6,153,125 warrants (the "**Purchaser Private Placement Warrants**") to purchase, at an exercise price of \$11.50 per share, an aggregate of 6,153,125 shares of Purchaser Common Stock;

WHEREAS, at the First Effective Time, pursuant to the First Merger and subject to, and in accordance with, the Business Combination Agreement, the shares of Purchaser Common Stock held by the Original Holders will be automatically cancelled and extinguished and converted into an equal number of shares of common stock of the Company, par value \$0.0001 per share (the "Common Stock");

WHEREAS, at the First Effective Time, pursuant to the First Merger and subject to, and in accordance with, the Business Combination Agreement, the Purchaser Private Placement Warrants held by certain of the Original Holders will be assumed by the Company and converted into warrants (the "**Private Placement Warrants**") to purchase, at an exercise price of \$11.50 per share, an aggregate of 6,153,125 shares of Common Stock;

WHEREAS, at the Second Effective Time, pursuant to the Second Merger and subject to, and in accordance with, the Business Combination Agreement, the shares of common stock of AtlasClear held by certain of the Holders designated as New Holders on Schedule A hereto (the "**New Holders**") will be automatically cancelled and extinguished and converted into the right to receive the Merger Consideration (as defined in the Business Combination Agreement);

WHEREAS, following the closing of the transactions contemplated by the Business Combination Agreement, the New Holders may receive Earnout Shares (as defined in the Business Combination Agreement);

WHEREAS, the parties to the Prior Agreement desire to terminate the Prior Agreement effective following the closing of the transactions contemplated by the Business Combination Agreement; and

WHEREAS, in connection with the transactions contemplated by the Business Combination Agreement, the Company and the Holders desire to enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights following the closing of the transactions contemplated by the Business Combination Agreement with respect to certain securities of the Company, as set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and agreements contained herein, the parties agree as follows:

ARTICLE I DEFINITIONS

Section 1.1 Definitions. For purposes of this Agreement, the following terms and variations thereof have the meanings set forth below:

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or Chief Financial Officer of the Company, after consultation with outside counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.

"Agreement" shall have the meaning given in the Preamble.

"Board" shall mean the Board of Directors of the Company.

2

"Business Combination Agreement" shall have the meaning given in the Recitals hereto.

"Business Day" means a day other than Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by law to close.

"Change in Control" means the transfer (whether by tender offer, merger, stock purchase, consolidation or other similar transaction), in one transaction or a series of related transactions, to a person or group of affiliated persons of the Company's voting securities if, after such transfer, such person or group of affiliated persons would hold more than 50% of outstanding voting securities of the Company (or surviving entity) or would otherwise have the power to control the Board or to direct the operations of the Company.

"Commission" shall mean the Securities and Exchange Commission.

"Common Stock" shall have the meaning given in the Recitals hereto.

"Company" shall have the meaning given in the Preamble.

"Company Shelf Takedown Notice" shall have the meaning given in subsection 2.1.4.

"Demand Registration" shall have the meaning given in subsection 2.2.1.

"Demand Requesting Holder" shall have the meaning given in subsection 2.2.1.

"Demanding Holders" shall have the meaning given in <u>subsection 2.2.1</u>.

"Effectiveness Deadline" shall have the meaning given in subsection 2.1.1.

"Escrow Agent" shall have the meaning given in <u>subsection 5.1.2</u>.

"Escrow Agreement" shall have the meaning given in subsection 5.1.2.

"Escrow Shares" shall have the meaning given in subsection 5.1.2.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.

"Form S-1" means a Registration Statement on Form S-1.

"Form S-3" means a Registration Statement on Form S-3.

"Holders" shall have the meaning given in the Preamble.

"Lock-up Period" shall have the meaning given in <u>subsection 5.1.1</u>.

3

"Maximum Number of Securities" shall have the meaning given in subsection 2.2.4.

"Minimum Amount" shall have the meaning given in <u>subsection 2.1.4</u>.

"Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus in the light of the circumstances under which they were made not misleading.

"New Holders" shall have the meaning given in the Recitals hereto.

"New Registration Statement" shall have the meaning given in subsection 2.1.5.

"Original Holders" shall have the meaning given in the Recitals hereto.

"Piggyback Registration" shall have the meaning given in subsection 2.3.1.

"Prior Agreement" shall have the meaning given in the Recitals hereto.

"Private Placement Warrants" shall have the meaning given in the Recitals hereto.

"Pro Rata" shall have the meaning given in subsection 2.2.4.

"Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Purchaser Common Stock" shall have the meaning given in the Recitals hereto.

"Purchaser Private Placement Warrants" shall have the meaning given in the Recitals hereto.

"Registrable Security", "Registrable Securities" shall mean (a) the Private Placement Warrants (including any shares of Common Stock issued or issuable upon the exercise of any such Private Placement Warrants), (b) any outstanding share of Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the Closing Date (including the Escrow Shares), (c) the Earnout Shares and (d) any other equity security of the Company issued or issuable with respect to any such security described in (a), (b) or (c) above by way of

a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; <u>provided</u>, <u>however</u>, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume or other restrictions or limitations); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

4

"**Registration**" shall mean a registration effected by preparing and filing a Registration Statement in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such Registration Statement becoming effective.

"Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

(A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;

(B) fees and expenses of compliance with securities or blue sky laws (including reasonable and documented fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);

(C) printing, messenger, telephone and delivery expenses;

(D) reasonable and documented fees and disbursements of counsel for the Company;

(E) reasonable and documented fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and

(F) reasonable and documented fees and expenses of one (1) legal counsel selected by (i) the majority-in-interest of the Demanding Holders initiating a Demand Registration to be registered for offer and sale in the applicable Registration, (ii) the majority-in-interest of the Demanding Holders initiating a Shelf Underwritten Offering, or (iii) the majority-in-interest of participating Holders under <u>Section 2.1</u> if the Registration was initiated by the Company for its own account or that of a Company stockholder other than pursuant to rights under this Agreement, in each case to be registered for offer and sale in the applicable Registration.

"Registration Statement" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

"Resale Shelf Registration Statement" shall have the meaning given in subsection 2.1.1.

"Securities Act" shall mean the Securities Act of 1933, as amended from time to time.

"SEC Guidance" shall have the meaning given in subsection 2.1.5.

"Shelf Takedown Notice" shall have the meaning given in subsection 2.1.4.

"Shelf Underwritten Offering" shall have the meaning given in subsection 2.1.4.

"SUO Requesting Holder" shall have the meaning given in subsection 2.1.4.

"**Transfer**" means to, directly or indirectly, sell, transfer, assign, pledge, encumber, hypothecate or similarly dispose of, either voluntarily or involuntarily, or to enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, assignment, pledge, encumbrance, hypothecation or similar disposition of, any interest owned by a person or any interest (including a beneficial interest) in, or the ownership, control or possession of, any interest owned by a person.

"Underwriter" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.

"Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

ARTICLE II REGISTRATION

Section 2.1 Resale Shelf Registration Rights

2.1.1 Registration Statement Covering Resale of Registrable Securities. The Company shall prepare and file or cause to be prepared and filed with the Commission, no later than forty-five (45) days following the Closing Date, a Registration Statement to permit the public resale of all the Registrable Securities held by the Holders from time to time as permitted by Rule 415 of the Securities Act or any successor thereto on the terms and conditions specified in this subsection 2.1.1 (the "Resale Shelf Registration Statement"). The Resale Shelf Registration Statement shall be on Form S-1 (or such other form of registration statement as is then available to permit Registration of such Registrable Securities for resale). The Company shall use reasonable best efforts to cause the Resale Shelf Registration Statement to be declared effective as soon as possible after filing, but in no event later than sixty (60) days following the filing deadline (the "Effectiveness Deadline"); provided, that the Effectiveness Deadline shall be extended to ninety (90) days after the filing deadline if the Registration Statement is reviewed by, and receives comments from, the Commission. Once effective, the Company shall use reasonable best efforts to cause the Resale Shelf Registration Statement to remain effective and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, to ensure that another Registration Statement is available, under the Securities Act at all times until all Registrable Securities have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or have ceased to be Registrable Securities. The Registration Statement filed with the Commission pursuant to this subsection 2.1.1 shall contain a prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Registration Statement (subject to lock-up restrictions provided in Section 5.1 of this Agreement), and shall provide that such Registrable Securities may be sold pursuant to any method or combination of methods legally available to, and requested by, Holders. The Company shall use reasonable best efforts to convert the Resale Shelf Registration Statement on Form S-1 to a Resale Shelf Registration Statement on Form S-3 as promptly as practicable after the Company is eligible to use a Resale Shelf Registration Statement on Form S-3 and have the Resale Shelf Registration Statement on Form S-3 declared effective as promptly as practicable and to cause such Resale Shelf Registration Statement on Form S-3 to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, to ensure that another Registration Statement is available, under the Securities Act at all times until all Registrable Securities have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or have ceased to be Registrable Securities.



2.1.2 <u>Notification and Distribution of Materials</u>. The Company shall notify the Holders in writing of the effectiveness of the Resale Shelf Registration Statement as soon as practicable, and in any event within one (1) Business Day after the Resale Shelf Registration Statement becomes effective, and shall furnish to them, without charge, such number of copies of the Resale Shelf Registration Statement (including any amendments, supplements and exhibits), the Prospectus contained therein (including each preliminary prospectus and all related amendments and supplements) and any documents incorporated by reference in the Resale Shelf

Registration Statement or such other documents as the Holders may reasonably request in order to facilitate the sale of the Registrable Securities in the manner described in the Resale Shelf Registration Statement.

2.1.3 <u>Amendments and Supplements</u>. Subject to the provisions of <u>Section 2.1.1</u> above, the Company shall promptly prepare and file with the Commission from time to time such amendments and supplements to the Resale Shelf Registration Statement and Prospectus used in connection therewith as may be necessary to keep the Resale Shelf Registration Statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all the Registrable Securities. If any Resale Shelf Registration Statement filed pursuant to <u>Section 2.1.1</u> is filed on Form S-3 and thereafter the Company becomes ineligible to use Form S-3 for secondary sales, the Company shall promptly notify the Holders of such ineligibility and use its reasonable best efforts to file a shelf registration on an appropriate form as promptly as practicable to replace the shelf registration statement on Form S-3 and have such replacement Resale Shelf Registration Statement declared effective as promptly as practicable and to cause such replacement Resale Shelf Registration Statement to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Resale Shelf Registration Statement is available or, if not available, that another Resale Shelf Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities; provided, however, that at any time the Company once again becomes eligible to use Form S-3, the Company shall cause such replacement Resale Shelf Registration Statement to be amended, or shall file a new replacement Resale Shelf Registration Statement, such that the Resale Shelf Registration Statement is once again on Form S-3.

2.1.4 At any time and from time to time following the effectiveness of the shelf registration statement required by subsection 2.1.1 or subsection 2.1.2, any Holder may request to sell all or a portion of their Registrable Securities in an underwritten offering that is registered pursuant to such shelf registration statement (a "Shelf Underwritten Offering"); provided that such Holder(s) reasonably expects to sell Registrable Securities yielding aggregate gross proceeds in excess of \$15,000,000 from such Shelf Underwritten Offering (such amount of Registrable Securities, as applicable, the "Minimum Amount"). All requests for a Shelf Underwritten Offering shall be made by giving written notice to the Company (the "Shelf Takedown Notice"). Each Shelf Takedown Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Shelf Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Shelf Underwritten Offering. Within two (2) business days after receipt of any Shelf Takedown Notice, the Company shall give written notice of such requested Shelf Underwritten Offering to all other Holders of Registrable Securities (the "Company Shelf Takedown Notice") and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Shelf Underwritten Offering (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Shelf Underwritten Offering, a "SUO Requesting Holder") shall so notify the Company of its intent to participate in such Shelf Underwritten Offering, in writing, within three (3) business days after the receipt by such Holder of the Company Shelf Takedown Notice. Upon receipt by the Company of any such written notification from a SUO Requesting Holder(s) to the Company, subject to the provisions of subsection 2.2.4, the Company shall include in such Shelf Underwritten Offering all Registrable Securities of such SUO Requesting Holder(s). The Company shall, together with all participating Holders of Registrable Securities of the Company proposing (and permitted) to distribute their securities through such Shelf Underwritten Offering, enter into an underwriting agreement in customary form for such Shelf Underwritten Offering with the managing Underwriter or Underwriters selected by the majority-in-interest of the participating Holders after consultation with the Company and shall take all such other reasonable actions as are reasonably requested by the managing Underwriter or Underwriters in order to facilitate the disposition of such Registrable Securities. In connection with any Shelf Underwritten Offering contemplated by this subsection 2.1.4, subject to Section 3.3 and Article IV, the underwriting agreement into which each Holder and the Company shall enter shall contain representations, covenants, indemnities and other rights and obligations in customary form for such Shelf Underwritten Offering by the Company.

2.1.5 Notwithstanding the registration obligations set forth in this <u>Section 2.1</u>, in the event the Commission informs the Company that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single Registration Statement, the Company agrees to promptly (i) inform each of the Holders thereof and use its reasonable best efforts to file amendments to the Resale Shelf Registration Statement as required by the Commission and/or (ii) withdraw the Resale Shelf Registration Statement and file a new registration statement (a "**New Registration Statement**") on Form S-1 (or such other form of registration statement as is then available to permit Registration of such Registrable Securities for resale); provided, however, that prior to filing such amendment or New Registrable Securities in accordance with any publicly-available written or oral guidance, comments, requirements or requests of the Commission staff (the "**SEC Guidance**"). Notwithstanding any other provision of this Agreement, if any SEC Guidance sets forth a limitation of the number of Registrable Securities permitted to be registered on a particular

⁷

Registration Statement as a secondary offering (and notwithstanding that the Company used its reasonable best efforts to advocate with the Commission for the registration of all or a greater number of Registrable Securities), unless otherwise directed in writing by a Holder as to its Registrable Securities, the number of Registrable Securities to be registered on such Registration Statement will be reduced on a pro rata basis based on the total number of Registrable Securities held by the Holders, subject to a determination by the Commission that certain Holders must be reduced first based on the number of Registrable Securities held by such Holders. In the event the Company amends the Resale Shelf Registration Statement or files a New Registration Statement, as the case may be, under clause (i) or (ii) above, the Company will use its reasonable best efforts to file with the Commission, as promptly as allowed by Commission or SEC Guidance provided to the Company or to registration of securities in general, one or more registration statements on Form S-1 or such other form available to register for resale those Registrable Securities that were not registered for resale on the Resale Shelf Registration Statement, as amended, or the New Registration Statement.

2.1.6 Registrations effected pursuant to this <u>Section 2.1</u> shall not be counted as Demand Registrations effected pursuant to <u>Section 2.2</u>.

Section 2.2 Demand Registration.

2.2.1 Request for Registration. Subject to the provisions of subsection 2.2.4 and Section 2.4 hereof and provided that the Company does not have an effective Registration Statement pursuant to subsection 2.1.1 or subsection 2.1.2 covering Registrable Securities, at any time and from time to time on or after the Effective Time, Holders holding at least \$10,000,000 of the then-outstanding number of Registrable Securities held by all Holders (such Holders, the "Demanding Holders"), may make a written demand for Registration of all or part of their Registrable Securities on Form S-1 (or such other form of registration statement as is then available to permit Registration of such Registrable Securities for resale by such Demanding Holders), which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand, a "Demand Registration"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Demand Requesting Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Demand Requesting Holder(s) to the Company, such Demand Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall (i) file a Registration Statement in respect of all Registrable Securities requested by the Demanding Holders and Demand Requesting Holder(s) pursuant to such Demand Registration, not more than forty five (45) days immediately after the Company's receipt of the Demand Registration, and (ii) effect the Registration thereunder as soon thereafter as practicable. Under no circumstances shall the Company be obligated to effect more than an aggregate of five (5) Demand Registrations under this subsection 2.2.1.

8

2.2.2 Effective Registration. Notwithstanding the provisions of subsection 2.2.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency, the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days after such stop order or injunction is removed, rescinded or otherwise terminated, of such election; provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.

2.2.3 <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.2.4</u> and <u>Section 2.4</u> hereof, if a majority-ininterest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Demand Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.2.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration.

2.2.4 Reduction of Underwritten Offering. If the managing Underwriter or Underwritters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company, the Demanding Holders and the Demand Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Demand Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggyback registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Demand Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Demand Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Demand Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

9

2.2.5 <u>Demand Registration Withdrawal</u>. A majority-in-interest of the combined Demanding Holders and Demand Requesting Holders (if any), pursuant to a Registration under <u>subsection 2.2.1</u> shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration prior to the effectiveness of the Registration. If a majority-in-interest of the Demanding Holders and Demand Requesting Holders (if any), withdraws from a proposed offering pursuant to this <u>Section 2.2.5</u>, then such registration shall not count as a Demand Registration provided for in <u>Section 2.2.2</u>. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this <u>subsection 2.2.5</u>.

Section 2.3 Piggyback Registration.

2.3.1 Piggyback Rights. If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (other than Holders of Registrable Securities, which offerings are covered by Section 2.1 or Section 2.2), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company, (iv) for a dividend reinvestment plan, or (v) filed in connection with any business combination or acquisition involving the Company, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.3.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable

Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.3.1</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

2.3.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of shares of Common Stock or other equity securities that the Company desires to sell, taken together with (i) the shares of Common Stock or other equity securities, if any, as to which Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to this <u>Section 2.3</u>, and (iii) the shares of Common Stock or other equity securities, if any, as to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:

(i)

If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.3.1</u> hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), Common Stock, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities; and

If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.3.1, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.

2.3.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.3.3</u>.

2.3.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.3</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.2</u> hereof.

Section 2.4 Restrictions on Registration Rights. If (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, the Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to <u>subsection 2.2.1</u> and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be materially detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement to be filed in the right to defer such filing for a period of not more than thirty (30) days; provided, however, that the Company shall not defer its obligation in this manner more than once in any 12 month period.

11

ARTICLE III COMPANY PROCEDURES

Section 3.1 General Procedures. If at any time on or after the Effective Time the Company is required to effect the Registration of Registrable Securities, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:

3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;

3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by the Holders or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;

3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;

3.1.4 prior to any public offering of Registrable Securities, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action reasonably necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be reasonably necessary, following opinion of Company legal counsel, by virtue of the business and operations of the Company and do any and all other acts and things that may be reasonably necessary or advisable, following opinion of Company legal counsel, to enable the Holders of Registrable Securities included in such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;

3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;

3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

3.1.7 advise each Holder of such Registrable Securities covered by such Registration Statement, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;

3.1.8 advise each Holder of Registrable Securities covered by such Registration Statement, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any Prospectus forming a part of such registration statement has been filed;

3.1.9 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus, furnish a copy thereof to each Holder of such Registrable Securities or its counsel;

3.1.10 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in <u>Section 3.4</u> hereof;

3.1.11 permit a representative of the Holders, the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; <u>provided</u>, <u>however</u>, that such representatives or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;

13

3.1.12 obtain a "cold comfort" letter from the Company's independent registered public accountants (and the independent accountant of any other entity whose financial statements are included in (or incorporated by reference in) a Registration Statement) in the event of an Underwritten Registration, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders and such managing Underwriter;

3.1.13 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;

3.1.14 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;

3.1.15 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of <u>Section 11(a)</u> of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);

3.1.16 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$20,000,000, use its reasonable best efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and

3.1.17 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.

Section 3.2 Registration Expenses. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "*Registration Expenses*," all reasonable fees and expenses of any legal counsel representing the Holders.

14

Section 3.3 Requirements for Participation in Underwritten Offerings. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.

Section 3.4 Suspension of Sales; Adverse Disclosure. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until he, she or it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than thirty (30) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this <u>Section 3.4</u>.

Section 3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the Closing Date pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings upon request. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Registrable Securities held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

Section 3.6 Limitations on Registration Rights. The Company shall not hereafter enter into any agreement with respect to its securities which is inconsistent with or violates the rights granted to the Holders of Registrable Securities in this Agreement and in the event of any conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

Section 4.1 Indemnification

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and agents and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including reasonable and documented attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein.

4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement.

4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

16

4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of Registrable Securities. The Company and each Holder of Registrable Securities participating in an offering also agree to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.

4.1.5 If the indemnification provided under <u>Section 4.1</u> hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by,

or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; <u>provided</u>, <u>however</u>, that the liability of any Holder under this <u>subsection 4.1.5</u> shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in <u>subsections 4.1.1</u>, <u>4.1.2</u> and <u>4.1.3</u> above, any reasonable and documented legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>subsection 4.1.5</u> were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this <u>subsection 4.1.5</u>. No person guilty of fraudulent misrepresentation (within the meaning of <u>Section 11(f)</u> of the Securities Act) shall be entitled to contribution pursuant to this <u>subsection 4.1.5</u> from any person who was not guilty of such fraudulent misrepresentation.

ARTICLE V LOCK-UP

Section 5.1 Lock-Up.

5.1.1 Except as permitted by <u>Section 5.2</u>, until the earliest of: (i) the date that is one (1) year from the Closing Date, (ii) the last consecutive trading day where the sale price of the Common Stock equals or exceeds \$12.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing after the Closing Date, or (iii) such date on which the Company completes a liquidation, merger, stock exchange or other similar transaction which results in all of the stockholders of such entity having the right to exchange their shares of Common Stock for cash, securities or other property (the "Lock-up Period"), each New Holder shall not Transfer any shares of Common Stock beneficially owned or owned of record by such Holder.

17

5.1.2 The Original Holders shall not Transfer (a) any shares of Common Stock beneficially owned or owned of record by such Original Holder prior to the Purchaser's initial public offering (the "Escrow Shares"), other than as provided in Section 4.3 of the Escrow Agreement, dated February 4, 2021, (the "Escrow Agreement") by and among the Purchaser, Continental Stock Transfer & Trust Company (the "Escrow Agent") and the Original Holders. The Escrow Agent shall hold (i) 50% of the Escrow Shares, until the earlier of (x) six months after the Closing Date and (y) the date on which the closing price of the Common Stock equals or exceeds \$12.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing after the Closing Date and (ii) the remaining 50% of the Escrow Shares, until six months after the Closing Date; provided, however, that if, within six months after the Closing Date, the Company subsequently consummates a liquidation, merger, stock exchange or other similar transaction which results in all of the stockholders of such entity having the right to exchange their shares of Common Stock for cash, securities or other property, then the Escrow Agent will, upon the requisite notice, release the Escrow Shares to the Original Holders, or (b) any Private Placement Warrants (including any shares of Common Stock issued or issuable upon the exercise of any such Private Placement Warrants) until 30 days after the Closing Date, other than as provided in the Private Placement Warrants Purchaser and certain Original Holders, dated February 4, 2021.

Section 5.2 Exceptions. The provisions of <u>Section 5.1.1</u> shall not apply to:

5.2.1 transactions relating to shares of Common Stock acquired in open market transactions;

5.2.2 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock as a bona fide gift;

5.2.3 Transfers of shares of Common Stock to a trust, or other entity formed for estate planning purposes for the primary benefit of the spouse, domestic partner, parent, sibling, child or grandchild of the Holder or any other person with whom the Holder has a relationship by blood, marriage or adoption not more remote than first cousin;

5.2.4 Transfers by will or intestate succession upon the death of the Holder;

5.2.5 the Transfer of shares of Common Stock pursuant to a qualified domestic order or in connection with a divorce settlement;

5.2.6 if the Holder is a corporation, partnership (whether general, limited or otherwise), limited liability company, trust or other business entity, (i) Transfers to another corporation, partnership, limited liability company, trust or other business entity that controls, is controlled by or is under common control or management with the undersigned, (ii) distributions of shares of Common Stock to partners, limited liability company members or stockholders of the undersigned;

5.2.7 Transfers to the Company's officers, directors or their affiliates;

5.2.8 pledges of shares of Common Stock or other Registrable Securities as security or collateral in connection with any borrowing or the incurrence of any indebtedness by any Holder (provided such borrowing or incurrence of indebtedness is secured by a portfolio of assets or equity interests issued by multiple issuers);

5.2.9 pursuant to a bona fide third-party tender offer, merger, stock sale, recapitalization, consolidation or other transaction involving a Change in Control of the Company, provided that in the event that such tender offer, merger, recapitalization, consolidation or other such transaction is not completed, the Common Stock subject to this Agreement shall remain subject to this Agreement; and

5.2.10 the establishment of a trading plan pursuant to Rule 10b5-1 promulgated under the Exchange Act, provided that such plan does not provide for the transfer of Common Stock or any securities convertible into or exercisable or exchangeable for Common Stock during the Lock-up Period;

provided, that in the case of any Transfer or distribution pursuant to <u>Sections 5.2.2</u> through <u>5.2.7</u>, each donee, distributee or other transferee shall agree in writing, in form and substance reasonably satisfactory to the Company, to be bound by the provisions of this Agreement.

ARTICLE VI GENERAL PROVISIONS

Section 6.1 Entire Agreement. This Agreement (including <u>Schedule A</u> hereto) constitutes the entire understanding and agreement between the parties as to the matters covered herein and supersedes and replaces any prior understanding, agreement or statement of intent, in each case, written or oral, of any and every nature with respect thereto.

Section 6.2 Prior Agreement. The Purchaser and the Original Holders, as parties to the Prior Agreement, hereby agree that the Prior Agreement is terminated with respect to such parties as of the Closing Date and is replaced in its entirety by this Agreement and none of the Original Holders shall have any further rights thereunder.

Section 6.3 Notices. Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) upon receipt when delivered by hand, (b) upon transmission, if sent by facsimile or electronic transmission (in each case with receipt verified by electronic confirmation), or (c) one (1) Business Day after being sent by courier or express delivery service, specifying next day delivery, with proof of receipt. The addresses, email addresses and facsimile numbers for such notices and communications are those set forth on the signature pages hereof, or such other address, email address or facsimile numbers as may be designated in writing hereafter, in the same manner, by any such person.

Section 6.4 Assignment; No Third-Party Beneficiaries. This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part. This Agreement and the rights, duties and obligations of the Holders of Registrable Securities hereunder may be freely assigned or delegated by such Holder of Registrable Securities in conjunction with and to the extent of any transfer of Registrable Securities by any such Holder. This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and the permitted assigns of the applicable Holder of Registrable Securities. This Agreement is not intended to confer any rights or benefits on any persons that are not party hereto other than as expressly set forth in <u>Article IV</u> and this <u>Section 6.4</u>. No assignment by any

party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement).

Section 6.5 Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties, it being understood that all parties need not sign the same counterpart and such counterparts may be delivered by the parties hereto via facsimile or electronic transmission.

Section 6.6 Amendment; Waiver. This Agreement may be amended or modified, and any provision hereof may be waived, in whole or in part, at any time pursuant to an agreement in writing executed by the Company and Holders holding a majority of the Registrable Securities at such time; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in his, her or its capacity as a Holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected; provided further, that this Agreement may be amended by the Company, without the consent of any of the Holders, to (i) add as "Original Holders" Chardan Quantum LLC and any securities held by Chardan Quantum LLC at the Closing Date as "Registrable Securities", (ii) add as "Holders" any stockholders of Commercial Bancorp that receive securities of the Company in a private placement pursuant to the Amended and Restated Agreement and Plan of Merger, dated as of the date hereof, between the Company, AtlasClear, CB Merger Sub, Inc., a subsidiary of AtlasClear, and Commercial Bancorp and (iii) include any such securities as "Registrable Securities". Any failure by any party at any time to enforce any of the provisions of this Agreement shall not be construed a waiver of such provision or any other provisions hereof.

Section 6.7 Severability. In the event that any provision of this Agreement or the application thereof becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other persons or circumstances will be interpreted so as reasonably to effect the intent of the parties hereto.

20

Section 6.8 Governing Law; Venue. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware applicable to contracts executed in and to be performed in that State. All legal actions and proceedings arising out of or relating to this Agreement shall be heard and determined exclusively in any Delaware Chancery Court; provided, that if jurisdiction is not then available in the Delaware Chancery Court, then any such legal action may be brought in any federal court located in the State of Delaware or any other Delaware state court. The parties hereto hereby (a) irrevocably submit to the exclusive jurisdiction of the aforesaid courts for themselves and with respect to their respective properties for the purpose of any action arising out of or relating to this Agreement brought by any party hereto, and (b) agree not to commence any action relating thereto except in the courts described above in Delaware, other than actions in any court of competent jurisdiction to enforce any judgment, decree or award rendered by any such court in Delaware as described herein. Each of the parties further agrees that notice as provided herein shall constitute sufficient service of process and the parties further waive any argument that such service is insufficient. Each of the parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any action arising out of or relating to this Agreement or the transactions contemplated hereby, (a) any claim that it is not personally subject to the jurisdiction of the courts in Delaware as described herein for any reason, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) that (i) the action in any such court is brought in an inconvenient forum, (ii) the venue of such action is improper or (iii) this Agreement, or the subject matter hereof, may not be enforced in or by such courts.

Section 6.9 Specific Performance. Each party acknowledges and agrees that the other parties hereto would be irreparably harmed and would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed by such first party in accordance with their specific terms or were otherwise breached by such first party. Accordingly, each party agrees that the other parties hereto shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which such parties are entitled at law or in equity.

Section 6.10 Term. This Agreement shall terminate (a) with respect to any Holder on the date on which such Holder ceases to hold Registrable Securities and (b) otherwise upon the date as of which all of the Registrable Securities have been sold pursuant to a Registration Statement (but in each case in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)). The provisions of Article IV shall survive any termination.

(Next Page is Signature Page)

21

IN WITNESS WHEREOF, each of the parties has executed this Agreement as of the date first written above.

COMPANY:

QUANTUM	FINTECH	ACQUISITION
CORPORATION		

By: /s/ John Schaible Name: John Schaible Title: Chief Executive Officer

Address for Notice:

4221 W. Boy Scout Blvd., Suite 300 Tampa, FL 33607

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

Name:Quantum Ventures LLC

By: /s/ John Schaible

Name: John Schaible Title: Chief Executive Officer

Address for Notice: [Intentionally Omitted]

Telephone No.:

Facsimile No.:

Email Address:

[Signature Page to Registration Rights Agreement]

HOLDER:

/s/ John Schaible	
Name: John Schaible	

Address for Notice: [Intentionally Omitted]

Telephone No.:

Facsimile No.:

Email Address:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

/s/ Miguel Leon Name: Miguel Leon

Address for Notice: [Intentionally Omitted]

Telephone No.:

Facsimile No.:

Email Address:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

/s/ Sandip I. Patel

Name: Sandip I. Patel

Address for Noti	ce: [Intentionally Omitted]
Telephone No.:	
Facsimile No.:	
Email Address:	
[Signature Page to Registration Rights Agreement]	

HOLDER:

/s/ Thomas J. Hamm	ond	
Name: Thomas J. Ha	Name: Thomas J. Hammond	
Address for Notice:	[Intentionally Omitted]	
Telephone No.:		
Facsimile No.:		
Email Address:		
[Signature Page to Registration Rights Agreement]		

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

/s/ Richard Korhamn	ner
Name: Richard Korha	ammer
Address for Notice:	Intentionally Omitted]
Telephone No.:	
Facsimile No.:	
Email Address:	
Rights Agreement]	

[Signature Page to Registration Rights Agreement]

HOLDER:

/s/ Steven J. Carlson Name: Steven J. Carlson

Address for Notice: [Intentionally Omitted]

Telephone No.:

Facsimile No.:

Email Address:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

/s/ Michael Devlin Name: Michael Devlin

Address for Notice: [Intentionally Omitted]

Telephone No.:

Facsimile No.:

Email Address:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date first written above.

HOLDER:

/s/ Robert McBey

Name: Robert McBey

	Address for Notice: [Intentionally Omitted]
	Telephone No.:
	Facsimile No.:
	Email Address:
[Signature Page to Registration Right	s Agreement]

HOLDER:

Name: Atlas FinTech Holdings Corp.

By:	/s/ Craig Ridenhour	
	Name: Craig Ridenhour	
	Title: Chief Business Development Officer	
Address for Notice: [Intentionally Omitted]		
Tele	phone No.:	
Facs	imile No.:	
Ema	il Address:	
[Signature Page to Registration Rights Agreement]		

<u>Schedule A</u>

Original Holders

Name of Holder	Number of Shares of Common Stock	Number of Private Placement Warrants
Quantum Ventures LLC	3,796,335	4,922,500
John Schaible	40,833	0
Miguel Leon	40,833	0
Daniel Caamano, V	41,233	0
Sandip I. Patel	41,334	0
Thomas J. Hammond	40,833	0
Richard Korhammer	40,833	0
Steven J. Carlson	40,833	0
Michael Devlin		

New Holders

Name of Holder

Atlas FinTech Holdings Corp. Robert McBey Number of Shares of Common Stock

AGREEMENT AND PLAN OF MERGER

dated as of

November 16, 2022

by and among AtlasClear, Inc.

And

Commercial Bancorp

And, with respect to Section 6.16 only,

Calculator New Pubco, Inc.

TABLE OF CONTENTS

AGRI	EEMENT AND PLAN OF MERGER	1
RECI	TALS	1
ARTI	CLE I CERTAIN DEFINITIONS	2
1.01	Certain Definitions	2
ARTI	CLE II THE MERGER	8
2.01	The Merger	8
2.02	Effective Time	8
2.03	Escrow Deposit	8
ARTI	CLE III CONSIDERATION; EXCHANGE PROCEDURES	8
3.01	Calculation of Merger Consideration	8
3.02	Effect on Capital Stock	9
3.03	Rights as Shareholders; Stock Transfers	9
3.04	Election Procedures; Exchange Procedures	9
	CLE IV ACTIONS PENDING ACQUISITION	12
4.01	Forbearances of CB and Bank	12
4.02	Forbearances of Clear	13
	CLE V REPRESENTATIONS AND WARRANTIES	14
5.01	Disclosure Schedules	14
5.02	Representations and Warranties of CB and Bank	14
5.03	Representations and Warranties of Clear	24
	CLE VI COVENANTS	25
6.01	Reasonable Best Efforts	25
6.02	Shareholder Approval; Appraisal Rights	25
6.03	Press Releases	26
6.04	Access; Information; Board Observation Rights	26
6.05	Acquisition Proposals	27

6.06	Reports and Filings	27
6.07	Regulatory Applications	28
6.08	Notification of Certain Matters	28
6.09	No Breaches of Representations or Warranties	28
6.10	Consents	28
6.11	Insurance Coverage	28

i

6.12	Correction of Information	28
6.13	Confidentiality	28
6.14	Allowance for Loan and Lease Losses; Pre-Closing Loan Review	28
6.15	Continuity of Interest	29
6.16	Offering of New Pubco Stock	29
ARTIC	LE VII ESCROW ACCOUNT	30
ARTIC	LE VIII. CONDITIONS TO CONSUMMATION OF THE MERGER	30
8.01	Conditions to Each Party's Obligation to Effect the Merger	30
8.02	Conditions to Obligation of CB	31
8.03	Conditions to Obligation of Clear	31
	LE IX TERMINATION	32
9.01	Termination by Mutual Consent	32
9.02	Termination by Either Clear or CB	32
9.03	Termination by CB	32
9.04	Termination by Clear	32
9.05	Effect of Termination and Abandonment	32
	LE X MISCELLANEOUS	33
10.01	Survival	33
10.02	Waiver; Amendment	33
10.03	Counterparts	33
10.04	Governing Law and Venue; Waiver of Jury Trial	33
10.05	Counterparts	33
10.06	Expenses	33
10.07	Notices	33
10.08	Entire Understanding; No Third Party Beneficiaries	35
10.09	Interpretation; Effect	35
10.10	Severability	35
10.11	Assignment	35

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER, dated as of November 16, 2022 (this "Agreement"), by and among AtlasClear, Inc., a Wyoming corporation ("Clear"), Commercial Bancorp, a Wyoming corporation ("CB"), and, with respect to <u>Section 6.16</u> only, Calculator New Pubco, Inc., a Delaware corporation ("New Pubco").

RECITALS

CB is a Wyoming corporation, a registered bank holding company and the owner of 100% of the outstanding stock of Farmers State Bank (the "Bank"), located in Pine Bluffs, Wyoming.

A. As of the date hereof, Clear and Quantum FinTech Acquisition Corp. ("QFTA"), Calculator New Pubco, Inc. ("New Pubco"), Calculator Merger Sub 1, Inc., Calculator Merger Sub 2, Inc., Atlas FinTech Holdings Corp. and Robert McBey are entering into a Business Combination Agreement (the "Business Combination Agreement") providing for the acquisition by New Pubco of Quantum and Clear;

B. It is a condition to the consummation of the transactions contemplated by the Business Combination Agreement that Clear enter into this Agreement.

C. It is a condition to the transaction contemplated by this Agreement that the transactions contemplated by the Business Combination Agreement have occurred.

D. It is contemplated that following the date of this Agreement, (i) New Pubco will send to CB shareholders either a Prospectus or an Offering Memorandum (each as defined herein) setting forth information with respect to the transactions contemplated by the Business Combination Agreement, New Pubco and the expected operations of New Pubco giving effect to the transactions contemplated by the Business Combination Agreement; and (ii) following receipt of the Offering Memorandum, CB shareholders will be asked to approve this Agreement and, in the event this Agreement is approved, elect to receive the Option A merger consideration or Option B merger consideration as described herein.

E. The respective boards of directors of Clear and CB have determined that it is in the best interests of their respective companies and their shareholders to consummate the Merger and the other transactions contemplated by this Agreement.

NOW, THEREFORE, in consideration of the foregoing premises and of the mutual covenants, representations, warranties and agreements contained herein, intending to be legally bound hereby, the parties agree as follows:

1

ARTICLE I CERTAIN DEFINITION

1.01 Certain Definitions. The following terms are used in this Agreement with the meanings set forth below:

"Acquisition Proposal" has the meaning set forth in Section 6.05.

"Adjusted Estimated Book Value" ("AEBV") means the mutually agreed upon book value of CB, to be used to calculate the "Merger Consideration"

"Affiliate" means, as to any Person, any other Person which, directly or indirectly, is in control of, is controlled by or is under common control with such Person. For purposes of this definition, "control" of a Person shall mean the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors or other management of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.

"Agreement" means this Agreement, as amended or modified from time to time in accordance with Section 10.02.

"Anti-Money Laundering Laws" has the meaning set forth in Section 5.02(x)

"Appraisal Rights" means a shareholders right to appraisal pursuant to the WBCA

"Appraised Value" means the conclusive assignment of value reached by a professional commercial real estate appraiser jointly selected and retained by CB and Clear, provided said assigned value is not less than \$400,000

"Bank" has the meaning set forth in the preamble to this Agreement.

"Bank Articles" means the Articles of Incorporation of Bank.

"Bank Board" means the Board of Directors of Bank.

"Bank Bylaws" means the Bylaws of Bank.

"Banking Code" means the Wyoming Banking Code, as amended.

"Bank Common Stock" means the common stock of Bank.

"Bank Equity Capital" means the total equity capital of the Bank as of the Calculation Date, determined using Call report Methodology.

"Bank Financial Statements" has the meaning set forth in Section 5.02(g).

"BHC Act" means the Bank Holding Company Act of 1956, as amended.

"Business Combination Agreement" has the meaning set forth in the preamble to this Agreement.

2

"Business Day" means Monday through Friday of each week, except a legal holiday recognized as such by the United States federal government or any day on which banking institutions in the State of Wyoming are authorized or obligated to close.

"Calculation Date" means the last day of the month prior to the Effective Time.

"Calculation Statement" has the meaning set forth in Section 3.01.

"Certificate" has the meaning set forth in Section 3.02.

"Clear Board" means the Board of Directors of Clear.

"Code" means the Internal Revenue Code of 1986, as amended.

"Disclosure Schedule" has the meaning set forth in Section 5.01.

"Closing Date" shall have the meaning set forth in Section 2.02.

"Earnest Money" means the sum of \$100,000 to be deposited by Clear with the Escrow Agent upon mutual execution of this Agreement.

"Election Deadline" has the meaning set forth in Section 3.04 (b).

"Election Form" has the meaning set forth in Section 3.04 (b).

"Effective Date" means the date in which the Effective Time falls.

"Effective Time" has the meaning set forth in Section 2.02.

"Equity Capital" means the total equity capital of CB as of the Calculation Date, determined using Federal Reserve Report methodology for the preparation of the applicable Federal Reserve form for registered small bank holding companies.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any Person treated as a single employer with the Company under Section 414(b), (c), (m), or (o) of the Code.

"Exchange Fund" has the meaning set forth in Section 3.04(b).

"Escrow Account" means the account maintained by the Escrow Agent.

"Escrow Agent" means a qualified entity mutually agreed upon by the parties.

"FDIA" has the meaning set forth in Section 5.02(v).

"FDIC" means the Federal Deposit Insurance Corporation.

"Federal Reserve Board" means the Board of Governors of the Federal Reserve System.

"GAAP" means accounting principles generally accepted in the United States.

"Governmental Authority" means any court, administrative agency or commission or other federal, state or local governmental authority or instrumentality.

"Hazardous Material" means, collectively, (i) any "hazardous substance" as defined by CERCLA, (ii) any "hazardous waste" as defined by the Resource Conservation and Recovery Act, as amended through the date hereof, and (iii) any pollutant or contaminant or hazardous, dangerous or toxic chemical, material or substance within the meaning of any other Federal, state or local law, regulation, ordinance or requirement (including consent decrees and administrative orders) relating to or imposing liability or standards of conduct concerning any hazardous, toxic or dangerous waste, substance or material, all as now in effect.

"Holdback Funds" means the funds in the Escrow held back at Closing Date.

"Investor Questionnaire" has the meaning set forth in Section 6.16.

"IRS" has the meaning set forth in Section 5.02(n).

"Knowledge" means, with respect to a party hereto, actual knowledge and the knowledge that each person would have reasonably obtained after making due and appropriate inquiry with respect to the particular matter in question or of any officer of that party with the title of not less than a vice president.

"Lien" means any charge, mortgage, pledge, security interest, restriction, claim, lien, or encumbrance of any kind.

"Material Adverse Effect" means any effect, circumstance, occurrence or change that (i) is material and adverse to the business, assets or deposit liabilities, properties, operations, results of operations, condition (financial or otherwise) or prospects of CB or the Bank, taken as a whole, or would materially impair the ability of CB or the Bank to perform its obligations under this Agreement or otherwise materially threaten or materially impede the consummation of the Merger and the other transactions contemplated by this Agreement provided, however, that Material Adverse Effect shall not be deemed to include the impact of any (a) change in law, rule or regulation or GAAP or interpretations thereof that applies to CB or the Bank, (b) changes in economic conditions affecting commercial banks generally (including, without limitation, any changes in interest rates), (c) acts of terrorism or war, (d) any modifications or changes to valuation policies and practices in connection with the Merger or restructuring charges taken in connection with the Merger, in each case in accordance with GAAP or RAP, as applicable, (e) changes resulting from fees and expenses (including, but not limited to legal, accounting and investment bankers' fees) incurred in connection with this Agreement or the transactions contemplated herein, including any fees or expenses relating to litigation that may be filed to contest the transactions contemplated herein, (f) the payment of any amounts due, or the provision of any benefits to, any officer or employee under employment contracts or employee benefit plans, severance agreements, consulting agreements or other arrangements disclosed in the Disclosure Schedules; or (g) actions or omissions of CB or the Bank taken in connection with the Merger; or which have been waived by Clear.

4

"Merger" has the meaning set forth in Section 2.01(a).

"Merger Consideration" means: the per share consideration to be paid to each CB shareholder in cash and/ or New Pubco common stock in accordance with each shareholders election to receive a combination of stock and cash ("Option A" as defined), or cash only ("Option B" as defined). For purposes of clarity and by way of example only, a sample calculation of the Merger Consideration and Per Share Merger Price based on CB and Bank financials, adjusted as agreed upon, as of the Calculation Date is attached as Schedule I to this Agreement.

"Merger Sub" means the subsidiary corporation to be formed by Clear and with which CB will merger as set forth in Section 2.01

"New Pubco" has the meaning set forth in the preamble to this Agreement.

"NOL Tax Benefit" means \$150,000, an arbitrary number agreed upon by the Parties

"Non-Election Shares" has the meaning set forth in Section 3.04.

"*OFAC*" has the meaning set forth in Section 5.02(x).

"Offering Memorandum" has the meaning set forth in Section 6.16.

Option A" means consideration for each share of CB stock consisting of three times the per share AEBV of CB common stock, plus a pro rata share of net Premises valuation and NOL Tax Benefit, paid out 1/3 cash and 2/3 New Pubco common stock (valued at \$10 per share). In the event the receipt of New Pubco stock pursuant to this Agreement does not, in the reasonable determination of Clear, qualify as part of a "reorganization" within the meaning of Code Section 368 (and does not otherwise qualify for tax deferral under the Code), Clear will make an additional payment at Closing to those shareholders electing Option A who provide in a reasonable time frame prior to the Closing any information reasonably requested or necessary to determine the amount of such payment, an amount equal to the U.S. federal and state income taxes they incur with respect to the receipt of New Pubco stock pursuant to this Agreement, that are in effect at the time of Closing per the U.S. Tax Code, which currently include up to a 20% Capital Gain rate and a 3.9% Medicare Tax rate and the appropriate rate for any applicable state income taxes.

"Option B" means an all cash payment for each share of CB common stock consisting of two times the per share AEBV of CB common stock, plus the pro rata per share amount of net Premises valuation and NOL Tax Benefit.

5

"Paying Agent" means a qualified entity mutually agreed upon by the parties hereto.

"Pension Plan" has the meaning set forth in Section 5.02(m)(ii).

"Per Share Merger Price" means the quotient of (a) the Merger Consideration; divided by (b) the total number of Shares outstanding immediately prior to the Effective Time.

"Person" means any individual, bank, corporation (including not-for-profit), joint-stock company, general or limited partnership, limited liability company, joint venture, estate, business trust, trust, association, organization, Governmental Authority or other entity of any kind or nature.

"Premises" means the existing building and land comprising the physical location of the Bank.

"Privacy and Security Requirements" means, to the extent directly applicable to CB and the Bank, (a) any provisions Laws that regulate the processing of Protected Data; (b) provisions of Contracts between CB or the Bank and any Person that directly apply to the processing of Protected Data; and (c) all written policies relating to the processing of Protected Data.

"Protected Data" means any personal information or data that CB or the Bank is required by Contract or Law to keep confidential.

"*QFTA*" has the meaning set forth in the preamble to this Agreement.

"RAP" means regulatory accounting.

"Regulatory Accounting" means the accounting treatment, rules, and reporting processes required by CB and the Banks regulators.

"Regulatory Authority" shall mean any federal or state Governmental Authority charged with the supervision or regulation of financial institutions or issuers of securities or engaged in the insurance of deposits (including, without limitation, the Wyoming Division of Banking, the Federal Reserve Board and the FDIC).

"Resale Registration Statement" means the Resale Registration Statement referenced in Section 3.04(h).

"Rights" means, with respect to any Person, securities or obligations convertible into or exercisable or exchangeable for, or giving any Person any right to subscribe for or acquire, or any options, calls or commitments relating to, or any stock appreciation right or other instrument the value of which is determined in whole or in part by reference to the market price or value of, shares of capital stock of such Person.

"Security Breach" means any (a) material unauthorized use, disclosure, acquisition of, or access to, Protected Data; or (b) any breach of information technology security safeguards that had a material impact on CB's or the Bank's operations.

6

"Share" and "Shares" has the meaning set forth in Section 3.02(a).

"Statement of Merger" has the meaning set forth in Section 2.02.

"Subsidiary" as to any Person, a corporation, limited liability company, partnership or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, limited liability company, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person.

"*Tax*" or "*Taxes*" means any and all federal, state, local or foreign taxes, charges, fees, levies, duties, tariffs, imposts, other assessments and other similar fees or similar charges, however denominated (together with any and all interest, penalties, additions to tax and additional amounts imposed with respect thereto), the liability for which is imposed by any government or taxing authority, by contractual agreement, as a result of being a member of any affiliated, consolidated, combined, unitary or similar group, as a successor to or transferee of another person, or otherwise including, without limitation: taxes on or with respect to income, franchises, windfall or other profits, gross receipts, license, property, sales, use, service, service use, capital stock, payroll, employment, social security, disability, severance, workers' compensation, employer health, unemployment compensation, or net worth; taxes or other charges in the nature of excise, withholding, estimated, severance, occupation, customs, duties, fees, ad valorem, property, environmental, stamp, transfer, value added, or gains taxes; license, registration, recording and documentation fees whether arising before, on or after the Effective Time.

"Tax Return" or "Tax Returns" means returns, declarations, reports, statements, elections, estimates, claims for refund, information returns or other documents (including any related or supporting schedules, exhibits, statements or information, any amendment to the foregoing, and any sales and use and resale certificates) filed or required to be filed in connection with the determination, assessment, payment, deposit, collection or reporting of any Taxes of any party or the administration of any laws, regulations or administrative requirements relating to any Taxes.

"Transaction Costs" means any and all costs, fees and expenses incurred by CB in connection with this Agreement and the transactions contemplated hereby, including but not limited to the fees and expenses of CB's attorneys, accountants and investment advisors.

"WBCA means the Wyoming Business Corporation Act, (WBCA 17-16-101 et seq)

7

ARTICLE II THE MERGER

2.01 The Merger.

(a) A series of related mergers will occur pursuant to the terms of this Agreement and the transactions described in the Business Combination Agreement. On the Closing Date, CB shall be merged with the Merger Sub pursuant to all applicable legal requirements, whereupon the separate existence of the Merger Sub shall cease, and CB shall be the surviving corporation (the "Surviving Corporation"). Immediately following on the Closing Date CB as the surviving corporation shall be merged with Clear, with Clear as "Surviving Corporation" and as a wholly owned subsidiary of New Pubco.

(b) *Continuity of Operation*. The parties will continue the operation of the Bank for not less than two (2) years following the Effective Time of the merger.

2.02 *Effective Time*. Subject to the satisfaction or waiver of the conditions set forth in Article VIII (other than those conditions that by their nature are to be satisfied at the consummation of the Merger, but subject to the fulfillment or waiver of those conditions), on the tenth (10th) Business Day of the month immediately following the month in which receipt of all necessary regulatory, corporate and other approvals have been obtained and the expiration of any mandatory waiting periods and satisfaction of the material conditions to closing hereunder have occurred and the parties have consummated the transactions in this Agreement (herein, the "Closing Date"), a Statement of Merger (the "Statement of Merger") shall be filed with the Secretaries of State of the States of Wyoming and Delaware in accordance with applicable law together with such certificates, documents or other instruments as may be required by law, and the Merger shall become effective upon such filing. The "Effective Time" of the Merger shall be the time of such filing or as set forth in such filing if other than the time of filing.

2.03 *Escrow Deposit.* Upon execution of this Agreement, Clear shall deposit into escrow with the Escrow Agent \$100,000 prorated among CB shareholders to be held as a good faith deposit, and, if not previously released, to serve as Holdback funds on Closing Date in accordance with ARTICLE VII. Clear shall cause said \$100,000 good faith deposit to be released to CB in the event that the proposed transaction fails to close within two (2) years from the date of this Agreement (provided regulatory applications are no longer pending); provided, however, no such payment will be required if Clear and New PubCo fail to enter into the transactions described in the RECITALS; no such payment will be required if any regulatory approval required pursuant to Article VIII of this Agreement has been denied; CB or the Bank have breached the terms of this Agreement; there has been a "material adverse change" in the condition of CB or the Bank; or if CB has failed to obtain the required approval of CB shareholders approving this Agreement and the Merger.

ARTICLE III CONSIDERATION; EXCHANGE PROCEDURES

3.01 *Calculation of Merger Consideration.* The calculation of adjusted book value for these purposes requires the Bank to make available to Clear at least fifteen (15) days prior to the Closing Date all reasonably requested information regarding the status of each loan contained in the loan portfolio of the Bank. Based on Clear's review, any adjustment to the Bank's allowance for loan or lease losses, which are allowable under all applicable regulatory and GAAP, will be made by Bank prior to Closing Date. At least five (5) days prior to the Closing Date, CB shall calculate its Adjusted Estimated Book Value, taking into consideration the above referenced adjustment to the allowance for loan and lease losses, the Merger Consideration and the Per Share Merger Price and shall provide Clear with a copy of such calculations together with such supporting documentation as may be reasonably requested by Clear (the "Calculation Statement"). The Calculation Statement and calculations hereon shall fairly present the financial position of CB and the Bank in accordance with GAAP and RAP, as applicable, and each of the Equity Capital, the Merger Consideration and the Per Share Merger Price calculated in

accordance with the terms of this Agreement. No fractional shares of New Pubco Common Stock shall be issued, any such fractions to be paid in cash based upon a \$10 per share value of New Pubco Common Stock.

8

3.02 *Effect on Capital Stock.* At the Effective Time, as a result of the Merger and without any action on the part of any holder of shares of CB Common Stock:

(a) *Outstanding CB Common Stock*. Each share of CB Common Stock issued and outstanding immediately prior to the Effective Time (each, a "Share" and, collectively, "Shares"), shall become and be converted into the right to receive the Per Share Merger Price per each shareholders election under Option A or Option B. At the Effective Time, all Shares shall no longer be outstanding and shall automatically be cancelled and retired and shall cease to exist, and each holder of a certificate that immediately prior to the Effective Time represented any Shares (a "Certificate") shall cease to have any rights with respect thereto, except the right to receive the Merger Consideration.

3.03 *Rights as Shareholders; Stock Transfers*. At the Effective Time, holders of CB Common Stock shall cease to be, and shall have no rights as, shareholders of CB other than to receive the Merger Consideration.

3.04 Election Procedures; Exchange Procedures.

(a) Election Procedures.

Holders of record of CB Common Stock have the right to submit an Election Form (defined below) specifying thea. number of shares of CB Common Stock, if any, as to which such holder desires to select Option A and the number of shares of CB Common Stock, if any, as to which such holder desires to select Option B.

An election form ("Election Form"), together with a Letter of Transmittal and Offering Memorandum shall be mailed no less than twenty (20) Business Days prior to the election deadline specified in the Election Form (the "Election Deadline") or on such earlier date the parties shall mutually agree (the "Mailing Date") to each holder of record of CB Common Stock as of five (5) Business Days prior to the Mailing Date. Holders of record of shares of CB Common

b. Stock who hold such shares as nominees, trustees or in other representative capacities may submit multiple Election Forms, provided that each such Election Form covers all the shares of CB Common Stock held by each representative for a particular beneficial owner. Any shares owned by a holder who has not, as of the Election Deadline, made an election by submission to the Paying Agent of an effective, properly completed Election Form shall be deemed "Non-Election Shares" and shall be deemed to have selected Option B (all cash).

9

An election shall have been properly made only if the Paying Agent shall have actually received a properly completed Election Form by the Election Deadline. Any Election Form may be revoked or changed by the Person submitting such Election Form to the Paying Agent by written notice to the Paying Agent only if such notice of revocation or change is actually received by the Paying Agent at or prior to the Election Deadline. Shares of CB Common Stock held by holders who acquired such shares subsequent to the Election Deadline will be designated Non-Election Shares. In addition, if a Holder of CB Common Stock either (1) does not submit a properly completed Election Form in a timely fashion or

c. (2) revokes its Election Form prior to the Election Deadline and fails to file a new properly completed Election Form before the deadline, such shares shall be designated Non-Election Shares. Subject to the terms of this Agreement and of the Election Form, the Paying Agent shall have discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in the Election Forms, and any good faith decisions of the Paying Agent regarding such matters shall be binding and conclusive. Neither CB, Clear, New Pubco nor the Paying Agent shall be under any obligation to notify any Person of any defect in an Election Form.

(b) *Paying Agent*. At or prior to the Effective Time, New Pubco shall deposit, or shall cause to be deposited, with the Paying Agent, for the benefit of the holders of Shares the aggregate amount of Merger Consideration payable in respect of Shares pursuant to Section 3.01 upon surrender of Certificates (the "Exchange Fund").

(c) Exchange of Certificates. As soon as practicable after the Effective Time, but in no event later than one Business Day thereafter, Clear shall cause the Paying Agent to mail to each former holder of record of Shares of CB (i) a letter of transmittal specifying that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates (or affidavits of loss in lieu thereof and, if required, indemnities in accordance with Section 3.04(f) to the Paying Agent, such letter of transmittal to be in such form and have such other provisions as Clear and CB may reasonably agree, and (ii) instructions for use in effecting the surrender of the Certificates in exchange for the Merger Consideration. Upon surrender of a Certificate for cancellation to the Paying Agent (or an affidavit of loss in lieu thereof and, if required, an indemnity in accordance with Section 3.04(f)) together with such letter of transmittal, duly executed, the holder of such Certificate shall be entitled to receive in exchange therefor the Merger Consideration (after giving effect to any required Tax deductions and withholdings in accordance with Section 3.04(g)), and the Certificates so surrendered shall forthwith be cancelled. Clear shall direct the Paying Agent to promptly deliver the Merger Consideration to each former holder of record of Shares who properly complies with the transmittal instructions described herein, and in no case shall such payment be made later than two (2) Business Days after receipt by the Paying Agent of such instructions. No interest will be paid or accrued on any amount payable upon due surrender of the Certificates. In the event of a transfer of ownership of Shares that is not registered in the transfer records of CB, the proper amount of the Merger Consideration may be paid in exchange therefor to a Person other than the Person in whose name the Certificate so delivered is registered if the Certificate formerly representing such Shares is presented to the Paying Agent, accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable Taxes have been paid.

10

(d) *No Liability*. Notwithstanding the foregoing of CB, neither the Paying Agent nor any party hereto shall be liable to any former holder of Shares for any amount properly delivered to a Governmental Authority pursuant to applicable abandoned property, escheat or similar laws.

(e) Unclaimed Funds. Any portion of the Exchange Fund that remains unclaimed by former holders of Shares of CB for one hundred eighty (180) days after the Effective Time shall be transferred from Paying Agent to New Pubco. Any former holders of Shares of CB who have not theretofore complied with this Article III shall thereafter look only to Clear for payment of the Merger Consideration deliverable in respect of their Shares upon due surrender of their Certificates (or affidavits of loss in lieu thereof) pursuant to this Article III, in each case, without any interest thereon.

(f) Lost, Stolen or Destroyed Certificates. In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by Clear or the Paying Agent, the posting by such Person of a bond in customary amount as indemnity against any claim that may be made against it with respect to such Certificate, the Paying Agent will pay in exchange for such lost, stolen or destroyed Certificate the Merger Consideration deliverable in respect of such Shares represented by such Certificate.

Withholding Rights. The Paying Agent or Clear shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of Shares such amounts as Clear or the Paying Agent is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of state, local or foreign tax law. To the extent

(g) while respect to the making of such payment under the code, of any provision of state, rocal of foreign and new rocale entering that amounts are so deducted and withheld by the Paying Agent or Clear, such amounts shall be treated for all purposes of this Agreement as having been paid to the holder of Shares in respect of which such deduction and withholding was made by the Paying Agent or Clear.

Resale Registration Statement. As soon as practicable (and in any event within 30 Calendar days following the Closing date) New Pubco shall file a registration statement with the SEC providing for the resale by CB shareholders of shares received as part of the Merger Consideration (the "Resale Registration Statement"). New Pubco shall use commercially reasonable efforts

(h) New Pubeo shall like a registration statement with the SLC providing for the restrict by CD shall only of shall be received a part of the Merger Consideration (the "Resale Registration Statement"). New Pubeo shall use commercially reasonable effort to cause such Resale Registration Statement to become effective within 60 Calendar days from Closing Date.

ARTICLE IV ACTIONS PENDING ACQUISITION

4.01 *Forbearances of CB and Bank*. From the date hereof until the Effective Time, except as expressly contemplated by this Agreement and/or disclosed on CB and the Bank's Disclosure Schedule, without the prior written consent of Clear, which consent shall not be unreasonably withheld, CB and Bank will not:

(a) Ordinary Course. (i) Conduct the business of CB and Bank other than in the ordinary and usual course or fail to use reasonable efforts to preserve intact their business organizations and assets and maintain its rights, franchises and existing relations with customers, suppliers, employees and business associates, or voluntarily take any action which, at the time taken, has or is reasonably likely to have an adverse effect upon CB and Bank's ability to perform any of its material obligations under this Agreement, or (ii) enter into any new material line of business or change its lending, investment, underwriting, risk, asset liability management or other banking and operating policies, except as required by applicable law, regulation or policies imposed by any Governmental Authority.

(b) *Capital Stock.* Other than pursuant to Rights, if any, as set forth on the Disclosure Schedule and outstanding on the date hereof, (i) issue, sell or otherwise permit to become outstanding, or authorize the creation of, any additional shares of CB or Bank Stock or any Rights, (ii) enter into any agreement with respect to the foregoing, or (iii) permit any additional shares of CB or Bank Stock to become subject to new grants of employee or director stock options, other Rights or similar stock-based employee rights.

(c) *Dividends, Etc.* (i) Make, declare, pay or set aside for payment any dividend, other than quarterly cash dividends on CB or Bank Common Stock in amounts consistent with past practice, or (ii) directly or indirectly adjust, split, combine, redeem, reclassify, purchase or otherwise acquire, any shares of its capital stock.

(d) Compensation; Employment Agreements, Etc. Enter into, amend, or renew any employment, consulting, severance or similar agreements or arrangements with any director, officer or employee of CB or Bank, or grant any salary or wage increase or increase any employee benefit (including incentive or bonus payments), other than officer salary increases that the board of directors deem to be in the best interests of the Bank, except (i) for normal individual increases in compensation to non-officer employees in the ordinary course of business consistent with past practice which in no case shall exceed a reasonable cost of living increase, (ii) for other changes that are required by applicable law, and (iii) to satisfy contractual obligations existing as of the date hereof set forth in the Section 5.02(k) or (m) of CB Bank's Disclosure Schedule.

(e) *Benefit Plans*. (i) Enter into, establish, adopt, terminate, or amend any (a) collective bargaining or agreement with a union, works council or labor organization covering any employee or (b) Compensation Plan (except (i) as may be required by applicable law, (ii) to satisfy contractual obligations existing as of the date hereof which are disclosed in the Section 5.02(k) or (m) of CB and Bank's Disclosure Schedule, or (iii) within the regular annual renewal of insurance contracts, or (ii) take any action or make any promise to accelerate the payment, vesting or exercisability of stock options, restricted stock, severance, or other compensation or benefits for any employee, director, officer, or other individual service provider of CB or Bank.

12

(f) *Dispositions*. Sell, transfer, mortgage, encumber or otherwise dispose of or discontinue any of its assets, deposits, business or properties except in the ordinary course of business.

(g) Acquisitions. Acquire (other than by way of foreclosures or acquisitions of control in a bona fide fiduciary capacity or in satisfaction of debts previously contracted in good faith, in each case in the ordinary and usual course of business consistent with past practice) all or any portion of, the assets, business, deposits or properties of any other Person (other than purchases of loans and loan participations made in the ordinary and usual course of business consistent with past practice).

(h) Governing Documents. Amend the CB or Bank Articles, CB or Bank Bylaws (or similar governing documents).

(i) Accounting Methods; Maintenance of Books and Records. Implement or adopt any change in the CB or Bank's book or tax accounting principles, practices or methods, other than as may be required by GAAP or RAP, and as concurred in by CB or the Bank's independent public accountants; or fail to maintain and prepare CB and the Bank's books and records other than in accordance with GAAP or RAP, as applicable, consistent with past practice.

(j) Contracts. Except in the ordinary course of business consistent with past practice, enter into or terminate any material Contract or amend or modify in any material respect any of its existing material Contracts.

(k) *Claims*. Settle any claim, action or proceeding that requires the payment by CB or the Bank of an amount in excess of \$ 10,000 for an individual claim, and \$ 50,000 for claims in the aggregate.

(1) *Adverse Actions*. Take any action that is intended or is reasonably likely to result in (i) any of its representations and warranties set forth in this Agreement being or becoming untrue in any material respect at any time at or prior to the Effective Time, (ii) any of the conditions to the Merger set forth in Article VIII not being satisfied, or (iii) a material violation of any provision of this Agreement except, in each case, as may be required by applicable law or regulation.

(m) *Risk Management*. Except pursuant to applicable law or regulation or by order of any relevant Regulatory Authority, (i) implement or adopt any material change in its interest rate risk management and other risk management policies, procedures or practices; (ii) fail to follow its existing policies or practices with respect to managing its exposure to interest rate and other risk; or (iii) fail to use commercially reasonable means to avoid any material increase in its aggregate exposure to interest rate risk and other risk.

(n) *Indebtedness*. Incur any indebtedness for borrowed money other than in the ordinary course of business consistent with past practice.

(o) *Capital Expenditures*. Other than expenditures for updates to the Bank's core operating systems, make any capital expenditure or commitments with respect thereto in an amount in excess of \$ 10,000 for any item or project, or \$ 30,000 in the aggregate for any related items or projects, except as have been previously committed to prior to the date hereof. This limitation excludes any update for Bank's core operating system.

(p) *Taxes*. Commence or settle any litigation or proceeding with respect to any liability for Taxes, make or change any material express or deemed Tax election, file any amended Tax Return, take any action which is reasonably likely to have a material adverse effect on any Tax position of CB or the Bank or, after the Merger, which is reasonably likely to have a material adverse effect on any Tax position of Clear, QFTA or New Pubco, change any of its methods of reporting income or deductions for Tax purposes or take any other action with respect to Taxes that is outside the ordinary and usual course of business or inconsistent with past practices.

(q) Conflict of Interest Transactions. Engage or agree to engage in any "covered transaction" within the meaning of Sections 23A or 23B of the Federal Reserve Act (without regard to the applicability of any exemptions contained in Section 23A) unless the Bank has complied with Sections 23A and 23B of the Federal Reserve Act.

(r) *Loan Loss Allowances*. Fail to maintain an allowance for possible loan and lease losses which is adequate in all material respects under the requirements of GAAP and RAP, as applicable, to provide for possible losses, net of recoveries relating to loans previously charged off, on loans outstanding (including accrued interest receivable), or fail to charge off any loans or leases that would be deemed uncollectible in accordance with the policies of the Bank.

(s) Commitments. Agree or commit to do any of the foregoing.

4.02 Forbearances of Clear. From the date hereof until the Effective Time, except as expressly contemplated by this Agreement, without the prior written consent of CB (which consent will not be unreasonably withheld), Clear will not take or omit to take, or agree or commit to take or omit to take, any action that would result in (i) any of Clear's representations and warranties set forth in this Agreement being or becoming untrue in any material respect at any time at or prior to the Effective Time, (ii) any of the conditions to the Merger set forth in Article VIII not being satisfied or (iii) a material violation of any provision of this Agreement, except as may be required by applicable law or regulation.

ARTICLE V REPRESENTATIONS AND WARRANTIES

5.01 *Disclosure Schedules*. On or prior to the date hereof, (or at such later date as agreed to by the parties hereto) CB will have delivered to Clear a schedule (the "Disclosure Schedule") setting forth, among other things, items, the disclosure of which are necessary or appropriate either in response to an express disclosure requirement contained in a provision hereof or as an exception to one or more representations or warranties contained in Section 5.02. An exception or matter disclosed with respect to one representation or warranty shall also be deemed disclosed with respect to each other representation or warranty to the extent such disclosure of such item would be reasonably understood to be applicable to such other schedule or representation. The inclusion of any item in the Disclosure Statement shall not be deemed an admission that such item is a material fact, event or circumstance or that such item has had or would be reasonably likely to have a Material Adverse Effect.

5.02 Representations and Warranties of CB and Bank. Subject to Section 5.01, CB and Bank hereby represent and warrant to Clear:

(a) Organization, Standing and Authority. CB is a Wyoming corporation duly organized, validly existing and in good standing under the laws of the State of Wyoming. Bank is a Wyoming state banking corporation duly organized, validly existing and in good standing under the laws of the State of Wyoming.

(b) Capital Structures; Organizational Documents and Records.

(i) The authorized capital stock of the Bank consists of 10,000,000 shares of Common Stock with no par value of which 100% of the outstanding shares are owned by CB as of the date of this Agreement. The authorized capital stock of CB consists of an unlimited number of shares of CB common stock, no par value, of which 449,043 shares are outstanding as of the date of this Agreement. The outstanding shares of CB and Bank Common Stock have been duly authorized, are validly issued and outstanding, fully paid and nonassessable, and are not subject to any preemptive rights (and were not issued in violation of any preemptive rights). Except as set forth on Section 5.02(b) of the Disclosure Schedules, as of the date hereof, (A) there were no shares of CB or Bank Common Stock authorized and reserved for issuance, (B) Neither CB nor Bank have any Rights issued or outstanding with respect to Common Stock, and (C) Neither CB nor Bank have any commitment to authorize, issue or sell any Common Stock or Rights. All shares of Common Stock were issued in compliance with all applicable laws, including federal and state securities laws.

(ii) The Bank and CB have furnished to Clear copies of their articles of incorporation and by-laws , in each case as amended to the date hereof, together with such other documents, including the charter of the bank, as reasonably requested by Clear relating to the authority of each to conduct its business. All such copies are complete and correct. The stock registers and minute books of CB and the Bank are each complete and correct in all material respects.

14

(c) *Subsidiaries*. Bank does not have any Subsidiaries or own beneficially, directly or indirectly, any equity securities or similar interests of any Person, or any interest in a partnership or joint venture of any kind.

(d) *Corporate Power*. CB and the Bank have full corporate power and authority to carry on business as it is now being conducted and to own all properties and assets. CB and Bank have the corporate power and authority to execute, deliver and perform its obligations under this Agreement.

(e) *Corporate Authority*. Subject to receipt of the requisite adoption of this Agreement by the holders of the outstanding shares of CB and Bank Common Stock entitled to vote thereon, this Agreement and the transactions contemplated hereby have been authorized by all necessary corporate action of CB and Bank prior to the date hereof. This Agreement is a valid and legally binding obligation of CB and Bank, enforceable in accordance with its terms (except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and similar laws of general applicability relating to or affecting creditors' rights or by general equity principles).

(f) Regulatory Filings; No Defaults.

(i) No consents or approvals of, or filings or registrations with, any Governmental Authority or with any third party are required to be made or obtained by CB and Bank or any of its Subsidiaries in connection with the execution, delivery or performance by CB and Bank of this Agreement or to consummate the Merger except for (A) filings of applications or notices with, or seeking approvals and waivers from, as applicable, federal and state banking authorities, including the Federal Reserve Board and the

Wyoming Division of Banking, (B) filings with state and federal securities authorities, (C) the filing of the Statement of Merger with the Wyoming Secretary of State pursuant to the WBCA, (D) the approval of the Merger by the holders of two-thirds (2/3) of the outstanding shares of CB Common Stock entitled to vote thereon, and (E) the third party consents set forth on the Disclosure Schedule under Section 5.02. As of the date hereof, CB and Bank are not aware of any reason why the approvals set forth in Section 8.01(b) will not be received without the imposition of a condition, restriction or requirement of the type described in Section 8.01(b).

(ii) Subject to receipt of the regulatory and shareholder approvals and third party consents referred to above and the expiration of certain regulatory waiting periods, and required filings under federal and state securities laws, to our Knowledge the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby do not and will not (A) constitute or result in a breach or violation of, or a default under, the acceleration of any obligations or the creation of a Lien on the assets of CB or the Bank (with or without notice, lapse of time, or both) pursuant to, any agreement, lease, license, contract, insurance policy, note, mortgage, indenture, instrument, arrangement or other obligation (each, a "Contract" and, collectively, "Contracts") binding upon CB or the Bank or to which they or any of their properties are subject or bound or any law or governmental or non-governmental permit or license to which CB or the Bank is subject, (B) constitute a breach or violation of, or a default under, CB or Bank Articles or Bylaws, or (C) require any consent or approval under any such law, rule, regulation, judgment, decree, order, governmental permit or license, agreement, indenture or instrument.

15

(g) Financial Reports; Undisclosed Liabilities.

(i) CB and Bank Call Reports and other periodic filings with the Federal Reserve of financial information (the "Financial Statements"), fairly present in all material respects the financial position of CB and the Bank as of filing dates. The books and records of CB and the Bank have been, and are being, maintained in accordance with GAAP and RAP, as applicable.

(ii) CB and the Bank has or will timely file all reports, registrations and statements, together with any amendments required to be made with respect thereto, that they are required to file with (A) the Federal Reserve Board and the Wyoming Division of Banking, as the case may be, and (B) any other Regulatory Authority (collectively, the "Regulatory Filings"), and all other material reports, registrations and statements required to be filed by it since December 31, 2021, including, without limitation, any report, registration or statement required to be filed pursuant to the laws of the United States or the State of Wyoming and the rules and regulations of the Federal Reserve Board, the Wyoming Division of Banking or any other Regulatory Authority, and has paid all fees and assessments due and payable in connection therewith. As of their respective dates, such reports, registrations and statements complied in all material respects with all the laws, rules and regulations of the applicable Regulatory Authority with which they were filed.

(iii) To their knowledge, except as set forth in Section 5.02 of the Disclosure Schedule, since December 31, 2020, CB and the Bank have not incurred any obligations or liabilities of any kind or nature (whether or not accrued, contingent or otherwise and whether or not required to be disclosed, including those related to environmental and occupational safety and health matters) other than in the ordinary and usual course of business consistent with past practice.

(iv) Since December 31, 2020, (A) CB and the Bank have conducted business only in, and have not engaged in any material transaction other than according to, the ordinary and usual course of such business consistent with past practice and (B) no event has occurred or circumstance arisen that, individually or taken together with all other facts, circumstances and events (described in any paragraph of this Section 5.02 or otherwise) has had or could be reasonably likely to have a Material Adverse Effect or prevent, materially delay or materially impair the ability of CB and the Bank to consummate the transactions contemplated by this Agreement.

(v) Since December 31, 2020, there has not been (A) any material damage, destruction or other casualty loss with respect to any material asset or property owned, leased or otherwise used by CB or the Bank, whether or not covered by insurance, (B) any declaration, setting aside or payment of any dividend or other distribution in cash, stock or property in respect of the capital stock of CB or the Bank, or (C) any change by CB or the Bank in accounting principles, practices, procedures or methods.

(h) *Litigation*. Except as disclosed in Section 5.02(h) of CB and the Bank's Disclosure Schedules, there is no suit, action, investigation, audit or proceeding (whether judicial, arbitral, administrative or other) pending or, to CB and the Bank's knowledge, threatened against or affecting CB or the Bank, nor is there any judgment, decree, injunction, rule or order of any Governmental Authority or arbitration outstanding against CB or the Bank. Neither CB nor the Bank have pending or threatened any legal proceeding against any third parties, other than collection, foreclosure or other similar actions in the ordinary course of business.

(i) Regulatory Matters.

(i) Neither CB nor the Bank or any of its properties is a party to or is subject to any order, decree, agreement, memorandum of understanding or similar arrangement with, or a commitment letter or similar submission to, or extraordinary supervisory letter from, any Regulatory Authority charged with the supervision or regulation of financial institutions and their subsidiaries (including their holding companies) or issuers of securities.

(ii) Neither CB nor the Bank has been advised by any Regulatory Authority that such Regulatory Authority is contemplating issuing or requesting (or is considering the appropriateness of issuing or requesting) any such order, decree, agreement, memorandum of understanding, commitment letter, supervisory letter or similar submission nor to its knowledge has any Regulatory Authority commenced an investigation in connection therewith.

(j) Compliance with Laws. Except as set forth in Section 5.02(j) of the Disclosure Schedule, each of CB and the Bank:

(i) is in compliance in all material respects with all applicable federal, state, local and foreign statutes, laws, regulations, ordinances, rules, judgments, orders or decrees applicable thereto or to the employees conducting such businesses, including, without limitation, the Equal Credit Opportunity Act, the Fair Housing Act, the Community Reinvestment Act (which includes a CRA Rating of "satisfactory" or better), the Home Mortgage Disclosure Act and all other applicable fair lending laws and other laws relating to discriminatory business practices;

(ii) has all permits, licenses, authorizations, orders and approvals of, and has made all filings, applications and registrations with, all Governmental Authorities that are required in order to permit them to own or lease their properties and to conduct their businesses as presently conducted; all such permits, licenses, certificates of authority, orders and approvals are in full force and effect and, to CB and the Bank's knowledge, no suspension or cancellation of any of them is threatened; and

(iii) has not received, since December 31, 2020, any notification or communication from any Governmental Authority or in a written communication from any other third party (A) asserting that CB or the Bank is not in compliance with any statute, regulation, or ordinance or other law or (B) threatening to revoke any license, franchise, permit, or governmental authorization (nor, to CB and the Bank's knowledge, do any grounds for any of the foregoing exist).

17

(k) Material Contracts; Defaults.

(i) Except as set forth in Section 5.02(k)(i) of the Disclosure Schedule (each a "Material Contract"), CB and the Bank are not a party to, bound by or subject to any Contract (whether written or oral) (i) that would be a "material contract" within the meaning of Item 601(b)(10) of the SEC's Regulation S-K; (ii) that purports to limit in any material respect either the type of business in which CB or the Bank (or, after giving effect to the Merger, New Pubco) may engage or the manner or locations in which any of them may so engage in any business; (iii) that constitute consulting arrangements and contracts for professional, advisory, and other services, including contracts under which the Bank performs services for others; (iv) that is a lease of real property; (v) that is a lease for equipment and computer hardware; (vi) that is a data processing agreement; (vii) that is a contract that involves performance of services or delivery of goods or materials by Bank that was not entered into in the ordinary course of business; or (viii) is a contract not referred to elsewhere in this section that materially affects the business or financial condition of the Bank, provided however, other than core system update contracts, the types of Material Contracts set forth in subsections (iii) through (viii) of this sentence are limited to and include only such contracts that could be reasonably expected to involve payments to or by the Bank of more than \$25,000 in any twelve-month period. True and correct copies of the Material Contracts have been delivered to the Clear. Each of the Material Contracts is a legal, valid and binding obligation of, and enforceable against CB or the Bank, and to the knowledge of CB and the Bank, the other party thereto; and is in full force and effect on the date hereof. Neither CB nor the Bank is in default under any Material Contract, and there has not occurred

any event that, with the lapse of time or the giving of notice or both, would constitute such a default. No power of attorney or similar authorization given directly or indirectly by CB or the Bank is currently outstanding.

(ii) Section 5.02(k)(ii) of the Disclosure Schedule sets forth a true and complete list of (x) all Contracts pursuant to which consents or waivers are or may be required and (y) all notices which are or may be required to be given, in each case, prior to the performance by CB and the Bank of this Agreement and the consummation of the Merger and the other transactions contemplated hereby.

(1) *No Brokers*. No action has been taken by CB or the Bank that would give rise to any valid claim against any party hereto for a brokerage commission, finder's fee or other like payment with respect to the transactions contemplated by this Agreement.

(m) Employee Benefit Plans and Labor Matters.

(i) Section 5.02(m) of the Bank's Disclosure Schedule contains a list of all "employee benefit plans" (within the meaning of Section 3(3) of ERISA), stock purchase and other equity-based plans, deferred compensation arrangements, and other employment, compensation, individual independent contractor or consultant, termination, bonus, commission, severance, change in control, transaction or retention bonus, or other similar plans, programs, policies, contracts, or arrangements (whether written or unwritten) maintained or contributed to, or required to be contributed to by CB or Bank for the benefit of any current or former employee, director, officer or individual independent contractor or consultant thereof or under which CB or Bank may have any Liability ("Compensation Plans"). All Compensation Plans have been operated in substantial compliance with the terms of such Compensation Plan, ERISA and other applicable Law.

18

(ii) Neither CB, Bank, nor any of their ERISA Affiliates has ever sponsored, maintained, administered, or contributed to, or has had or could have any Liability with respect to, (a) any plan subject to Title IV of ERISA, Section 302 of ERISA or Section 412 of the Code, (b) any "multiemployer plan" (as defined in Section 3(37) of ERISA), (c) any "multiple employer plan" (as defined in Section 413(c) of the Code), or (d) any "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA). Except as may be required by the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, or any similar state Laws (and for which the beneficiary pays the full cost of coverage), no Compensation Plan provides medical, health or other welfare benefits or coverage for any current or former employees following retirement or other termination of service.

(iii) CB and Bank are not party to or bound by any collective bargaining agreement or other Contract with a union, works council or labor organization with respect to any employees. As of the date hereof, there are no pending or, to the Knowledge of CB and Bank, threatened, arbitrations, grievances, organizing campaigns, strikes, or other labor disputes or disruptions against or affecting the Company and no such event has occurred within the last three years. CB and Bank are in compliance, in all material respects, with all applicable Laws governing employment and labor.

(iv) There are no claims, proceedings, audits, investigations, or other similar matters pending or, to the Knowledge of CB and Bank, threatened, against, relating to, involving or otherwise affecting CB, Bank or any current or former employee, officer, director, applicant for employment, temporary or leased employee, or independent contractor or consultant thereof with respect to the Compensation Plans (other than routine claims for benefits) or any employment, labor, or employee benefit matters. No claims or allegations have been made against CB, Bank or any current or former individual service provider thereof for discrimination, sexual or other harassment, sexual misconduct or retaliation within the last three years, nor, to Knowledge of CB and Bank, are any such claims threatened or pending nor is there any reasonable basis for such a claim

(n) *Tax Matters*. (i) All Tax Returns that are required to be filed by or with respect to CB and the Bank have been duly filed, (ii) all Taxes shown to be due on the Tax Returns referred to in clause (i) have been paid in full, (iii) except as set forth in Section 5.02(n) of CB and the Bank's Disclosure Schedule, the Tax Returns referred to in clause (i) have been examined by the IRS or the appropriate state, local or foreign taxing authority or the period for assessment of the Taxes in respect of which such Tax Returns were required to be filed has expired, (iv) all deficiencies asserted or assessments made as a result of such examinations have been paid in full, (v) no issues that have been raised by the relevant taxing authority in connection with the examination of any of the Tax Returns referred to in clause (i) are currently pending, (vi) there are no pending or, to CB and the Bank's knowledge, threatened suits, proceedings, actions or claims and, to CB and the Bank's knowledge, no pending or threatened audits, in each case, for or relating to any tax liability with respect to CB or the Bank, and (vii) no waivers of statutes of limitation have been given by or requested with respect to any Taxes of

CB or the Bank. CB and the Bank have made or will make available to Clear true and correct copies of the United States federal income Tax Returns filed by it or its Affiliates for each of the three most recent fiscal years ended on or before December 31, 2021. Neither CB nor the Bank has any liability with respect to income, franchise or similar Taxes that accrued on or before the end of the period ended December 31, 2021 in excess of the amounts accrued with respect thereto that are reflected in the Financial Statements.

(o) Risk Management Instruments; Investments.

(i) All material interest rate swaps, caps, floors, option agreements, futures and forward contracts and other similar risk management arrangements, whether entered into for Bank's own account, or for the account of one or more of Bank or its customers (all of which are listed on Bank's Disclosure Schedule), were entered into (i) in accordance with prudent business practices and all applicable laws, rules, regulations and regulatory policies and (ii) with counterparties believed to be financially responsible at the time; and each of them constitutes the valid and legally binding obligation of Bank, enforceable in accordance with its terms (except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and similar laws of general applicability relating to or affecting creditors' rights or by general equity principles), and is in full force and effect. Neither Bank, nor to Bank's knowledge any other party thereto, is in breach of any of its obligations under any such agreement or arrangement.

(ii) Set forth on Section 5.02(o) of the Disclosure Schedule is a complete and correct list of all investment and debt securities, mortgage- backed and related securities, marketable equity securities and securities purchased under agreements to resell that are owned by the Bank, other than in a fiduciary or agency capacity (the "Bank Investment Securities"). Bank has good and marketable title to all Bank Investment Securities held by it, free and clear of all encumbrances, and except to the extent such Bank Investment Securities are pledged in the ordinary course of business consistent with prudent banking practices to secure obligations of the Bank. The Bank Investment Securities are valued on the books of the Bank in accordance with GAAP, as may be modified by RAP. Except as set forth on Section 5.02(o) of the Disclosure Schedule, and as may be imposed by applicable securities laws and documents governing the terms of such securities, none of the Bank Investment Securities is subject to any restriction, whether contractual or statutory, that materially impairs the ability of the Bank to freely dispose of such investment at any time. With respect to all material repurchase agreements to which the Bank is a party, the Bank has a valid, perfected first lien or security interest in the securities or other collateral securing each such repurchase agreement equals or exceeds the amount of debt secured by such collateral under such agreement.

20

(p) *Books and Records*. The books and records of CB and the Bank have been fully, properly and accurately maintained in all material respects, have been maintained in accordance with sound business practices in the banking industry, and there are no material inaccuracies or discrepancies of any kind contained or reflected therein and they fairly reflect the substance of events and transactions included therein.

(q) *Ins*urance. The Disclosure Schedule sets forth all of the insurance policies, binders, or bonds maintained by CB and the Bank. CB and the Bank are insured with reputable insurers against such risks and in such amounts as the management of CB and the Bank reasonably have determined to be prudent in accordance with industry practices. All such insurance policies are in full force and effect; Neither CB nor Bank is in material default thereunder; and all claims thereunder have been filed in due and timely fashion.

(r) *Transactions With Affiliates*. Except as set forth in Section 5.02(r) of the Disclosure Schedule, neither CB nor the Bank has any transactions with Affiliates within the meaning of Sections 23A and 23B of the Federal Reserve Act, as implemented by Regulation W of the Federal Reserve Board.

(s) *Properties*. Section 5.02(s) of the Disclosure Schedule sets forth all real property owned or leased by the Bank or in which the Bank has an interest (other than as a mortgagee). The Bank owns, or has a valid right to use or a leasehold interest in, all real property used by it in the conduct of its business as such business is presently conducted. Bank has good and marketable title, free and clear of all liens, encumbrances, charges, defaults or equitable interests to all of the properties and assets, real and personal, reflected on the Bank Financial Statements as being owned by Bank as of December 31, 2021 or acquired after such date, except (i) statutory

liens for amounts not yet due and payable, (ii) pledges to secure deposits and other liens incurred in the ordinary course of banking business, (iii) such imperfections of title, easements, encumbrances, liens, charges, defaults or equitable interests, if any, as do not affect the use of properties or assets subject thereto or affected thereby or otherwise materially impair business operations at such properties, (iv) dispositions and encumbrances in the ordinary course of business, and (v) liens on properties acquired in foreclosure or on account of debts previously contracted. All leases pursuant to which Bank, as lessee, leases real or personal property (except for leases that have expired by their terms or that Bank has agreed to terminate since the date hereof) are valid without default thereunder by the lessee or, to Bank's knowledge, the lessor. The condition of the properties and assets of the Bank is sufficient, in all material respects, for the operation of the business of the Bank as currently conducted by the Bank.

(t) Loans. Each loan reflected as an asset in the Bank Financial Statements and each balance sheet date subsequent thereto (i) is evidenced by notes, agreements or other evidences of indebtedness that are true and genuine, (ii) to the extent secured, has been secured by valid liens and security interests that have been perfected, and (iii) is the legal, valid and binding obligation of the obligor named therein, enforceable in accordance with its terms, subject to bankruptcy, insolvency, fraudulent conveyance and other laws of general applicability relating to or affecting creditors' rights and to general equity principles. The reserves for loan and lease losses shown on each of the balance sheets contained in the Bank Financial Statements are adequate in the judgment of management and consistent with the standards of the Federal Reserve and under GAAP to provide for incurred and probable losses, net of recoveries relating to loans and leases outstanding (including accrued interest receivable) as of the applicable date of such balance sheet. The aggregate loan balances of the Bank as of December 31, 2021, in excess of such reserves as shown on the balance sheet as of that date included in the Bank Financial Statements are, to the knowledge of CB and the Bank, collectible in accordance with their terms. All loans and extensions of credit that have been made by the Bank that are subject either to Section 22(b) of the Federal Reserve Act, as amended, or to Part 349 of the rules and regulations of the FDIC, comply therewith.

21

(u) Allowance for Loan Losses. The allowance for loan losses reflected on the Bank Financial Statements, as of their respective dates, is adequate in all material respects under the requirements of RAP to provide for reasonably incurred losses on outstanding loans.

(v) *Deposit Insurance*. The deposits of the Bank are insured by the FDIC in accordance with the Federal Deposit Insurance Act of 1933, as amended ("FDIA"), and Bank has paid all assessments and filed all reports required by the FDIA.

(w) Annual Disclosure Statement. Bank is in compliance with Part 350 of the rules and regulations promulgated by the FDIC concerning disclosure requirements, including the preparation of an annual disclosure statement, and the signature and attestation requirements provided and to be provided pursuant to such Part are accurate.

(x) Bank Secrecy Act, Anti-Money Laundering, Trade Compliance and Customer Information.

(i) Neither CB, the Bank nor any of their representatives acting on their behalf, is aware of, based on its internal compliance procedures, and has not been advised of, and has no reason to believe that any facts or circumstances exist, which would cause it to be deemed (i) to be operating in violation in any material respect of the Bank Secrecy Act, the Patriot Act, any order issued with respect to anti-money laundering by the U.S. Department of the Treasury's Office of Foreign Assets Control ("OFAC"), or any other applicable anti-money laundering statute (including but not limited to the USA PATRIOT Act of 2011, the United States Bank Secrecy Act of 1970, the United States Money Laundering Control Act of 1986, and the International Money Laundering Abatement and Anti-Terrorist Financing Act of 2001, the laws and executive orders administered by state agencies, and any regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority; or (ii) not to be in satisfactory compliance in any material respect with the applicable privacy and customer information requirements contained in any federal and state privacy laws and regulations, including, without limitation, in Title V of the Gramm-Leach-Bliley Act of 1999 and the regulations promulgated thereunder, as well as the provisions of the information security program adopted by Bank pursuant to 12 C.F.R Part 40. No suit, action, investigation, audit or proceeding (whether judicial, arbitral, administrative or other) involving CB or the Bank with respect to any of the foregoing is pending or, to the knowledge of CB or the Bank, threatened.

(ii) Neither CB, the Bank nor any of their representatives acting on their behalf, is aware of any facts or circumstances that would cause it to believe that any non-public customer information has been disclosed to or accessed by an unauthorized third party in a manner that would cause it or any of its Subsidiaries to undertake any material remedial action. The Bank Board has adopted and implemented an anti-money laundering program that contains adequate and appropriate customer identification verification procedures that comply with Section 326 of the Patriot Act and such anti-money laundering program meets the requirements in all material respects of Section 352 of the Patriot Act and the regulations thereunder, and it (or such other of its Subsidiaries) has complied in all material respects with any requirements to file reports and other necessary documents as required by the Patriot Act and the regulations thereunder. The Bank's internal compliance procedures comply with all applicable laws and regulations. Neither CB nor the Bank have received any written or oral communication from any Governmental Authority that alleges that CB, the Bank, or any of their agents are in violation of, or have, or may have any liability under, any international trade laws or Anti-Money Laundering Laws. Further, neither CB nor the Bank has, during the past five (5) years, made any voluntary or involuntary disclosure to a Governmental Authority or conducted any internal investigation or audit concerning any actual or potential violation or wrongdoing related to international trade laws or Anti-Money Laundering Laws.

(iii) Neither CB, the Bank, nor any of their representatives acting on their behalf have (1) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (2) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 or any other local or foreign anti-corruption or bribery Law or (3) made any other unlawful payment. Neither CB, the Bank, nor any of their representatives acting on their behalf have directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder CB or the Bank or assist CB or the Bank in connection with any actual or proposed transaction.

(iv) Neither CB, the Bank, nor any of their shareholders, directors, or officers, nor, to the knowledge of CB or the Bank, any other representative acting on behalf of CB or the Bank is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any U.S. sanctions administered by OFAC, and the Parties have not, since their incorporation, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any Subsidiary, joint venture partner or other Person, in connection with any sales or operations in Cuba, Iran, North Korea, Syria, the Donetsk People's Republic, Luhansk People's Republic, and Crimea regions of Ukraine, or any other country sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC.

(v) Except as specifically authorized by a governmental license, license exception, or other permit or applicable authorization of a Governmental Authority, neither CB nor the Bank have exported, reexported, transferred, facilitated or brokered the sale of any goods, services, technology, or technical data to or from, or entered into any transaction or had any dealing with, any person or entity for whom a license or other authorization is required under the U.S. Export Administration Regulations (the "EAR," 15 C.F.R. § 730 et seq.), the International Traffic in Arms Regulations (the "ITAR," 22 C.F.R. § 120 et seq.) or any other U.S. or non-U.S. export control regime, nor has CB or the Bank entered into any transaction prohibited by such laws.

23

(y) *Intellectual Property*. Section 5.02(y) of the Disclosure Schedule sets forth a complete list of all (i) registered trademarks, service marks, copyrights and patents; (ii) applications for registration or grant of any of the foregoing; and (iii) licenses for any of the foregoing, in each case, owned by CB or the Bank. The items on Section 5.02(y), together with all other trademarks, service marks, trade names, logos, assumed names, patents, copyrights, trade secrets, computer software, licenses, formulae, customer lists or other databases, business application designs and inventions currently used in or necessary to conduct the business of the Bank, in whatever form or medium, constitute the "Bank Intellectual Property." Except as set forth on Section 5.02(y) of the Disclosure Schedule, CB or the Bank has ownership of, or such other rights by license, lease or other agreement in and to, the Bank Intellectual Property as is necessary to permit the Bank to use the Bank Intellectual Property in the conduct of its businesses as presently conducted.

(z) *Data Privacy*. CB and the Bank (i) is and has been in compliance in all material respects with applicable Privacy and Security Requirements; (ii) has not experienced any Security Breaches; and (iii) has implemented commercially reasonable safeguards designed to protect Protected Data. Neither CB nor the Bank has received any written complaints regarding the unauthorized processing of Protected Data or non-compliance with applicable Privacy and Security Requirements. The execution or performance of this

Agreement will not, in any material respect, affect CB's or the Bank's rights to process Protected Data in the same manner as prior to the closing or violate any applicable Privacy and Security Requirements.

5.03. Representations and Warranties of Clear. Subject to Section 5.01, Clear hereby represents and warrants to CB and the Bank as follows:

(a) *Organization, Standing and Authority.* Clear is a corporation duly organized, validly existing and in good standing under the laws of the State of Wyoming. Clear is duly qualified to do business and is in good standing in the State of Wyoming.

(b) *Corporate Power*. Clear has the corporate power and authority to carry on its business as it is now being conducted and to own all its properties and assets; and Clear has the corporate power and authority to execute, deliver and perform its obligations under this Agreement and to consummate the transactions contemplated hereby.

(c) Corporate Authority; Authorized and Effective Agreement. This Agreement and the transactions contemplated hereby have been authorized by all necessary corporate action of Clear, and the Clear Board prior to the date hereof. This Agreement is a valid and legally binding agreement of Clear enforceable in accordance with its terms (except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer and similar laws of general applicability relating to or affecting creditors' rights or by general equity principles).

24

(d) Regulatory Approvals; No Defaults.

(i) No consents or approvals of, or filings or registrations with, any Governmental Authority or with any third party are required to be made or obtained in connection with the execution, delivery or performance by Clear of this Agreement or to consummate the Merger except for (A) filings of applications or notices with, and approvals or waivers by, the Federal Reserve Board, and the Wyoming Division of Banking, as may be required, (B) the filing of the Statement of Merger with the Secretary of State of the State of Wyoming pursuant to the WBCA; and (C) receipt of the approvals set forth in Section 8.01(b). As of the date hereof, Clear is not aware of any reason why the approvals set forth in Section 8.01(b) will not be received without the imposition of a condition, restriction or requirement of the type described in Section 8.01(b).

(ii) Subject to the satisfaction of the requirements referred to in the preceding paragraph and expiration of the related waiting periods, and required filings under federal and state securities laws, the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby do not and will not (A) constitute a breach or violation of, or a default under, or give rise to any Lien, any acceleration of remedies or any right of termination under, any law, rule or regulation or any judgment, decree, order, governmental permit or license, or agreement, indenture or instrument of Clear, any of its Subsidiaries, to which Clear, any of its Subsidiaries, or their properties is subject or bound, (B) constitute a breach or violation of, or a default under, the Articles of Incorporation or Bylaws (or similar governing documents) of Clear or any of its Subsidiaries, or (C) require any consent or approval under any such law, rule, regulation, judgment, decree, order, governmental permit or license, agreement, indenture or instrument.

ARTICLE VI COVENANTS

6.01 *Reasonable Best Efforts*. Subject to the terms and conditions of this Agreement, each of CB and Clear and agrees to use its respective reasonable best efforts in good faith to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or desirable, or advisable under applicable laws, expressly including Clear's best efforts to ensure that New Pubco execute all documents as may be reasonably requested by CB in connection with this Agreement, so as to permit consummation of the Merger as promptly as practicable and otherwise to enable consummation of the transactions contemplated hereby, including the satisfaction of the conditions set forth in Article VIII hereof, and shall cooperate fully with the other party hereto to that end.

6.02 Shareholder Approval/ Appraisal Rights.

(a) CB agrees to take, in accordance with applicable law and the CB Articles and the CB Bylaws, all action necessary to vote upon the approval of the principal terms of the Merger and any other matters required to be approved by CB's shareholders for consummation of the Merger. The CB Board shall at all times recommend such approval and shall take all reasonable lawful action to

solicit such approval by its shareholders (including the approvals, waivers and consents set forth in Section 8.01(c); provided, that nothing in this Agreement shall prevent the CB Board from withholding, withdrawing, amending or modifying its recommendation if the CB Board determines, after consulting with outside legal counsel, that such action is required in order for the directors to comply with their fiduciary duties to the CB shareholders under applicable law.

(b) Any Shareholder who neither votes in favor of nor consents to the Merger may, under certain circumstances and by following procedures under Section 17-16-1302 of the WBCA, a copy of which is attached hereto as <u>Exhibit A</u>, exercise appraisal rights to receive cash in an amount equal to the "fair value" of such Shareholders shares of CB common stock to which such Shareholder has exercised such appraisal or dissenters rights.

6.03 *Press Releases.* CB and Clear shall consult with each other before issuing any press release with respect to the Merger or this Agreement and shall not issue any such press release or make any such public statements without the prior consent of the other party, which consent shall not be unreasonably withheld or delayed. CB and Clear shall cooperate to develop all public announcement materials and make appropriate management available at presentations related to the transactions contemplated by this Agreement as reasonably requested by the other party.

6.04 Access; Information; Board Observation Rights.

(a) CB and Bank agree that upon reasonable notice and subject to applicable laws relating to the exchange of information, they shall afford Clear and Clear's officers, employees, counsel, accountants and other authorized representatives, such access during normal business hours throughout the period prior to the Effective Time to the books, records (including, without limitation, tax returns and work papers of independent auditors), properties, personnel and to such other information as Clear may reasonably request and, during such period, it shall furnish promptly to Clear (i) a copy of each material report, schedule and other document filed by CB or Bank pursuant to federal or state securities or banking laws, and (ii) all other information concerning the business, properties and personnel of CB and Bank as Clear may reasonably request.

(b) Each of Clear, CB and Bank agrees that it will not, and will cause its representatives not to use any information obtained pursuant to this Section 6.04 (as well as any other information obtained prior to the date hereof in connection with the entering into of this Agreement) for any purpose unrelated to the consummation of the transactions contemplated by this Agreement. Subject to the requirements of law, each party will keep confidential, and will cause its representatives to keep confidential, all information and documents obtained pursuant to this Section 6.04 (as well as any other information obtained prior to the date hereof in connection with the entering into of this Agreement) unless such information (i) was already known to such party, (ii) becomes available to such party from other sources not known by such party to be bound by a confidentiality obligation, (iii) is disclosed with the prior written approval of the party to which such information pertains, (iv) is or becomes readily ascertainable from published information or trade sources or (v) is disclosed to a potential investor in New Pubco pursuant to a confidentiality agreement in form and substance reasonably agreed to by CB and the Bank which is enforceable by CB and the Bank. No investigation by either party of the business and affairs of the other shall affect or be deemed to modify or waive any representation, warranty, covenant or agreement in this Agreement, or the conditions to either party's obligation to consummate the transactions contemplated by this Agreement. Notwithstanding the foregoing or anything to the contrary in this Agreement, Clear shall be permitted to disclose (and to permit QFTA and New Pubco to disclose) any information regarding the Company and/or its business as may be necessary or appropriate to include in the Registration Statement / Proxy Statement (as such term is defined in the Business Combination Agreement), any other filings to be made with the SEC by QFTA or New Pubco in connection with the transactions contemplated by this Agreement and the Business Combination Agreement, and/or any materials that may be provided to potential investors in the Potential Financing (as such term is defined in the Business Combination Agreement), unless otherwise prohibited by the Federal Reserve Board and the State of Wyoming Banking Division.

(c) During the period from the date of this Agreement to the Effective Time, CB and the Bank shall promptly furnish Clear copies of all monthly and other interim financial statements produced in the ordinary course of business as the same shall become available.

(d) Without limiting the generality of the foregoing, between the date hereof and the Final Closing, the Company shall (a) provide to Clear all financial statements of CB required to be provided pursuant to, and at such times and in conformity with all applicable requirements set forth in, Section 6.4 of the Business Combination Agreement; (b) use its reasonable best efforts (i) to assist Clear in causing to be prepared in a timely manner any other financial information or statements (including customary pro forma financial statements) that are required to be included in the Registration Statement / Proxy Statement and any other filings to be made by QFTA or New Pubco with the SEC in connection with the transactions contemplated by this Agreement and the Definitive Acquisition Agreement and (ii) to obtain the consents of CB or the Bank's auditors with respect thereto as may be required by applicable Law or requested by the SEC; and (c) notwithstanding anything to the contrary in this Agreement, reasonably cooperate with Clear, as may be requested by Clear from time to time, in assisting with the preparation of (i) the Registration Statement / Proxy Statement and any other filings to be made with the SEC by QFTA or New Pubco in connection with the transactions contemplated by this Agreement and the Definitive Acquisition Acquisition Agreement and (ii) any materials that may be provided to potential investors in the Potential Financing.

(e) CB and the Bank shall invite and allow up to three (3) Clear observers to attend and observe (but not vote or otherwise participate in) all regular and special meetings of the board of directors of CB or the Bank and all regular and special meetings of any senior management committee (including, but not limited to, the executive committee and the loan committee) of the Bank; provided that such observer may be excluded from any portion of any meeting in order to maintain any confidentiality, confidential privileges, including attorney-client privileged communications, or to allow confidential discussion of matters pertaining to the Bank, this Agreement and the transactions contemplated hereby. CB and the Bank shall share all board packages, financial reports and updated loan information with Clear at and in anticipation of such meetings; provided that CB and the Bank may exclude from any such materials provided to Clear any portion of such materials necessary to maintain any confidential privileges, including attorney-client privileged communications of matters pertaining to the Bank, this Agreement and the transactions confidential discussion of matters pertaining to the Bank may exclude from any such materials provided to Clear any portion of such materials necessary to maintain any confidential privileges, including attorney-client privileged communications, or to allow confidential discussion of matters pertaining to the Bank, this Agreement and the transactions contemplated hereby. If the Merger is finally disapproved by any appropriate regulatory authority or if this Agreement is terminated pursuant to its terms, Clear will no longer be entitled to notice of and permission to attend such meetings. All information obtained by Clear in connection with these meetings shall be treated in confidence as provided in Section 6.13 hereof.

Acquisition Proposals. CB and Bank agree that neither it nor its officers, directors or Affiliates shall, and that it shall direct and use its best efforts to cause its employees, agents and representatives (including any financial advisor, attorney or accountant retained by it) not to, directly or indirectly, initiate, solicit, encourage or otherwise facilitate any inquiries or the making of any proposal or offer with respect to a merger, reorganization, share exchange, consolidation or similar transaction involving, or any purchase of all or substantially all of the assets of or more than 10% of the outstanding equity securities of, CB (any such proposal or offer being hereinafter referred to as an "Acquisition Proposal"). CB and Bank further agree that neither it nor any of its officers and directors shall, and that it shall direct and use its reasonable best efforts to cause its employees, Affiliates, agents and representatives (including any financial advisor, attorney or accountant retained by it) not to, directly or indirectly, engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions with, any Person relating to an Acquisition Proposal, or otherwise facilitate any effort or attempt to make or implement an Acquisition Proposal.

6.06 *Reports and Filings*. Each of CB, the Bank and Clear shall file between the date of this Agreement and the Effective Time, all reports required to be filed by it with all Regulatory Authorities having jurisdiction over such party. Any financial statements contained in any reports to a Regulatory Authority shall be prepared in accordance with requirements applicable to such reports.

27

6.07 Regulatory Applications.

(a) CB, Bank and Clear, (and, when timely, Clear will ensure that New Pubco), shall cooperate and use their respective reasonable best efforts to prepare all documentation, to timely effect all filings and to obtain all permits, consents, approvals and authorizations of all third parties and Governmental Authorities necessary to consummate the transactions contemplated by this Agreement. Clear, QFTA and New Pubco will assume primary responsibility to effect all filings and will pay all costs and expenses related thereto per Section 10.06. Each party hereto agrees that it will consult with the other party hereto with respect to the obtaining of all material permits, consents, approvals and authorizations of all third parties and Governmental Authorities necessary or advisable to consummate the transactions contemplated by this Agreement and each party will keep the other party apprised of the status of material matters relating to completion of the transactions contemplated hereby. Clear, QFTA and New Pubco will ensure that any initial filing with Governmental Authorities shall be made as soon as reasonably practicable after the execution hereof. Subject to applicable laws

relating to the exchange of information, each of CB and Bank shall have the right to review in advance and, to the extent practicable, consult with the other on all material written information submitted to any third party and/or any Governmental Authority in connection with the Merger and the other transactions contemplated by this Agreement. In exercising the foregoing right, each of such parties agrees to act reasonably and as promptly as practicable.

(b) Each party agrees, upon request, to furnish the other party with all information concerning itself, its Subsidiaries, directors, officers and shareholders and such other matters as may be reasonably necessary or advisable in connection with any filing, notice or application made by or on behalf of such other party or any of its Subsidiaries to any third party or Governmental Authority.

6.08 Notification of Certain Matters. Each of CB, the Bank and Clear shall give prompt notice to the other of any fact, event or circumstance known to it that (i) is reasonably likely, individually or taken together with all other facts, events and circumstances known to it, to result in a Material Adverse Effect or to prevent, materially delay or materially impair the ability of CB or Clear, as the case may be, to consummate the transactions contemplated by this Agreement or (ii) would cause or constitute a material breach of any of its representations, warranties, covenants or agreements contained herein.

6.09 No Breaches of Representations or Warranties. Between the date of this Agreement and the Effective Time, without the written consent of the other party, each of Clear, CB, and Bank will not do any act or suffer any omission of any nature whatsoever that would cause any of the representations or warranties made in Article V of this Agreement to become untrue or incorrect in any material respect.

6.10 Consents. Each of Clear, CB and Bank shall use its best commercial efforts to obtain any required consents to the transactions contemplated by this Agreement.

6.11 *Insurance Coverage.* CB and Bank shall cause each of the policies of insurance listed in its Disclosure Schedule to remain in effect between the date of this Agreement and the Effective Date.

6.12 *Correction of Information.* Each of Clear, CB and Bank shall promptly correct and supplement any information furnished under this Agreement so that such information shall be correct and complete in all material respects at all times, and shall include all facts necessary to make such information correct and complete in all material respects at all times, provided that any such correction or supplement that may result in a change to a party's Disclosure Schedule ("Updated Disclosure Schedule") shall not be made without the prior written consent of the other party, which shall not be unreasonably withheld.

6.13 *Confidentiality.* Except for the use in connection with the governmental filings required in order to complete the transactions contemplated by this Agreement, all information (collectively, the "Information") received by each of CB, Bank and Clear, pursuant to the terms of this Agreement shall be kept in strictest confidence; CB, Bank, and Clear agree that the Information will be used only for the purpose of completing the transactions contemplated by this Agreement. CB, Bank, and Clear agree to hold the Information in strictest confidence and shall not use, and shall not disclose directly or indirectly any of such Information except when, after and to the extent such Information (i) is or becomes generally available to the public other than through the failure of CB, Bank, or Clear to fulfill its obligations hereunder, (ii) was already known to the party receiving the Information on a nonconfidential basis prior to the disclosure or (iii) is subsequently disclosed to the party receiving the Information on a nonconfidential basis by a third party having no obligation of confidentiality to the party disclosing the Information.

6.14 Allowance for Loan and Lease Losses; Pre-Closing Loan Review.

(a) The Bank will continue to adhere to its policies and procedures, as well as all guidance from any applicable Regulatory Authority and GAAP and RAP, related to its allowance for loan and lease losses as the Bank has in its normal course of business.

28

(b) Any additional adjustments to the Bank's allowance for loan and lease losses or other accounts required by the Clear, which are allowable under all applicable Regulatory Authority and GAAP and RAP, will be made by the Bank at or immediately prior to Effective Time. Any such adjustments will not be included in the Calculation Statement or otherwise impact the calculation of the Merger Consideration, or the total proceeds to be received by the shareholders of CB.

(c) The Bank shall make available to Clear all reasonably requested information regarding the status of each loan contained in the loan portfolio of the Bank, as of a date not more than 15 days prior to the Effective Time.

6.15 *Continuity of Interest.* The parties agree to continue the operation of the Bank for not less than two (2) years following the Effective Time in a manner substantially similar to the operation of the Bank prior to the Effective Time.

6.16 Offering of New Pubco Common Stock.

New Pubco will use reasonable best efforts to cause the offer and sale of the New Pubco Common Stock issuable in the Merger to be registered on the Registration Statement on Form S-4 filed in connection with the transactions contemplated by the Business Combination Agreement (the "Form S-4"). If the SEC does not permit the offer and issuance of the registration of the New

- (a) Pubco Common Stock on the Form S-4, New Pubco shall offer and issue the New Pubco Common Stock in a transaction exempt from the registration requirements of Section 5 of the Securities Act of 1933, as amended (a "Private Placement"); provided that New Pubco shall not be required to issue any New Pubco Common Stock unless such issuance complies with all federal and state securities laws.
- (b) If the New Pubco Common Stock is registered on the Form S-4, as promptly as practicable after the effective date of the Form S-4, New Pubco shall mail to each CB shareholder a Prospectus, together with an Election Form and Letter of Transmittal.

If New Pubco is not permitted to register the New Pubco Common Stock issuable in the Merger on the Form S-4, as soon as practicable after such determination, New Pubco shall mail to each CB Stockholder an Investor Questionnaire requesting that each CB Stockholder return such Investor Questionnaire by the Certification Deadline specified therein, which shall be ten (10) Business Days after the mailing date of such Investor Questionnaire. CB shall promptly provide New Pubco with a copy of all completed Investor Questionnaires. If New Pubco is not permitted to register the New Pubco Common Stock is not permitted to register the New Pubco Com

(c) issuable in the Merger, New Pubco shall prepare an Offering Memorandum and cause such Offering Memorandum to be mailed to CB shareholders that have provided a properly completed Investor Questionnaire certifying that such holder of CB Common Stock is an accredited investor and to each unaccredited investor that has appointed a Purchaser Representative. Any CB Stockholder that (i) does not return an Investor Questionnaire by the Certification Deadline or (ii) does not certify in the Investor Questionnaire that it is an "accredited investor" (or that it is an unaccredited investor and has appointed a purchaser representative in accordance with Rule 502) shall be deemed to have selected Option B (all cash).

New Pubco shall not be required to offer or issue shares of New Pubco Common Stock to any CB Stockholder if such offer or issuance would, in its reasonable judgement, be in violation of any federal or state securities laws. New Pubco shall be
(d) under no obligation to issue any New Pubco Common Stock unless the transactions contemplated by the Business Combination Agreement have been consummated. Subject to the foregoing, New Pubco shall cause to be deposited with the Paying Agent, the shares of New Pubco Common Stock issuable pursuant to Article III hereof.

(e) The parties hereto agree that all obligations of New Pubco under this Section 6.16 to issue any New Pubco Common Stock shall be conditioned upon the consummation of the transactions contemplated the Business Combination Agreement and that all obligations of New Pubco under this Section 6.16 shall terminate without action of any party upon termination of the Business Combination Agreement.

If New Pubco is not permitted to register the New Pubco Common Stock issuable in the Merger on the Form S-4 and instead issues such stock in a Private Placement, as soon as practicable (and in any event within 30 calendar days following the Closing

(f) Date) New Pubco shall file a registration statement with the SEC providing for the resale by CB shareholders of shares received as part of the Merger Consideration (the "Resale Registration Statement"). Clear shall use commercially reasonable efforts to cause such Resale Registration Statement to become effective within 60 calendar days from Closing Date.

ARTICLE VII ESCROW AND HOLDBACK ACCOUNT

The funds deposited in the Escrow Account will comprise a \$100,000 good faith deposit by Clear and, unless previously released, will convert on Closing Date to a Holdback fund to be reserved in the Escrow for a period of 12 months post-Closing as a fund

for potential reimbursement to Clear, as the circumstances dictate, in the event of costs incurred by them through payment of claims for tax or other liabilities arising from breach by CB or Bank of the representations, warranties or covenants contained in this Agreement.

ARTICLE VIII CONDITIONS TO CONSUMMATION OF THE MERGER

8.01 Conditions to Each Party's Obligation to Effect the Merger. The respective obligation of each of Clear and CB to consummate the Merger is subject to the fulfillment or written waiver by Clear and CB prior to the Effective Time of each of the following conditions:

(a) *Shareholder Approval*. This Agreement shall have been duly adopted by the requisite vote of Clear, New Pubco, Bank and CB's shareholders, and the issuance of New Pubco Stock shall have been duly approved by the requisite vote of New Pubco stockholders if required.

(b) *Regulatory Approvals*. All regulatory approvals set forth on Schedule 8.01(b) shall have been obtained and shall remain in full force and effect and all statutory waiting periods in respect thereof shall have expired and no such approvals shall contain (i) any restrictions or requirements that the Clear Board reasonably determines would either before or after the Effective Time have a material adverse effect on Clear, New Pubco, Bank or CB after giving effect to the consummation of the Merger, or (ii) reduce the benefits of the transactions contemplated hereby to such a degree that Clear, New Pubco, CB and Bank would not have entered into this Agreement had such conditions, restrictions or requirements been known as of the date hereof.

(c) *No Injunction*. No Governmental Authority of competent jurisdiction shall have enacted, issued, promulgated, enforced, or entered any statute, rule, regulation, judgment, decree, injunction or other order (whether temporary, preliminary or permanent) that is in effect and prohibits consummation of the transactions contemplated by this Agreement.

(d) *Business Combination*. The transactions contemplated by the Business Combination Agreement, including the acquisition of private company Wilson-Davis by Clear, shall have been completed.

3	0
~	~

8.02 *Conditions to Obligation of CB*. The obligation of CB to consummate the Merger is also subject to the fulfillment or written waiver by CB prior to the Effective Time of each of the following conditions:

(a) *Representations and Warranties*. The representations and warranties of Clear set forth in this Agreement shall be true and correct in all material respects, as of the date of this Agreement and as of the Effective Date as though made on and as of the Effective Date (except that representations and warranties that by their terms speak as of the date of this Agreement or some other date shall be true and correct as of such date), and CB shall have received a certificate, dated the Effective Date, signed on behalf of Clear by the Chief Executive Officer of Clear to such effect.

(b) *Performance of Obligations of Clear*. Clear shall have performed in all material respects all obligations required to be performed by Clear under this Agreement at or prior to the Effective Time, and CB shall have received a certificate, dated the Effective Date, signed on behalf of Clear by the Chief Executive Officer of Clear to such effect.

(c) New Pubco Duties and Obligations. New Pubco shall have executed this Agreement.

8.03 *Conditions to Obligation of Clear*. The obligation of Clear to consummate the Merger is also subject to the fulfillment or written waiver by Clear prior to the Effective Time of each of the following conditions:

(a) *Representations and Warranties*. The representations and warranties of CB and Bank set forth in this Agreement (without giving effect to any Updated Disclosure Schedule delivered by CB and Bank pursuant to Section 6.12, except for changes to Section 5.02 Schedules reflecting actions taken by CB and the Bank in compliance with Section 4.01) shall be true and correct in all material respects, as of the date of this Agreement and as of the Effective Date as though made on and as of the Effective Date (except that representations and warranties that by their terms speak as of the date of this Agreement or some other date shall be true and correct as of such date) and Clear shall have received a certificate, dated the Effective Date, signed on behalf of CB and the Bank by their respective authorized officers to such effect.

(b) *Performance of Obligations of CB and Bank*. CB and Bank shall have performed in all material respects all obligations required to be performed by CB or Bank under this Agreement at or prior to the Effective Time, and Clear shall have received a certificate, dated the Effective Date, signed on behalf of CB and Bank by an authorized officer of each to such effect.

(c) *Consents*. CB and the Bank shall have obtained each of the consents listed in Section 5.02(k) of the Disclosure Schedules and any consents of the type required to be identified in Section 5.02(k) of the Disclosure Schedule but which were not so identified as of the date of this Agreement. A copy of each such consent shall have been delivered to Clear.

(d) *No Adverse Changes*. Between the date of this Agreement and the Effective Date there shall not have occurred and be existing any change or any condition, event, circumstance, fact or occurrence, other than as provided in this Agreement, that would have a Material Adverse Effect.

31

(e) *Transaction Costs*. CB and the Bank shall have used its reasonable best efforts to cause its legal, accounting, financial and other advisors to submit final bills or estimates of final bills for all professional fees related to the transactions contemplated by this Agreement by the Calculation Date to allow them to prepare the Calculation Statement accounting for such Transaction Costs owing by them. Based upon such final bills or estimates of such final bills, CB and the Bank shall have paid all such Transaction Costs in full prior to or immediately prior to the Effective Time. In no event shall Clear be liable for any such Transaction Costs in connection with the transactions contemplated by this Agreement, other than expenses, if any, incurred by CB or the Bank in connection with applications for regulatory approvals.

ARTICLE IX TERMINATION

9.01 *Termination by Mutual Consent*. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the approval by shareholders referred to in Section 8.01(a), by the mutual written consent of Clear and CB by action of their respective boards of directors.

9.02 *Termination by Either Clear or CB.* This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the approval by shareholders referred to in Section 8.01(a), by action of the board of directors of either Clear or CB, in the event:

(a) The Merger is not consummated within two (2) years of the date of this Agreement, (providing the applications for approval set forth in Section 6.06 are no longer pending) except to the extent that the failure of the Merger then to be consummated arises out of or results from the knowing action or inaction of the party seeking to terminate pursuant to this Section 9.02(a), which action or inaction is in violation of its obligations under this Agreement.

(b) (i) The approval of any Governmental Authority required for consummation of the Merger and the other transactions contemplated by this Agreement shall have been denied by final and nonappealable action of such Governmental Authority or an application therefor shall have been permanently withdrawn at the invitation, request or suggestion of a Governmental Authority or (ii) the shareholder approval referred to in Section 8.01(a) herein is not obtained.

(c) The Business Combination Agreement is terminated.

9.03 *Termination by CB*. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after approval by shareholders of CB referred to in Section 8.01(a), by action of the CB Board if there has been a breach of any representation, warranty, covenant or agreement made by Clear in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Sections 8.02(a) or 8.02(b) would not be satisfied and such breach or condition is not curable or, if curable, is not cured within thirty (30) days after written notice thereof is given by CB to Clear, provided that CB or Bank is not itself in material breach of any provision of this Agreement.

9.04 *Termination by Clear*. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after approval by shareholders referred to in Section 8.01(a), by action of the Clear Board, in the event that:

(a) There has been a breach of any representation, warranty, covenant or agreement made by CB or the Bank in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Sections 8.03(a), 8.03(b), 8.03(c) or 8.03(d) would not be satisfied and such breach or condition is not curable or, if curable, is not cured within thirty (30) days after written notice thereof is given by Clear to CB, provided that Clear is not itself in material breach of any provision of this Agreement.

(b) CB or the Bank shall have breached Section 6.06 or the CB or Bank Board shall have failed to make its recommendation referred to in Section 6.02, withdrawn such recommendation or adversely modified or changed such recommendation or failed to reconfirm its recommendation of this Agreement within five (5) Business Days after a written request by Clear to do so.

9.05 *Effect of Termination and Abandonment.* In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article IX, this Agreement (other than as set forth in Section 9.01) shall become void and of no effect with no liability or further obligation on the part of any party hereto (or of any of its directors, officers, employees, agents, legal and financial advisors or other representatives); provided, however, except as otherwise provided herein, no such termination shall relieve any party hereto of any liability or damages resulting from any willful breach of this Agreement

32

ARTICLE X MISCELLANEOUS

10.01 *Survival.* Except as relates to the Holdback in the Escrow, (see ARTICLE VII) the representations, warranties, agreements and covenants contained in this Agreement shall not survive the Effective Time, provided, however, except as otherwise provided herein, no termination shall relieve any party hereto of any liability or damages resulting from any willful breach of this Agreement.

10.02 *Waiver; Amendment.* Prior to the Effective Time, any provision of this Agreement may be (i) waived by the party benefited by the provision, or (ii) amended or modified at any time, by an agreement in writing between the parties hereto executed in the same manner as this Agreement, except to the extent that any such amendment would violate applicable law or require resubmission of this Agreement or the plan of merger contained herein to the shareholders of CB.

10.03 *Counterparts*. This Agreement may be executed in one or more counterparts, each of which shall be deemed to constitute an original.

10.04 Governing Law and Venue; Waiver of Jury Trial.

(a) THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAWS OF THE STATE OF WYOMING, WITHOUT REGARD TO THE CONFLICT OF LAW PRINCIPLES THEREOF. The parties hereby irrevocably submit to the jurisdiction of the courts of the State of Wyoming and the federal courts of the United States of America located in Wyoming solely in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and in respect of the transactions contemplated hereby and thereby, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for the interpretation or enforcement hereof or of any such documents, that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such action or proceeding shall be heard and determined in such Wyoming state or federal court. The parties hereby consent to and grant any such court jurisdiction over the person of such parties and over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 10.04 or in such other manner as may be permitted by law, shall be valid and sufficient service thereof.

(b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH

SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF THE OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.04.

10.05 *Counterparts*. This Agreement may be executed in one or more counterparts, each of which shall be deemed to constitute an original.

10.06 *Expenses*. Whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the Merger and the other transactions contemplated by this Agreement, other than regulatory application expenses, (all of which will be paid by Clear), shall be paid by the party incurring such expense.

10.07 *Notices.* All notices, requests and other communications hereunder to a party shall be in writing and shall be deemed given if personally delivered, faxed (with confirmation) or mailed by registered or certified mail (return receipt requested) or private courier delivered to such party at its address set forth below or such other address as such party may specify by notice to the parties hereto.

33

214 Main Street	
Pine Bluffs, WY 82082	
Kenneth C. Wolfe, Esq	
Wolfe, Van Ackern & Stephenson LLP	
1008 Centre Avenue, Suite A	
Fort Collins, CO 80526	
Grea Gross	
Dine Bluffs WV 82082	
1 me Diuns, w 1 02002	
(same)	
Craig Ridenhour	
4221 W Boy Scout Blyd	
Tampa, 1155007	
Fairfield and Woods, P.C.	
1801 California Street, Suite 2600	
Denver, CO 80202	
Attn: Robert M. Vinton, Esq.	
Cell: 303 859 8596	
Con. 505 057 0570	
	1008 Centre Avenue, Suite A Fort Collins, CO 80526

If to New Pubco, to:

With a copy to:

John Schaible 4221 W. Boy Scout Blvd. Suite 300 Tampa FL 33607

Winston & Strawn LLP 200 Park Avenue New York, NY 10166-4700 Attention: Jason Osborn josborn@winson.com

34

10.08 Entire Understanding; No Third-Party Beneficiaries. Nothing in this Agreement, whether express or implied, is intended to confer upon any Person, other than the parties hereto or their respective successors, any rights, remedies, obligations or liabilities under or by reason of this Agreement; provided, however, that New Pubco and QFTA are express third-party beneficiaries of the representations and warranties of the Company and Bank set forth in Section 5.02 of this Agreement.

10.09 *Interpretation; Effect.* When a reference is made in this Agreement to Sections, Exhibits or Schedules, such reference shall be to a Section of, or Exhibit or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and are not part of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

10.10 Severability. If any provision of this Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions, or the application of such provision to Persons or circumstances other than those as to which it has been held invalid or unenforceable, will remain in full force and effect and will in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination, the parties will negotiate in good faith in an effort to agree upon a suitable and equitable substitute provision to effect the original intent of the parties.

10.11 Assignment. Clear, New Pubco and CB may not assign any of their rights or obligations under this Agreement to any other Person, except upon the prior written consent of the other Party, and subject to the express understanding that it is the intention that Clear and New Pubco will consummate between them an acquisition transaction.

[signature page follows]

35

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed in counterparts by their duly authorized officers, all as of the day and year first above written.

CB:

By: /s/ Wynema Engstrom

Name:	Wynema Engstrom
Title:	Chairwoman and President

Bank:

By: /s/ Wynema Engstrom

Name: Wynema Engstrom Title: Chairwoman and President

AtlasClear, Inc.:

By:	/s/ Craig Ridenhour	
Name:	Craig Ridenhour	
Title:	Chief Business Development Officer	
Calculator New Pubco, Inc.		
By:	/s/ Steven J. Carlson	
NT		

Name: Steven J. Carlson Title: President and Secretary

36

SCHEDULE I

Sample Merger Consideration Calculation

SCHEDULE 1-SAMPLE MERGER CONSIDERATION CALCULATION

For illustrative purposes, based on the following assumptions (final numbers will change other than \$10 value for New Pubco common stock and \$150,000 NOL Tax Benefit):

(1) Adjusted Estimated Book Value ("AEBV") as of 10/24/22 is \$2,604,000 (based on Equity Capital of \$2,700,000 minus \$96,000 loan impairment)

(2) New Pubco common stock will be delivered under Option A at \$10.00 per share

(3) CB common stock has 449,043 shares outstanding

(4) per above, per share AEBV is \$5.80

(5) in addition to AEBV, shareholders will receive a pro rata share of Premises valued at

400,000 (* number may change based on Appraised Value) less value on books of 121,350 = 278,650 (278,650 / 449,043) = 6205 per share

150,000 for NOL Tax Benefit (150,000/449,043) = 3340

As an example, assume 85% (381,686)) of shares outstanding are held by shareholders electing Option A, which is 3 X AEBV with 1/3 in cash and 2/3 in equity:

Sample Calculation:

Option A-3x per share AEBV: 5.80 X3 = 17.40 plus, 6205 for Premises above what it is carried on the books, plus <math>3340 for NOL Tax Benefit = 18.3545 per share

Total value to Option A electors -\$18.3545 X 381,686 = \$7,005,656

Stock portion X .66 = \$4,670,390.63

Number of shares of QFTA common at \$10 per share—467,039

Cash payment: (\$6,338,315.88 X .33) = \$2,335,195.31

Option B electors comprising 15% of outstanding shares (67,356 shares) receive:

AEBV per share of \$5.80 X2 = \$11.60, plus \$.6205 (Premises), plus \$.3340 for NOL Tax Benefit = \$12.5545 per share

Total B-(\$12.5545 X 67,356) = \$845,620.90

New Pubco common stock: \$4,670,390.63

Option A cash: \$2,335,195.31

Option B cash: \$845,620.90

TOTAL \$7,851.206.84

EXHIBIT A

17-16-1302 of the Wyoming Business Corporation Act

§ 17-16-1302. Right to appraisal.

(a) A shareholder is entitled to appraisal rights, and to obtain payment of the fair value of his shares in the event of, any of the following corporate actions:

(i) Consummation of a plan of merger or consolidation to which the corporation is a party if:

(A) Shareholder approval is required for the merger or the consolidation by W.S. 17-16-1104 or 17-16-1111 and the shareholder is entitled to vote on the merger or consolidation, except that appraisal rights shall not be available to any shareholder of the corporation with respect to shares of any class or series that remain outstanding after consummation of the merger; or

(B) The corporation is a subsidiary that is merged with its parent under W.S. 17-16-1105.

(ii) Consummation of a share exchange to which the corporation is a party as the corporation whose shares will be acquired, if the shareholder is entitled to vote on the exchange, except that appraisal rights shall not be available to any shareholder of the corporation with respect to any class or series of shares of the corporation that is not exchanged;

(iii) Consummation of a disposition of assets pursuant to W.S. 17-16-1202 if the shareholder is entitled to vote on the disposition;

(iv) An amendment of the articles of incorporation with respect to a class or series of shares that:

(A) Alters or abolishes a preferential right of the shares;

(B) Creates, alters or abolishes a right in respect of redemption, including a provision respecting a sinking fund for the redemption or repurchase, of the shares;

(C) Alters or abolishes a preemptive right of the holder of the shares to acquire shares or other securities;

(D) Excludes or limits the right of the shares to vote on any matter, or to cumulate votes, other than a limitation by dilution through issuance of shares or other securities with similar voting rights; or

(E) Reduces the number of shares of a class or series owned by the shareholder to a fraction of a share if the corporation has the obligation or right to repurchase the fractional share so created.

(v) Any other amendment to the articles of incorporation, merger, share exchange or disposition of assets if specifically provided in the articles of incorporation, bylaws or a resolution of the board of directors;

(vi) Consummation of a transfer or domestication if the shareholder does not receive shares in the foreign corporation resulting from the transfer or domestication that have terms as favorable to the shareholder in all material respects, and represent at least the same percentage interest of the total voting rights of the outstanding shares of the corporation, as the sharehold by the shareholder before the transfer or domestication;

(vii) Consummation of a conversion of the corporation to nonprofit status; or

(viii) Consummation of a conversion of the corporation to an unincorporated entity.

(b) Notwithstanding subsection (a) of this section, the availability of appraisal rights under paragraphs (a)(i), (ii), (iii), (iv), (vi) and (viii) of this section shall be limited in accordance with the following provisions:

(i) Appraisal rights shall not be available for the holders of shares of any class or series of shares which is:

(A) A covered security under section 18(b)(1)(A) or (B) of the Securities Act of 1933, as amended; or

(B) Traded in an organized market and has at least two thousand (2,000) shareholders and a market value of at least twenty million dollars (\$20,000,000.00), exclusive of the value of such shares held by the corporation's subsidiaries, senior executives, directors and beneficial shareholders owning more than ten percent (10%) of such shares; or

(C) Issued by an open end management investment company registered with the Securities and Exchange Commission under the Investment Company Act of 1940 and may be redeemed at the option of the holder at net asset value.

(ii) The applicability of paragraph (i) of this subsection shall be determined as of:

(A) The record date fixed to determine the shareholders entitled to receive notice of, and to vote at, the meeting of shareholders to act upon the corporate action requiring appraisal rights; or

(B) The day before the effective date of such corporate action if there is no meeting of shareholders.

(iii) Paragraph (i) of this subsection shall not be applicable and appraisal rights shall be available pursuant to subsection (a) of this section for the holders of any class or series of shares who are required by the terms of the corporate action requiring appraisal rights to accept for such shares anything other than cash or shares of any class or any series of shares of any corporation, or any other proprietary interest of any other entity, that satisfies the standards set forth in paragraph (i) of this subsection at the time the corporate action becomes effective;

(iv) Reserved.

Quantum FinTech Acquisition Corporation to Acquire AtlasClear

AtlasClear will operate through an acquisition of technology assets from Atlas Fintech Holdings Corp., combined with the acquisition of correspondent clearing broker dealer, Wilson Davis & Co., Inc.

AtlasClear also party to definitive agreement to acquire Federal Reserve Member, Commercial Bancorp of Wyoming; acquisition expected to close Q2 or Q3 2023 subject to regulatory approvals

Robert McBey, CEO of Wilson Davis & Co., Inc., and Craig Ridenhour, Chief Business Development Officer of Atlas FinTech Holdings Corp., will lead the combined company

TAMPA and SALT LAKE CITY, Nov. 16, 2022 (GLOBE NEWSWIRE) — AtlasClear, Inc. ("AtlasClear"), Atlas Fintech Holdings Corp. ("Atlas FinTech") and Quantum FinTech Acquisition Corporation ("QFTA") (NYSE: QFTA), a publicly traded special purpose acquisition company, today announced that they have entered into a definitive business combination agreement that will result in Atlas FinTech transferring its trading technology assets to AtlasClear and the acquisition by AtlasClear as part of the initial business combination of Wilson Davis & Co., Inc., a correspondent clearing broker-dealer ("WDCO"). AtlasClear has also entered into a definitive agreement to acquire a federal reserve member following consummation of the initial business combination, expected in the second or third quarter of 2023.

As a result of the transactions, AtlasClear will provide specialized banking and clearing services to other financial services firms, with an emphasis on global markets currently underserviced by larger vendors. Once properly integrated, synergies between the federal reserve member and the correspondent clearing company are expected to allow for lower cost of capital, higher net interest margins, expanded product development and greater credit extension. Upon closing, the combined company will operate as AtlasClear, Inc. and its common stock is expected to be listed on the New York Stock Exchange under the ticker symbol "ATCL".

"AtlasClear is expected to possess best in class FinTech capabilities for its target marketplace in trading, clearing and settling financial products. Together with the proposed acquisitions of a correspondent clearing broker dealer and a Federal Reserve Member Bank, we believe AtlasClear will be well positioned to execute its strategy to service small and mid-sized financial services firms," said Robert McBey, CEO of WDCO.

"We expect that going public will provide us with a source of capital to grow AtlasClear and advance our strategic priorities, while acquiring and integrating WDCO and CB will add the requisite regulatory structures. Together, we believe this business combination will advance our vision to create a FinTech forward alternative for market participants," said Craig Ridenhour, CBDO of Atlas FinTech Holdings.

Key Transaction Terms

The board of directors and stockholders of AtlasClear have approved the business combination. The board of directors of QFTA has approved the business combination based upon the unanimous recommendation of a special committee of independent directors. The transaction will require the approval of the stockholders of QFTA, and is subject to other customary closing conditions, including the receipt of certain regulatory approvals. The transaction is currently expected to close in the first quarter of 2023. The acquisition of WDCO is expected to be effected pursuant to the terms and conditions of the stock purchase agreement entered into by AtlasClear and WDCO and the acquisition of CB is expected to be effected pursuant to the terms and conditions of the merger agreement entered into by AtlasClear and CB, each of which contains customary closing conditions.

About AtlasClear

AtlasClear plans to build a cutting-edge technology enabled financial services firm that would create a more efficient platform for trading, clearing, settlement and banking of evolving and innovative financial products with a focus on the small and middle market financial

services firms. The team that will lead AtlasClear are respected financial services industry veterans that have founded and led other companies in the industry including Penson Clearing, Southwest Securities, NexTrade, and Anderen Bank.

About Quantum FinTech Acquisition Corporation

Quantum FinTech Acquisition Corporation is a blank check company, also commonly referred to as a special purpose acquisition company, or SPAC, that was formed for the purpose of entering into a merger, share exchange, asset acquisition, stock purchase, recapitalization, reorganization or other similar business combination with one or more businesses, with a principal focus on identifying high-growth financial services and fintech businesses as targets.

About the Financial Technology

The nature of the combined entity is expected to be supported by robust, proven, financial technologies with a full suite that will enable the flow of business and success of the enterprise. We expect that the combined entity will have a full exchange platform for a spectrum of financial products. In addition, we expect that the combined entity will have a full prime brokerage and, following the Commercial Bancorp acquisition, a prime banking platform with complete front-end delivery. The enterprise will be able to offer a fixed income risk management platform which can be expanded to a diverse application on financial products.

We anticipate that the combined entity will be run by a new digital suite of technologies that will be part of the transaction at closing.

About Wilson Davis & Co.

Wilson Davis & Co. is a full-service correspondent securities broker-dealer. The company is registered with the Securities and Exchange Commission ("SEC"), the Financial Industry Regulatory Authority and the Securities Investor Protection Organization. In addition, WDCO is a member of the Depository Trust and Clearing Corporation as well as the National Securities Clearing Corporation. Headquartered in Salt Lake City, Utah and Dallas, Texas, WDCO has been servicing the investment community since 1968, with satellite offices in California, Arizona, Colorado, New York, New Jersey and Florida.

About Commercial Bancorp of Wyoming

Commercial Bancorp of Wyoming began is operations as a bank through its wholly owned subsidiary, Farmer's State Bank ("FSB") in 1915 and has been servicing the local community in Pine Bluffs, WY. It has focused the majority of its services on private and corporate banking. A member of the Federal Reserve, we expect FSB to be a strategic asset for AtlasClear's long-term business model.

Additional Information and Where to Find It

In connection with the proposed business combination and related transactions contemplated in connection therewith (the "Proposed Transaction"), the new public holding company ("New Pubco") intends to file with the SEC a registration statement on Form S-4 containing a preliminary proxy statement of QFTA and prospectus of New Pubco, and after the registration statement is declared effective, QFTA will mail a definitive proxy statement/prospectus relating to the Proposed Transaction to its stockholders. This press release does not contain any information that should be considered by QFTA's stockholders concerning the Proposed Transaction and is not intended to constitute the basis of any voting or investment decision in respect of the Proposed Transaction or the securities of New Pubco. QFTA's stockholders and other interested persons are advised to read, when available, the preliminary proxy statement/prospectus and the amendments thereto and the definitive proxy statement/prospectus and other documents filed in connection with the Proposed Transaction. When available, the definitive proxy statement/prospectus will be mailed to stockholders of QFTA as of a record date to be established for voting on the Proposed Transaction. Stockholders will also be able to obtain copies of the preliminary proxy statement/ prospectus and other documents filed with the SEC, without charge, once available, at the SEC's website at www.sec.gov, or by directing a request to: Quantum FinTech Acquisition Corporation, 4221 W Boy Scout BLVD Suite 300, Tampa FL 33607, Attention: Investor Relations or by email at IR@qftacorp.com.

No Offer or Solicitation

This press release shall not constitute a "solicitation" as defined in Section 14 of the Securities Exchange Act of 1934, as amended. This press release does not constitute an offer, or a solicitation of an offer, to buy or sell any securities, investment or other specific product, or a solicitation of any vote or approval, nor shall there be any offer, solicitation or sale of securities, investment or other specific product in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended, or an exemption therefrom.

Participants in Solicitation

QFTA, AtlasClear and their respective directors and executive officers may be deemed participants in the solicitation of proxies from QFTA stockholders with respect to the Proposed Transaction. QFTA stockholders and other interested persons may obtain, without charge, more detailed information regarding the directors and officers of QFTA in its Annual Report on Form 10-K, filed with the SEC on March 10, 2022, as amended (the "2021 Form 10-K"), which is available free of charge at the SEC's website at www.sec.gov. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to QTFA stockholders will be set forth in the proxy statement/prospectus for the Proposed Transaction when available. Additional information regarding the interests of the participants in the solicitation of proxies from QFTA's stockholders with respect to the Proposed Transaction when available.

Cautionary Statement Regarding Forward-Looking Statements

This communication contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, as amended, that reflect AtlasClear's and QFTA's current views with respect to, among other things, the future operations and financial performance of AtlasClear, QFTA and the combined company. Forward-looking statements in this communication may be identified by the use of words such as "anticipate," "assume," "believe," "continue," "could," "estimate," "expect," "foreseeable," "future," "intend," "may," "outlook," "plan," "potential," "proposed" "predict," "project," "seek," "should," "target," "trends," "will," "would" and similar terms and phrases. Forward-looking statements contained in this communication include, but are not limited to, statements as to (i) expectations regarding the Proposed Transaction, including timing for its consummation, (ii) anticipated use of proceeds from the transaction, (iii) AtlasClear and QFTA's expectations as to various operational results and market conditions, (iv) AtlasClear's anticipated growth strategy, including the proposed acquisitions, (v) anticipated benefits of the Proposed Transaction and proposed acquisitions, (vi) the financial technology of the combined entity and (vii) expected listing of the combined company.

The forward-looking statements contained in this communication are based on the current expectations of AtlasClear, QFTA and their respective management and are subject to risks and uncertainties. No assurance can be given that future developments affecting AtlasClear, QFTA or the combined company will be those that are anticipated. Actual results may differ materially from current expectations due to changes in global, regional or local economic, business, competitive, market, regulatory and other factors, many of which are beyond the control of AtlasClear and QFTA. Should one or more of these risks or uncertainties materialize, or should any of the assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. Factors that could cause actual results to differ may emerge from time to time, and it is not possible to predict all of them.

Such factors include, but are not limited to: the risk that the transaction may not be completed in a timely manner or at all; the risk that the transaction closes but AtlasClear's acquisition of FSB does not close as a result of the failure to satisfy the conditions to closing such acquisition (including, without limitation, the receipt of approval of Commercial Bancorp stockholders and receipt of required regulatory approvals), the failure to obtain requisite approval for the transaction or meet other closing conditions; the occurrence of any event, change or other circumstances that could give rise to the termination of the definitive agreement in respect of the transaction; failure to achieve sufficient cash available (taking into account all available financing sources) following any redemptions of QFTA's public stockholders; failure to obtain the requisite approval of QFTA's stockholders; failure to meet relevant listing standards in connection with the consummation of the transaction; failure to recognize the anticipated benefits of the transaction, which may be affected by, among other things, competition, the ability of the combined entity to maintain relationships with customers and suppliers and strategic alliance third parties, and to retain its management and key employees; potential litigation relating to the proposed transaction; changes to the proposed structure of the transaction that may be required or appropriate as a result of the announcement and execution of the transaction; unexpected costs and expenses related to the transaction; estimates of AtlasClear and the combined company's financial performance being materially incorrect predictions; AtlasClear's failure to complete the proposed acquisitions on favorable terms to AtlasClear or at all; AtlasClear's inability to integrate, and to realize the benefits of, the proposed acquisitions; changes in general economic or political conditions; changes in the markets that AtlasClear targets or the combined company will target; slowdowns in securities or cryptocurrency trading or shifting demand for trading, clearing and settling financial products; the impact of the ongoing COVID-19 pandemic; any change in laws applicable to QFTA or AtlasClear or any regulatory or judicial interpretation thereof; and other factors, risks and uncertainties, including those to be included under the heading "Risk Factors" in the proxy statement/prospectus to be later filed with the SEC, and those included under the heading "Risk Factors" in QFTA's 2021 Form 10-K and its subsequent filings

with the SEC. AtlasClear and QFTA caution that the foregoing list of factors is not exhaustive. Any forward-looking statement made in this communication speaks only as of the date hereof. Plans, intentions or expectations disclosed in forward-looking statements may not be achieved and no one should place undue reliance on such forward-looking statements. Neither AtlasClear nor QFTA undertake any obligation to update, revise or review any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by any applicable securities laws.

Cover	Nov. 15, 2022
Document Type	8-K
Amendment Flag	false
Document Period End Date	Nov. 15, 2022
Entity File Number	001-40009
Entity Registrant Name	QUANTUM FINTECH ACQUISITION
	CORPORATION
Entity Central Index Key	0001830795
Entity Tax Identification Number	85-3286402
Entity Incorporation, State or Country Code	DE
Entity Address, Address Line One	4221 W. Boy Scout Blvd.
Entity Address, Address Line Two	Suite 300
Entity Address, City or Town	Tampa
Entity Address, State or Province	FL
Entity Address, Postal Zip Code	33607
City Area Code	813
Local Phone Number	257-9366
Written Communications	false
Soliciting Material	true
Pre-commencement Tender Offer	false
Pre-commencement Issuer Tender Offer	false
Entity Emerging Growth Company	true
Elected Not To Use the Extended Transition Period	false
Units, each consisting of one share of Common Stock	
and one redeemable Warrant	
<u>Title of 12(b) Security</u>	Units, each consisting of one share of Common Stock
	and one redeemable Warrant
Trading Symbol	QFTA.U
Security Exchange Name	NYSE
Common Stock, par value \$0.0001 per share	
Title of 12(b) Security	Common Stock, par value \$0.0001 per share
Trading Symbol	QFTA
Security Exchange Name	NYSE
Warrants, each warrant exercisable for one-half of one	
share of Common Stock at an exercise price of \$11.50	
Title of 12(b) Security	Warrants, each warrant exercisable for one-half of one share of Common Stock at an exercise price of \$11.50
Trading Symbol	QFTAW

"labellish": ("labellish": ("lasal": ("qfta-20221115_lab.aml"), "presentationLink": ("local": ["gfts-DIDDITIS_per.enl" }) "ashema"s ("lasal"s ("gita-20221115.and" Territoria de la construcción de "hard" "hard to be a set of the J. Section 7. Section 2018 Control 101 Co) Sector of the) Janualaana", "Amerikanat Kung", Tanata Y. Tahapay/Jaharana punyiani/2022", Tanana Y. Tahapay/Jaharana punyiani/2022", Tahapay/Jahanana, analysishi Jaharana Y. Tahanana analysishi Jaharana Y. Tahanana analysishi Jaharana analysishi 17⁻¹¹ μητη | "Ange", | "Ang 1 Single and the second information from " Single and the second information from " The second the second information from the second information of the second information "man's per ; "Eang", ("Ann", ("ann", ("ann", ("Announce and announce and a second] Sections", "ConstryDegies", Sections", "Assay/141, doi:genetics/2022", Sections", "Assay/141, doi:genetics/2022", Sections", "Sections", "Sections", "Sections", Sections, "Sections", "Sections, "Section) Transformers, Correct Floral Head Red Red Res Transformers, Correct Floral Learn, period Red / 2022 * , "http://plus.net.p.on/scient/Correct" The Start Star)) Tisologues", "Surgerstations to glassical", Provide ", "http://will.com.gov/doi/2021", Provide ", "http://will.com." Provide ", "http://will.com." "http://will.com." Provide ", "http://will.com." Provide ", "http://will.com." Provide ", "http://will.com." Provide ", "http://will.com.") Provide ", "http://will.com.")))

a April Annual

); "lecalease": "ScousechlassalBeport", "measi": "Schyp//skel.ese.gov/dei/2022", "presentales": ["http://qftacorp.com/r

)) constants, the second function of the second Technic (Technic) "decembrilian": "The start date of the period covered in the document, in YYY-88-50 format.7, "like?", "Decembril Period Date Date"); "inalians"; "December/decindlesiber,", "measir, "December/decider.exc.per/deci2022", "personalation", ["high//dptimerp.exclat/cover"]; indecider: high/dptimerp.exclat/cover"]; indecider: high/dptimer.ext. "indecider: "action ext." ["action ext."] "action ext."] 1 , and 1 ,) "Indiana": "December 2014 at indiates and "Second": "December 2014 "Second": "December 2014 "The second of the second of the second of the second "The second of the second of th]; "marvi" ("min"; ("sile" ("Account Alian": "Data of event repairing a shall company report,", "Account for Sciences Ball Company Ress Data" Auge + Auge) or Sangel | Margel | ting SEC submission type, or the word l: Target (Target) Talet (Talet) Talet (Talet Talet) 1 Translations¹, "Summarize bioregram and global mean balance", Translations¹, "Summarize bioregram and global mean balance", Translations¹, "Summarize bioregram and the second secon)) management, "Maining definition and definition of the set of ang . , Mercarts ("mile": ("decementation": "Name of the City or True", "label": "Ensity Address, City or True" ; 1 The second se

1 "Instance", "Achitybdanesoftensilligide", "west", "Aship/Inki.exe.gen(ds/2027, "Aship/Inki.

) "Incalman": "ActivitalianequicyDecomodingDeportingCoremot", Porest": "Activitaliane.quothini(2002", "presentation": " Activitalianequicane/color/2002",

Provide a service

a April Annual

tion period. The format of the date is TITT-00-12.

"wholeype": "hooleanleadype"); "del EntityCentralIndenEry": { "asD_ref": } "e2"

]; "mang": { "manus": { "sala": { "decementation": "A unique 10-digit UNC "label": "Entity Central Dodes Key" i value to identify motition that have filed disclosures with the SEC. It is commonly abbreviated as CHL",
$$\label{eq:constraints} \begin{split} & \int_{-\infty}^{1} & \\ & \int_{-\infty}^{1} & \int_{-\infty}^{1}$$
"ex-us": ("esla": ("documentation": "Indicate number of shares or other "lakel": "Entity Common Stoch, Shares Outstanding") constant": "Antipperformation with resolution main of the "main of the "interpret of the second]; "marvat { "marvat { "side"; { "Autor", "Daily Excepting Granch Company" "Autor", "Anity Excepting Granch Company")) Tenalasar, "Databgdargiaglanda kakingargi", Tenalasar, "Databgdargiaglanda kakingargi", Pangargiaglanda kakingargi Tenalasi yang sakingargi]; "serve"; { "serve"; { "serve"; } "decommonstation"; "Indicate if as emerging growth company has elected out to use the matended "Science", "Elected list To Ton the Estemated Tonaction Process". The second secon) "Institutions"+ "StatisyfileBooker", "Descrift", "Statyf/labl.exer.per/des/2022", "Pater / (definerep..em/scin/Dewe"), "statisyse", "Elektronerp.ye", ("statisyse"), (]; "many"; ["many"; ["side";] "Asserve "Society Files Company" "Asserve "Society Files Company" J.
 J. Statistical statistics and a statistical statistics and a statistical statistics.
 The statistical statistics and a statistical statistics and a statistical statistics.
 The statistic statistics and a statistic statistics.
 The statistic statistics and a statistic statistics.)) * analysis of the second 1 "Senal seam" + "Entity Determine UverBataCorrent", "Secard", "Satyp//sid.exe..gov/doi/2021", "Batyp//spin.seam/scale/Corrent" "Matypy,", "pendid teachyge" "Matypy,", "pendid teachyge" "Saty Entity Corrent ("sath exe"; ["saty]; "mernet" { "mernet" { "mernet" { "Mernemanation", "Primary Disaderd Industrial Classification (DII) Bodger for the Entity,", "Judget", "Doily Primary HIC Bodger") Schwalter, "Schlightenergisienergisienergis Schlightenergister $M_{\rm part}^{\rm eq}(1)$ () where 1 () where 1 is the second of the second of the second of the second of the second)."" "Tang": ["mang": ["maint: ["discumptions": Thisicates that the company is a Buller Reporting Company (BRC).", "Adapts": "Doily Bull Bullesters" ---Stanf"-1 "merwit-1 "slat": "Advancestation": "The Tax Identification Hodes" Isabel": "Donly Tax Identification Hodes"]) Josephinese ", "Sectory (Sector) (Sector) Sector (Sector) (Sector) Sector (Sector) (Sector) Sector () "leastern", "Materials", meast", "Materialser, gev/dek/2023", "Materialser, met/side/Exerct", "Astronomy of the second sec

2012

"Ani Learning (); "Sang": () "Sang": ()

) "mani none": "MontradingSpain-SEIng", "mani none": "MontradingSpain-SEIng", "mani none": "Montrading (Seinger, Seinger, Sein) Translammer, "Bolandapportagetandardlaminer", "sealanner", "Bolandapportagetandardlaminer", "sealanner", "Bolandardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Standardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Bolandardlaminer", "Standardlaminer", "Standar "gen""" | "Interfold" | "chart" | "Sementary | "Sementary "Therian Sing that is tore when the Form P-E filling is intended to testedy the filling abligation of the engine "Jaka", "Jack", "Arecommension Toutor Office" 1 Tendines", "Produces interpretation (2002), Tendines", "Antary / Maria Learning (2002), Tending / Aphanese, com/ania/Learning Tending / Aphanese, com/ania/Learning Tending (2002), "Second Comparison Second Comparison (2002), "Second Comparison Second Comparison (2002), "Second Comparison Second Comparison (2002), "Second Comparison Tending Comparison (2002), "Second Comparison (2002), "Second Comparison Tending Comparison (2002), "Second "to" "start": ("start": ("size": ("size": ("ident": Tible of 2100) Security" Yanger Yang Yanger Yanger Yanger Yanger Yanger Yang Yang Yanger Yang Yan 1 Tomasland", "America Stylenbargelland", Tomasland", "America Stylenbargelland", Tomasland State () T , and a set of the set interview i Televention": "a" [; "dist": ["Missel": Them 20-0"; "Missel": Tober "Multisher": "200"; "Multisher": "200; "Multisher": "200; "Multisher": "200";); "sll": ("Sime": "Press 10-K, 10-Q, 20-F" "Dominer": "200", "Problement": "200", "Section": "11", "Subsection": "s-1" 1. , s10"+ ("Esse": "Explation 2-1", "Sumber": "232", "Publisher": "EEC", "Section": "SEC",), "el?", { "Esse": "Describing Act", "Romber": "230",

Adduction (1997) The second s

No. of Concession, Name